

A framework for monitoring biodiversity in protected areas and other effective area-based conservation measures

Concepts, methods and technologies

Daniel Dalton, Vanessa Berger, Hanns Kirchmeir, Vanessa Adams, Judith Botha, Stephan Halloy, Robbie Hart, Vid Švara, Katia Torres Ribeiro, Sunita Chaudhary, Michael Jungmeier



IUCN WCPA Technical Report Series No. 7









IUCN WCPA PUBLICATIONS

IUCN WCPA publishes two landmark series providing authoritative resources for managers of protected areas and other effective area-based conservation measures (OECMs), policy makers and scientists. Definitions and details of guidance on protected areas and OECMs can be found on the inside back cover of this publication.

The Good Practice Guidelines and Technical Reports involve collaboration among specialist practitioners dedicated to supporting better implementation of conservation in the field, distilling learning and advice drawn from across IUCN WCPA. Applied in the field, they build institutional and individual capacity to manage conservation systems effectively, equitably and sustainably, and help cope with the myriad of challenges faced around the world. The Guidelines also assist national governments, protected area agencies, non-governmental organizations, communities and private sector partners in meeting their commitments and goals, and especially to the Convention on Biological Diversity.

For more information on all WCPA publications see: https://www.iucn.org/our-union/commissions/world-commission-protected-areas/our-work/wcpa-publications

For information on publishing with WCPA see: https://www.iucn.org/our-union/commissions/world-commission-protected-areas/our-work/wcpa-publications/publishing-wcpa#:~:text=WCPA%20publishes%20a%20variety%20 of,publication%20standards%20set%20by%20IUCN.

Complementary resources are available at: www.30x30.solutions

Contribute to developing capacity for a Protected Planet at: www.protectedplanet.net

A framework for monitoring biodiversity in protected areas and other effective area-based conservation measures











UNIVERZAV L]UBL]ANI University of Ljubljana



International Union for Conservation of Nature

IUCN is a membership Union uniquely composed of both government and civil society organisations. It provides public, private and non-governmental organisations with the knowledge and tools that enable human progress, economic development and nature conservation to take place together.

Created in 1948, IUCN is now the world's largest and most diverse environmental network, harnessing the knowledge, resources and reach of more than 1,400 Member organisations and around 16,000 experts. It is a leading provider of conservation data, assessments and analysis. Its broad membership enables IUCN to fill the role of incubator and trusted repository of best practices, tools and international standards.

IUCN provides a neutral space in which diverse stakeholders including governments, NGOs, scientists, businesses, local communities, Indigenous peoples' organisations and others can work together to forge and implement solutions to environmental challenges and achieve sustainable development.

Working with many partners and supporters, IUCN implements a large and diverse portfolio of conservation projects worldwide. Combining the latest science with the traditional knowledge of local communities, these projects work to reverse habitat loss, restore ecosystems and improve people's well-being.

www.iucn.org https://twitter.com/IUCN/



IUCN World Commission on Protected Areas

IUCN's World Commission on Protected Areas (WCPA) is the world's premier network of protected and conserved areas expertise. The Commission has over 2,500 members spanning 140 countries who provide strategic advice to policymakers and work to strengthen capacity and investment for protected areas establishment and management. The Technical Reports series is one of the Commission's flagship products, providing timely guidance on aspects of protected area planning, management and assessment.

iucn.org/wcpa



Carinthia University of Applied Sciences

Carinthia University of Applied Sciences is a leading institute of higher education in southern Austria and is home to the UNESCO Chair on Sustainable Management of Conservation Areas. Providing high quality educational offers, international partnerships and practical work experiences, the university has five campus locations featuring areas of expertise in Civil Engineering & Architecture, Engineering & IT, Health Sciences & Social Work and School of Management. Many faculty specialise in high-tech fields, including using state-of-the-art technologies for environmental analysis and monitoring. About 40 degree programmes are offered at the Bachelor's or Master's level, including the distinguished Master of Science programme Management of Conservation Areas.

cuas.at/en



UNESCO Chair on Sustainable Management of Conservation Areas

The UNESCO Chair on Sustainable Management of Conservation Areas is one of about 950 UNESCO Chair programmes worldwide and was established at Carinthia University of Applied Sciences, Austria in 2020. Our integrated programme is part of the UNITWIN Network, combining research, teaching, community engagement and training to advance UNESCO's role as a global observatory and source of innovative ideas. Our UNESCO Chair is composed of international scientists and a robust network of collaborators that bring diverse experiences and perspectives to the table, resulting in strong global partnerships.

unesco.org/en

ICIMOD

ICIMOD

The International Centre for Integrated Mountain Development (ICIMOD) is an intergovernmental knowledge and learning centre working in eight regional member countries of the Hindu Kush Himalaya (HKH) region – Afghanistan, Bangladesh, Bhutan, China, India, Myanmar, Nepal, and Pakistan. ICIMOD works to improve the lives and livelihoods of men, women and children of the HKH and protect mountain environments and cultures.

icimod.org



E.C.O. Institute of Ecology

E.C.O. Institute of Ecology is an Austrian company that is a world leader in planning, consulting, research and training for nature conservation, certified regions and protected areas since 1997. The company has a permanent staff of around 20 highly qualified professionals, supported by an international network of cooperating professionals and organisations. E.C.O. provides consultancy, research, training and project design services specifically for protected areas and for organisations or authorities concerned with conservation, natural resources, land-use planning or ecology. Based on many years of practical experience, E.C.O. develops ecological solutions in close cooperation with different stakeholders and projects.

e-c-o.at



ICMBio

The Chico Mendes Institute for Biodiversity Conservation (ICMBio) is a Brazilian federal institution linked to the Ministry of the Environment and Climate Change. It is in charge of 336 protected federal areas, in the most diverse IUCN categories, including areas of shared management with traditional communities. ICMBio is also responsible for articulating conservation strategies for endangered fauna and protecting the country's speleological heritage, through national conservation action plans. The main lines of action are promoting public use, strengthening biodiversity production chains, promoting research and monitoring. In addition, the institution supports activities such as enforcement and land regularization, having social participation as a transversal axis of action.

gov.br/icmbio



Missouri Botanical Garden

The Missouri Botanical Garden's mission is "To discover and share knowledge about plants and their environment in order to preserve and enrich life". Founded in 1859, the Missouri Botanical Garden is the USA's oldest botanical garden in continuous operation and a National Historic Landmark. The Garden is a centre for botanical research and science education, as well as an oasis in the city of St. Louis. The Garden offers 79 acres of beautiful horticultural display, including a 14-acre Japanese strolling garden, historic architecture, and one of the world's largest collections of rare and endangered flora. The Garden's Science and Conservation division is dedicated to exploring, conserving, and restoring global plant life and ecosystems while amplifying impact through collaboration to ensure a sustainable future for biodiversity and people.

missouribotanicalgarden.org



South African National Parks

South African National Parks (SANParks) is based in South Africa and is mandated to conserve both natural and cultural heritage through the network of protected areas that they manage. The bulk of the biophysical and social science monitoring and research is conducted by the Scientific Services division. The generated knowledge informs park management and promotes the conservation of biodiversity. Biophysical monitoring and research span the terrestrial, aquatic and marine domains and includes current threats such as climate change, invasive alien species and many more.

sanparks.org



University of Tasmania

Since its beginnings in 1890, the University of Tasmania has established a strong foundation in research excellence. The University of Tasmania strategy focuses on place-based but globally significant research, world-leading for Tasmania and from Tasmania to the world. Environmental research is central to this. The School of Geography, Planning, and Spatial Sciences builds on and advances the University's strategic pursuit of academic and applied multi-disciplinarity, encompassing and integrating human geography, physical geography, spatial sciences, planning, and environmental management. The School seeks to leverage its collective expertise in geography, planning, and spatial sciences to contribute to a sustainable and equitable future. We believe in the power of scientific inquiry, systems thinking, and community collaboration to generate solutions that address complex social and environmental challenges.

utas.edu.au



Bahir Dar University

Bahir Dar University (BDU) is situated in the beautiful city of Bahir Dar, Ethiopia at the southern shore of Lake Tana Biosphere Reserve. Lake Tana is the largest freshwater body protected by UNESCO. The establishment of BDU is associated with the beginning of Bahir Dar Polytechnic Institute and Bahir Dar Teachers' College in 1963 and 1972, respectively. Currently, BDU is one of the largest universities in Ethiopia enrolling more than 40,000 students in its 434 programmes (115 undergraduate and 297 postgraduate programmes, 16 specialty and subspecialty and six certificate programmes). Nationally, BDU has been ranked first in the higher education differentiation scrutiny by the Ethiopian Ministry of Education. It has also been ranked in the top 20 Sub-Sahara African universities. BDU aspires to be one of the leading research-intensive universities in Africa and the first choice in Ethiopia by 2030. It aligns its operational and educational priorities with relevant Sustainable Development Goals.

bdu.edu.et



Observatoire des Forêts d'Afrique Centrale

Created in 2007, the Observatoire des Forêts d'Afrique Centrale (OFAC) is a specialised unit of the Commission des Forêts d'Afrique Centrale (COMIFAC) that provides up-todate, relevant data on the region's forests and ecosystems, aimed at informing political decision-making and promoting better governance and sustainable management of natural resources. OFAC collaborates with several international partners (IUCN, UNEP-WCMC, FAO, JRC, ATIBT, etc.). Its information system is organised around an online interactive analytical platform (https://www.observatoire-comifac. net/analytical_platform), and the observatory publishes numerous analyses through maps and flagship publications such as the State of Forests and the State of Protected Areas. OFAC is supported by the RIOFAC project, financed by the European Union.

observatoire-comifac.net



Centre for Biodiversity Conservation Research

The Centre for Biodiversity Conservation Research (CBCR) is an autonomous research institution hosted by the University of Ghana. CBCR's mission is to promote the conservation of biological diversity for the benefit of current and future generations of people. The Centre's strategies for achieving its mission include, knowledge generation, training/capacity building, networking, policy analysis and research dissemination. CBCR's current programme focus includes: wetlands, a most vulnerable ecosystem in Ghana and globally; protected areas, a key tool for biodiversity conservation; species conservation to stem the tide of biodiversity losses and ecosystem services to demonstrate the value of biodiversity and highlight the nexus between nature conservation, livelihoods and development.

cbcr-ug.org



EarthRanger

EarthRanger is a data visualisation and analysis software platform that gives conservationists the real-time information they require to keep wildlife, habitats and communities safe. Easy to use and free for conservation missions, the platform collects, integrates, and displays all historical and available data and combines it with reports from the field to provide one unified view of collared wildlife, rangers and any other assets whether on land or sea. The EarthRanger programme has been in place for over nine years and has reduced threats to biodiversity and habitats in over 500 protected areas across more than 60 countries and on six continents. It has facilitated the reintroduction and restoration of diverse species and ecosystems that deliver global environmental and socioeconomic benefits. The programme is a philanthropic initiative by the Allen Institute for Artificial Intelligence (AI2), a non-profit founded by the late Paul Allen, co-founder of Microsoft.

EarthRanger.com



DivjaLabs

DivjaLabs, a spin-out company of the University of Ljubljana, Slovenia, was founded in 2022. Its diverse, international and multi-disciplinary team of scientists leverages stateof-the-art eDNA-based molecular and computational tools to tackle challenges in biodiversity conservation and wildlife management. With its dedication to innovation and appreciation of nature and wildlife, the DivjaLabs team has rapidly emerged as a frontrunner in European biodiversity and wildlife monitoring and research.

divjalabs.com



GEO BON

The Group on Earth Observations Biodiversity Observation Network (GEO BON) is a flagship of GEO and is hosted by the Quebec Centre for Biodiversity Science at McGill University in Montreal, Canada. GEO BON is a rapidly growing global research network and community of practice of nearly 3,000 members, from more than 1,700 organisations and 144 countries, dedicated to improved monitoring of Earth's biodiversity. GEO BON's mission is to improve the acquisition, coordination and delivery of biodiversity observations and related services to users including decision-makers and the scientific community. GEO BON initiates and coordinates efforts to design and implement interoperable national and regional biodiversity monitoring programmes with a vision of a global biodiversity observing system that contributes to effective management policies for the world's biodiversity and ecosystem services.

geobon.org



National Institute for Environmental Studies

The National Institute for Environmental Studies (NIES) was established in Japan in 1974 as the National Institute for Pollution Studies and was renamed in 1990. NIES focuses on research related to societal, social changes, and environmental challenges. The relationship between climate change and natural disasters, which have become increasingly common in recent years both in Japan and globally, has become a matter of major public concern, prompting Japanese government declarations that set goals for adaptation to climate change and achievement of carbon neutrality. The next decade will be a crucial period for achieving these goals and building a new society. At NIES, the primary mission is to study the many issues related to these challenges and provide scientific knowledge to inform the decisions of the Japanese government and the public. Since 2013, NIES has also collaborated with the International Union for Conservation of Nature-Japan (IUCN-J) to enhance biodiversity conservation efforts.



Stellenbosch University

Based at a world-class South African tertiary institution, Stellenbosch University, the School for Climate Studies is focused on developing and implementing an Africarelevant research programme that responds to existing and emerging issues in climate change impacts, adaptation and mitigation responses, thus supporting human climate resilience on the continent. By working across all faculties in the university, it is developing new academic capacity, career paths and infrastructure for climate studies, creating opportunities for students to develop the skills required to work in leading national and international public and private entities engaged in fundamental and applied climate studies. The School consolidates and integrates current disciplinary and transdisciplinary thinking on climate, conducts and coordinates primary research, publishes and makes accessible research results, and creates or supports the required platforms for data-intensive research and innovation.

climate.sun.ac.za



nies.go.jp/index-e.html

Universiti Teknologi MARA

Founded in 1956, Universiti Teknologi MARA (UiTM) is a mega public university in Malaysia. With its main campus in Shah Alam, the university has 34 campuses spread throughout the nation. UiTM is not only the largest university infra-structurally, but also demographically with 187,551 local and international student enrolment supported by 18,983 academic and non-academic staff. The university currently offers 510 academic programmes, and providing innovative education within four colleges of study, 14 faculties and seven academic centres across Malaysia. UiTM campuses are well equipped with modern facilities to accommodate its community's academic and research needs. The university has earned a reputation for teaching and research excellence over the last six decades.

uitm.edu.my



University of Florence

The University of Florence is a leading European research and higher education institution, including on biology and biodiversity science. The University offers 126 Degree courses (Bachelor and Master Degrees) to a population of approximately 50,000 students, and each year awards around 9,000 diplomas. The Department of Biology is one of the 24 departments and carries out research mainly on conservation biology, functional and structural biology, and evolutionary biology. The various research groups that focus on biodiversity include the laboratory of animal ecology and biodiversity conservation.

ecologyandbiodiversity.unifi.it



University of Ljubljana

The University of Ljubljana, founded in 1919, stands as Slovenia's premier higher education and research institution. With over 40,000 students and 6,000 faculty and staff across 23 faculties and three arts academies, our rich tradition and modern facilities are central to Ljubljana's academic landscape. Consistently ranked among the top 500 universities globally by multiple analytics firms, University of Ljubljana is an important hub of academic and research activity.

uni-lj.si/

A framework for monitoring biodiversity in protected areas and other effective area-based conservation measures

Concepts, methods and technologies

Daniel Dalton, Vanessa Berger, Hanns Kirchmeir, Vanessa Adams, Judith Botha, Stephan Halloy, Robbie Hart, Vid Švara, Katia Torres Ribeiro, Sunita Chaudhary, Michael Jungmeier

IUCN WCPA Technical Report Series No. 7

The designation of geographical entities in this publication, and the presentation of the material, do not imply the expression of any opinion whatsoever on the part of IUCN or other participating organizations concerning the legal status of any country, territory, or area, or of its authorities, or concerning the delimitation of its frontiers or boundaries.

The views expressed in this publication do not necessarily reflect those of IUCN or other participating organizations. Mention of commercial products does not constitute endorsement by the sponsors of this publication.

IUCN is pleased to acknowledge the support of its Framework Partners who provide core funding: Ministry of Foreign Affairs, Denmark; Ministry for Foreign Affairs, Finland; Government of France and the French Development Agency (AFD); Ministry of Environment, Republic of Korea; Ministry of the Environment, Climate and Sustainable Development, Grand Duchy of Luxembourg; the Norwegian Agency for Development Cooperation (Norad); the Swedish International Development Cooperation Agency (Sida); the Swiss Agency for Development and Cooperation (SDC) and the United States Department of State.

This publication was supported through funding by the Austrian Research Promotion Agency (FFG) COIN project Biodiversity Monitoring Technologies Test, Development and Transfer of disruptive engineering technologies into conservation practice (BioMONITec) and the Austrian Federal Ministry of Labour and Economy (BMAW).

Published by:	IUCN, Gland, Switzerland
Produced by:	IUCN WCPA and Carinthia University of Applied Sciences UNESCO Chair on Sustainable Management of Conservation Areas
Copyright:	© 2024 IUCN, International Union for Conservation of Nature and Natural Resources
Developed to the set	this publication for advantional or ather parts companying purposes in

Reproduction of this publication for educational or other non-commercial purposes is authorised without prior written permission from the copyright holder provided the source is fully acknowledged.

Reproduction of this publication for resale or other commercial purposes is prohibited without prior written permission of the copyright holder.

Recommended	Dalton, D., Berger, V., Kirchmeir, H., Adams, V., Botha, J., Halloy, S., Hart, R.,
Citation.	A framework for monitoring biodiversity in protected areas and other effective
	area-based conservation measures: Concepts, methods and technologies.
	IUCN WCPA Technical Report Series No. 7, Gland, Switzerland: IUCN.

ISBN: 978-2-8317-2268-9 (PDF)

DOI: https://doi.org/10.2305/HRAP7908

- Cover photo: Red-eyed tree frog, *Agalychnis callidryas*, La Gamba, Costa Rica. © Jennifer Insupp
- Back cover: Großglockner, Austria © Vanessa Berger
- Figure design: Vanessa Berger, Daniel Dalton, Anneliese Fuchs, Jennifer Insupp, Michael Jungmeier, Larissa Keintzel, Elisabeth Wiegele
- Layout: Elisabeth Wiegele, Carinthia University of Applied Sciences

Template provided by Miller Design, millerdesign.co.uk

Contents

List of figures List of tables List of annex figures List of annex tables List of checklists List of boxes Preface Executive summary Acknowledgements Glossary	XIV XIV XIV XIV XIV XV XVI XVI XVII XIX XX
CHAPTER 1. INTRODUCTION	2
CHAPTER 2. PREPARATORY PHASE	7
2.1 Conventions and standards	7
2.2 Basic investigation and site assessment	8
2.3 Specifying conservation objectives	8
2.4 Management planning	9
2.5 Output: Statement of purpose	11
CHAPTER 3. CONCEPTUAL PHASE	14
3.1 Why: The purpose of monitoring	15
3.2 What: Indicators in a biodiversity monitoring programme	15
3.3 Where: Scale of spatial features	18
3.4 When: Scale of temporal features	19
3.5 Who: Identifying actors and stakeholders	21
3.6 Required resources: Identifying the resource pool	23
3.7 Output: Defined scope of the biodiversity monitoring programme	23
CHAPTER 4. IMPLEMENTATION PHASE	25
4.1 Deciding on sample design and methods	27
4.2 Acquisition and customisation of tools and materials	29
4.3 Elaboration of a field manual	30
4.4 Conducting test runs	30
4.5 Ongoing monitoring cycles	31

39

CHAPTER 5. RE-EVALUATION PHASE

CHAPTER 6. GENERAL CONSIDERATIONS	41
6.1 Obligations: International conventions and policies	41
6.2 Art of omission: Daring to simplify	44
6.3 Biodiversity monitoring systems	45
6.4 Combining forms of knowledge	47
6.5 Continuity risks: Avoiding disruptions and gaps in data	48
6.6 Detecting trends and correlations: The value of time series	49
6.7 Maintaining ecological balance: Establishing baselines and thresholds	51
6.8 Setting up monitoring systems: Costs and outcomes	51
6.9 Protected Area Management Effectiveness evaluation tools	52
CHAPTER 7. A REVIEW OF METHODS AND TECHNOLOGIES TO IMPLEMENT	55
7.1. Early bird or late adoptor: Drivers and barriers of technology deployment	56
7.2 Toolkit: Overview of the indicators	56
7.2 Tochnology based approaches for biodiversity monitoring	50
7.4 Analysis: Applying advanced computer technologies for hig data	64 64
	04
CHAPTER 8. SYNTHESIS: A NEW AGE OF BIODIVERSITY MONITORING	69

REFERENCES	70
ANNEXES	77
AUTHOR PROFILES	86

List of figures

Figure 1	Key phases of developing a biodiversity monitoring programme in protected areas and OECMs	XVIII
Figure 2	Scope of the framework for monitoring biodiversity in protected areas and OECMs	2
Figure 3	Ontology of the preparatory phase	7
Figure 4	The framework cycle of drivers, pressures, state, impacts and responses is the main tool for managers	10
Figure 5	Ontology of the conceptual phase	14
Figure 6	Indicator selection will be site-specific and based on biodiversity monitoring system objectives	15
Figure 7	Selection of key indicators is ideally based on two attributes	16
Figure 8	Spatial considerations of monitoring include area of interest, sampling format and statistical design	18
Figure 9	The spatial scale of a biodiversity monitoring programme can range from a genetic to landscape level	18
Figure 10	Temporal aspects of a biodiversity monitoring programme	19
Figure 11	Timescale of different biological processes	20
Figure 12	Example stakeholder worksheet listing key participants in a biodiversity monitoring programme	21
Figure 13	A biodiversity monitoring programme depends on the available staff, resources and knowledge	23
Figure 14	Ontology of the implementation phase	25
Figure 15	Example plot designs for biodiversity monitoring	27
Figure 16	Monitoring system implementation and data management	44
Figure 17	Data are brought together to determine a modular framework of new biodiversity monitoring programmes	46
Figure 18	Monitoring at an improper frequency may display faulty trends	48
Figure 19	Timeline of key achievements advancing monitoring capacities	49
Figure 20	Long-term monitoring provides increasingly accurate knowledge	50
Figure 21	The initial stages of a monitoring programme come with high up-front costs	51
Figure 22	Biodiversity assessment using genetic methods follows a uniform workflow	64

List of tables

Table 1	Examples of proxy indicators for biodiversity monitoring	16
Table 2	Plot designs used in field-level biodiversity monitoring programmes	28
Table 3	State-of-the-art approaches to biodiversity monitoring	29
Table 4	List of main international biodiversity treaties and conventions	42
Table 5	Criteria and assumptions for integrating Nature-based Solutions into management activities	43
Table 6	Overview of different techniques for biodiversity monitoring with examples, benefits and limitations	55
Table 7	A selection of apps and data logging tools for modern biodiversity assessment	65

List of annex figures

Annex Figure 1	Blank monitoring concept worksheet	77
Annex Figure 2	Cheat sheet for monitoring concept worksheet	78
Annex Figure 3	Blank stakeholder worksheet	79

List of annex tables

Annex Table 1	Point system to rate biodiversity monitoring programme priorities	80
Annex Table 2	Cost worksheet describing the human resources, travel and material expenses for data collection	81

List of checklists

Checklist 1	Ethical and cultural considerations of the implementation phase	82
Checklist 2	Field logistics	82
Checklist 3	Safety and field training	83
Checklist 4	Safe field work and data acquisition	83
Checklist 5	Data management	84
Checklist 6	Communication	85

List of boxes

Box 1	Scope of monitoring, biodiversity monitoring systems and monitoring programmes	XVII
Box 2	A global concept: Understanding indicators through the Essential Biodiversity Variables	8
Box 3	Bush encroachment in an African savanna	10
Box 4	South Georgia and South Sandwich Islands Marine Protected Area	11
Box 5	Indicator selection: Biosphere Reserve Integrated Monitoring in the Nock Mountains of Austria	17
Box 6	Integrating citizen science to estimate the Slovenian brown bear population: It counts to be involved	22
Box 7	Test and experimental sites: BioDivTecs Hub Carinthia, Austria	26
Box 8	Working with SMART and CyberTracker: An example at Lake Tana, Ethiopia	32
Box 9	Unified approach to scientific monitoring in the mountains: Example of the GLORIA network	34
Box 10	CARE and FAIR data sharing principles: Protecting the data sources	35
Box 11	Dashboard view: The example of EarthRanger	36
Box 12	Monitoring in Natural World Heritage sites: Japanese island ecosystems in focus	41
Box 13	'Best of' mistakes in monitoring: Succession in a fallow field in Austria	44
Box 14	From individual sites to a protected area system: The Brazilian Programa Monitora	45
Box 15	Forest restoration using traditional ecological knowledge in a transboundary biosphere reserve	47
Box 16	Monitoring and evaluation of environmental measures: Water usage in the south-east of the United States	50
Box 17	Integrated Management Effectiveness Tool (IMET): Implementation in central Africa	52
Box 18	High-tech approaches: Camera trapping for biodiversity monitoring	61
Box 19	High-tech approaches: Airborne laser scanning for forest biodiversity assessment	62
Box 20	High-tech approaches: Terrestrial laser scanning for forest biodiversity assessment	66



High-elevation arid plateau featuring giant lobelia (*Lobelia rhynchopetalum*), Bale Mountains National Park, Ethiopia. © *Michael Jungmeier*

Preface

The growing pressures on Earth's natural environments threaten biodiversity and the functioning of ecosystems. Protected areas and other effective area-based conservation measures provide areas of land and water to safeguard species populations. As their numbers and expanse increase, these landscapes play a key role in conservation. Yet, management of our protected areas requires a strategic and evidence-based approach. Accurate information is gained through monitoring the right indicators at the right time. Maintaining a balance between ecological and cultural elements is a major challenge to site management, especially in the absence of sufficient resources or when monitoring programmes are not performed consistently. To make the best use of limited resources, systematic monitoring of management outcomes that affect biodiversity is critical.

This Technical Report Series guideline provides a decision-making framework for managers to develop biodiversity monitoring programmes in protected areas and other effective area-based conservation measures. The guideline is organised into eight chapters and contains several workshop tools to assist in programme conceptualisation.

Chapter 1 introduces the road map for a four-step framework that will guide managers in the development of efficient and meaningful long-term biodiversity monitoring programmes.

Chapter 2 details the information required to complete the first step of the decision-making framework, the preparatory phase. Managers identify international conventions that require monitoring and gather all background information and site conservation objectives. This step helps to plan and prioritise management activities, leading to documentation of a monitoring statement of purpose.

Chapter 3 defines the six questions that guide the conceptual phase of a new biodiversity monitoring programme. Managers and site employees address the questions together to identify why monitoring is needed, what indicators are most appropriate to monitor, where the programme will take place, when monitoring will occur, who will be involved in the programme and what their responsibilities are, and how many resources are needed to accomplish the programme. This dialogue reveals a meaningful and realistic scope of monitoring.

Chapter 4 describes the implementation phase of the biodiversity monitoring programme. In this phase managers define how the decisions from the conceptual phase are put into action, including developing a statistically robust sampling design and determining which tools to use. Field monitoring cycles are outlined with consideration towards data management and analysis.

Chapter 5 briefly expresses the value of periodic programme re-evaluation to determine whether the programme should be continued in its original state, whether and how it should be modified to address site management, or whether it should be terminated.

Chapter 6 details the many general considerations that help make effective biodiversity monitoring programmes.

Chapter 7 provides an overview on the methods and technologies available today for biodiversity monitoring. Expert monitoring tips are provided for diverse species groups. These are followed by a review of the suitability of different types of monitoring tools and techniques for monitoring target species. Advanced data analytical techniques are introduced.

Chapter 8 provides a synthesis of the guideline and a vision of hope as we face the growing challenges to biodiversity.

To complete the guideline, a series of annex figures and tables is provided, followed by checklists to help prepare for efficient field work and data collection. These materials are designed to be used in a workshop setting to effectively communicate ideas and decisions to involved staff and stakeholders.

We express in this guideline that the value of consistent design and good methodological and technical preparation is fundamental for effective biodiversity monitoring. In addition, we show our enthusiasm for technologies and approaches that are currently transforming biodiversity monitoring. We cast a wide net to gain the perspectives of protected area managers, research scientists, IUCN scientists and other stakeholders.

Each monitoring programme is a commitment to describe our planet's rich biodiversity. May this publication empower and guide all those dedicated to managing our natural wonders.

Executive summary

Protected areas and other effective area-based conservation measures (OECMs) are important to stop the global decline in biodiversity. Systematic site-based monitoring of the state of biodiversity and conservation outcomes is necessary for evidence-based adaptive management in protected areas and OECMs. Biodiversity monitoring is also important to inform managers if they are meeting their conservation goals.

The framework described in this publication will help managers and site planners to consider all relevant details to develop effective biodiversity monitoring programmes for improved management outcomes (Box 1). A step-by-step approach is provided on how to establish biodiversity monitoring programmes as components of biodiversity monitoring systems in conservation areas including protected areas and OECMs, Key Biodiversity Areas and UNESCO sites including biosphere reserves, Global Geoparks and Natural World Heritage Sites. More than 295,000 protected areas and OECMs are represented on the World Database on Protected Areas (UNEP-WCMC, 2023a), indicating the great need for utilising standardised monitoring frameworks whilst meeting local objectives. Today's monitoring technologies are undergoing a rapid evolution, and this publication also provides a snapshot of the state-of-the-art of monitoring tools.

Box 1

Scope of monitoring, biodiversity monitoring systems and monitoring programmes

Monitoring is the process of regular data collection and analysis that is then used by managers to determine whether project objectives are being met. Monitoring may occur at multiple points within the framework cycle of drivers, pressures, state, impacts and responses and is the main tool for managers to determine overall management effectiveness at their sites. Multiple monitoring programmes are contained under a comprehensive biodiversity monitoring system, supporting management objectives of protected areas and OECMs.

A biodiversity monitoring programme is an ongoing module of the biodiversity monitoring system. It is designed to deliver benefits to the organisation that are aligned with its objectives (Weaver, 2010). Programmes may or may not be time-limited, as determined by the organisational structure of the protected area or OECM.

A biodiversity monitoring project is an activity that is usually limited in time and funding. A project is designed to deliver a specific output. Efficiency of the work is key to a successful project (Weaver, 2010). Monitoring projects serve as opportunities for protected areas and OECMs to showcase effective monitoring methods and their unique biodiversity.

The schematic shown in Figure 2 describes the different purposes of monitoring. If information on the indicator is missing, and management activities have not been determined, typically a research project will generate basic site information. If the state of the indicator is unknown, but management activities have already been determined, monitoring is conducted as a baseline study. If the status of the indicator is known, and the management activities are not decided, monitoring is most suitable for regional documentation. If information on the indicators is available and the desired management techniques are known, monitoring to support site management may proceed. This guideline focuses on this last scenario, monitoring for management purposes.

Source: Compiled by the report authors

The framework for developing biodiversity monitoring programmes is introduced in a stepby-step manner consisting of four phases: a) preparatory phase; b) conceptual phase; c) implementation phase with periodic interim evaluation guiding adaptive management; and d) periodic re-evaluation (Figure 1). We conclude with a discussion on the level of technology that is required for adequate monitoring, considering that monitoring is a long-term activity and the state-of-the-art is continually improving. We point out the importance of data continuity given the different methodologies and monitoring tools worldwide. Developing a good biodiversity monitoring system requires using different forms of knowledge, ethical issues, scientific evaluation and effective communication.



All site-based monitoring programmes should be well-integrated into the biodiversity monitoring system. Within individual protected areas or OECMs, several types of monitoring programmes may be established, contributing to protected area management effectiveness (PAME), law enforcement, threat assessment and resource use efficiency. Biodiversity monitoring programmes must produce accurate data supporting site management, and multiple sites will ideally be included in a programme for conclusions on long-term trends. Effective management will in turn help develop policy recommendations to remedy the loss of biodiversity and restore damaged ecosystems.

Figure 1 Key phases of developing a biodiversity monitoring programme in protected areas and other effective area-based conservation measures Effective biodiversity monitoring programme planning will correspond with site management goals. In the preparatory phase, a review of the site details and obligations should be conducted, resulting in a monitoring statement of purpose. The conceptual phase follows, where a series of basic questions are addressed about the intended monitoring programme. The result is an understanding of how the monitoring programme will be implemented. To verify the protocols, test runs are conducted in the implementation phase prior to the repeating cycles of the programme. Data analysis will guide management decisions based on findings. Detailed re-evaluation of the programme occurs after a predetermined number of cycles, providing opportunities for adaptive management. Source: Compiled by the report authors

Acknowledgements

This Technical Report Series guideline is the result of a large joint effort of conservation area management practitioners, scientists and stakeholders. We extend our gratitude towards peer reviewers Mr Ludi Apin, Dr Marta Múgica and Dr Stephen Woodley for their critical reviews of the ideas and content expressed in the guideline, and for the proofreading services of Mrs Caroline Snow. We thank all participants who contributed ideas, text, graphics and support to the production of this guideline. In particular, we thank IUCN World Commission on Protected Areas Series Editors Nigel Dudley and Sue Stolton, active members of the IUCN WCPA and other IUCN Commissions, namely Wendy Foden, Sue Gubbay, Paola Mejia, Florence Palla, Francesco Rovero, Nick Salafsky, Tomaž Skrbinšek, Andrej Sovinc, Erika Vaida-Bela and Sue Wells for their support and critical analyses of draft versions of the guideline. We acknowledge the advice of Working Group members Vanessa Adams, Maria Augusta Almeida Ferri, Vanessa Berger, Judith Botha, Sunita Chaudhary, Daniel Dalton, Stephan Halloy, Robbie Hart, Michael Jungmeier, Hanns Kirchmeir, Yaa Ntiamoa-Baidu, Mohd Nazip Suratman and Katia Torres Ribeiro. We recognise Tirusew Ebistu, Tobias Fremout, GEO BON, Kerry Grey, Donald Jomha Djossi, Instituto Chico Mendes de Conservação da Biodiversidade, Quentin Jungers, Clive Kaiser, Miha Krofel, Gernot Kunz, Daniel Mengistu, Frank Muller-Karger, Faizah Pardi, Harald Pauli, Michael Russell, Ulf Scherling, Jordan Steward and Yayoi Takeuchi for their contributions and reviews of the guideline. We extend our sincere appreciation for the support of the team at the UNESCO Chair on Sustainable Management of Conservation Areas, Villach, Austria, namely Melanie Erlacher, Jennifer Insupp, Larissa Keintzel, Klaus Steinbauer, Vid Švara, Ilja Svetnik, Elisabeth Wiegele, as well as colleagues Anneliese Fuchs and Michael Huber of E.C.O. Institute of Ecology, Klagenfurt, Austria. We thank the countless other colleagues worldwide who provided feedback to us in remote presentations and conference calls. We express our sincere thanks to the IUCN publishing team, Global Communications Unit, and IUCN WCPA Chair Dr Madhu Rao for their support throughout the review process.

The writing of this guideline was supported financially by the Austrian Research Promotion Agency (Österreichische Forschungsförderungsgesellschaft, FFG) through the COIN project, no. 884138, BioMONITec and the Austrian Federal Ministry of Labour and Economy (Bundesministerium für Arbeit und Wirtschaft, BMAW). This funding further supported the development of a prototype monitoring configurator, 'MoniConfig', that is designed to be an online decision support aid to assist managers of protected areas and OECMs in meeting their monitoring objectives.

All opinions, findings, conclusions and recommendations expressed in this Technical Report Series guideline are those of the authors and do not necessarily reflect the view of the funding agencies or the IUCN WCPA. All textual errors are the responsibility of the authors.



Primeval beech forest, Poloniny National Park, Slovakia, part of the Serial World Heritage Site "Ancient and Primeval Beech Forests of the Carpathians and Other Regions of Europe". © Hanns Kirchmeir

Glossary

Adaptive management. A systematic process of continually improving management policies and practices by learning from the outcomes of existing programmes (IUCN, 2022b).

Area of interest. Area representing a habitat, environment or ecosystem where biodiversity monitoring takes place.

Biodiversity monitoring. Regular, statistically designed counts of a population in order to watch its numbers, composition and distribution (IUCN, 2022b).

(Biodiversity) monitoring programme. An ongoing component of the site management framework that addresses site objectives (e.g. vegetation monitoring, bird monitoring).

Biodiversity monitoring system. Framework that considers management objectives, conceptual elements and best practices to support multiple biodiversity monitoring programmes.

Conservation goal. An identified target that is intended to be reached in conservation planning, including species, habitats, landscapes, biodiversity or ecosystem services (Lacher, 2018).

Conservation outcome. The result of a management action.

Convention on Biological Diversity (CBD). International agreement effective from 29 December 1993 containing three main goals: conservation of biodiversity; sustainable use of biodiversity resources; and fair and equitable sharing of benefits from genetic resources (CBD, 1992).

Drivers, pressures, state, impacts and responses. Causal chain framework where driving forces such as economic development put pressure on the environment, changing its state. These changes lead to impacts on ecosystems, leading to a societal response that feeds back into the causal chain (Niemeijer & de Groot, 2008).

Flagship species. Popular charismatic species that serve as symbols to stimulate conservation awareness and action locally, nationally, regionally or globally (IUCN, 2022b).

Governance. Relating to seven principles of legitimacy, transparency, accountability, inclusiveness, fairness, connectivity and resilience that enable positive management outcomes (Lockwood, 2010).

Indicator (species). A species sensitive to environmental change, which can therefore provide a measure of health for the ecosystem (IUCN, 2022b).

Indigenous peoples and local communities. The CBD does not recommend a formal definition of Indigenous peoples and local communities. However, the IPBES Global Assessment Report on Biodiversity and Ecosystem Services provides the following definition: "individuals and communities who are, on the one hand, self-identified as indigenous and, on the other hand, are members of local communities that maintain inter-generational connection to place and nature through livelihood, cultural identity and worldviews, institutions and ecological knowledge" (Brondizio et al., 2019).

Key Biodiversity Area. Site contributing significantly to the global persistence of biodiversity (IUCN, 2016).

Kunming-Montreal Global Biodiversity Framework

(GBF). Building on the CBD Strategic Plan for Biodiversity 2011–2020, the GBF calls for transformative changes in the world approach to conserving biodiversity by 2050, including placing at least 30 per cent of the Earth's terrestrial surface under effective conservation, with emphasis on management effectiveness.

Marine Protected Area. Any area of intertidal or subtidal terrain, together with its overlying water and associated flora, fauna, historical and cultural features, which has been reserved by law or other effective means to protect part or all of the enclosed environment (Kelleher, 1999).

Minimum mapping unit. The size of the smallest unit that can be reliably detected or mapped.

Monitoring. The process of regular data collection and analysis that is then used by managers to determine whether project objectives are being met.

Monitoring concept worksheet. Strategic template worksheet used by managers and staff to identify the available resource base and appropriate scope of a proposed biodiversity monitoring system.

Monitoring cycle. Single monitoring interval that includes complete routine from securing field workers to data analysis and archiving.

National Biodiversity Strategy and Action Plan (NBSAP). Principal instrument for implementing the CBD at the national level. Signatories are required to prepare a national biodiversity strategy or equivalent instrument. NBSAPs provide important information on national targets and commitments and on the activities planned to achieve them (CBD, 2020).

Other effective area-based conservation measure (OECM). A geographically defined area other than a protected area, which is governed and managed in ways that achieve positive and sustained long-term outcomes for the in situ conservation of biodiversity with associated ecosystem functions and services and, where applicable, cultural, spiritual, socio-economic and other locally relevant values are also conserved (IUCN, 2022b).

Protected area. A clearly defined geographical space, recognised, dedicated and managed, through legal or other effective means, to achieve the long-term conservation of nature with associated ecosystem services and cultural values (Dudley, 2008).

Protected Area Management Effectiveness (PAME).

Assessment of biodiversity, social, cultural and economic outcomes as a result of protected area management. This evaluation also includes understanding the context of management, planning, inputs, processes, outputs and outcomes (Hockings, Leverington & Cook, 2015).

Proxy indicator. Something that might not be directly important but helps to get information about an indicator of interest.

Red List of Threatened Species™. Listing of the

conservation status of the world's flora and fauna administered by IUCN. The IUCN Red List of Threatened Species[™], known as the IUCN Red List, is the world's most comprehensive inventory of the global conservation status of plant and animal species. It uses a set of criteria to evaluate the extinction risk of species and subspecies. The IUCN Red List is recognised as the most authoritative guide on the status of biological diversity (IUCN, 2022b).

Simulated data. Realistic mock data that can be used during project development to assist developing a robust statistical design.

Stakeholder. Actor (such as but not limited to landowners) socially endowed with legal or customary rights with respect to land, water and natural resources who possesses direct or indirect interests and concerns about these resources but does not necessarily enjoy a legally or socially recognised entitlement to them (Borrini-Feyerabend et al., 2013).

Statement of purpose. Brief summary of the purpose of a biodiversity monitoring system or management programme.

Traditional ecological knowledge. Knowledge from Indigenous or local communities that plays a significant role in facilitating or discouraging collaboration between Indigenous and non-Indigenous stakeholders (Whyte, 2013).

Umbrella species. A species whose conservation is expected to confer protection to a large number of naturally co-occurring species (Roberge & Angelstam, 2004).

United Nations Educational, Scientific and Cultural Organization (UNESCO). Organisation that promotes international cooperation in education, sciences and culture. UNESCO's programmes contribute to the achievement of the Sustainable Development Goals defined in the 2030 Agenda (UNESCO, 2022).

World Heritage Site. Unique area of Outstanding Universal Value that requires long-term protection, is non-renewable and irreplaceable, as identified by UNESCO and the World Heritage Committee (Zhang et al., 2022).



Floodplain forest along the Isar River, Bavaria, Germany. © Vid Švara

CHAPTER 1. Introduction

1 Introduction

Biodiversity monitoring is the main tool to assess the state of biodiversity at a site. Long-term monitoring can show the effect of management actions and helps managers determine the outcome of conservation measures. It is also an approach supporting establishing baseline information, scientific research and regional documentation (Figure 2). This publication presents a consistent decision-making framework for designing biodiversity monitoring programmes that support biodiversity monitoring systems in protected areas and OECMs. It is beyond the scope of the guideline to consider monitoring for general scientific research or for establishing baseline inventories of habitats and species. Besides monitoring, additional features of a management programme will contribute to positive conservation outcomes, including governance and social engagement (Jungmeier et al., 2013). An IUCN Best Practices guideline is available that details the many components of governance that contribute to protected area management effectiveness (Borrini-Feyerabend et al., 2013).



Figure 2 Scope of the framework for monitoring biodiversity in protected areas and other effective area-based conservation measures (OECMs) Biodiversity monitoring accomplishes many potential goals. The focus of the quideline is on achieving management objectives through monitoring of biodiversity in protected areas and OECMs. Monitoring for baseline data, scientific value and regional documentation are beyond the scope of the guideline. Source: Compiled by the report authors

Protected areas are sites where the main management objective is the conservation of biodiversity or the environment. By contrast, OECMs are sites where management activities should deliver positive conservation results even though such activities may not be the main purpose of site management. For OECMs, the most important biodiversity values may need to be identified in advance of developing a site-level biodiversity monitoring system. A three-step assessment tool requiring fulfilment of eight criteria is available for managers to determine whether their site qualifies as an OECM. After all criteria are satisfied, the assessment is transferred to the governing authority, who then reports the site through the World Database on Protected Areas (Jonas et al., 2023). Protected areas and OECMs are both considered important for achieving the targets established by the Kunming-Montreal Global Biodiversity Framework (see chapter 6.1).

The decision-making framework for biodiversity monitoring includes four phases (see Figure 1). The preparatory phase is designed to help managers identify the requirements of their sites to accomplish management objectives. In the conceptual phase, managers and staff evaluate the scope of individual biodiversity monitoring programmes. The outcome is a monitoring strategy that is applied in the implementation phase. The re-evaluation phase helps decision-makers determine whether the findings of the monitoring programme effectively guide adaptive management procedures, concluding with a decision to continue, modify or terminate the monitoring programme.

Preparatory phase. Site management activities are determined by management plan goals, protected assets, actual or potential threats, the demands of different stakeholders, and from changing uses of the protected area or OECM and surrounding areas. Site-specific conditions will impact the scope of the biodiversity monitoring programmes that make up the biodiversity monitoring system. In many cases, requirements on indicators and protocols already exist and should first be considered, such as the objectives of a country's National Biodiversity Strategy and Action Plan that is required by the Convention on Biological Diversity (CBD). Legal obligations from other national or international policies may also exist. Additional site-specific background information will reveal any knowledge gaps that can be improved in later phases. The output of the preparatory phase is a clear programme-specific statement of purpose that identifies priority questions for monitoring. Documenting the approaches used to address a particular monitoring question will help other sites in the network implement effective protocols.

Conceptual phase. With clearly defined biodiversity goals from the statement of purpose, the next step is to document the questions and answers that guide the specific biodiversity monitoring programme. This step is the conceptual phase. The conceptual phase is assisted by the use of the monitoring concept worksheet, a tool that helps to focus the many considerations of biodiversity monitoring programmes through a series of six basic questions: 'why' establish the monitoring programme, 'what' will be monitored, 'where' and 'when' will monitoring occur, 'who' are the stakeholders involved, and what 'required resources' are needed. We provide a printable poster-sized version of the monitoring concept worksheet to help managers and staff make the necessary considerations (Annex Figure 1). Example considerations of each part of the monitoring concept worksheet are provided (Annex Figure 2). Questions can be considered in any order and can be revisited at any time. It is recommended to review all questions at least two times. After the questions have been considered, the framework of 'how' monitoring will be conducted and the synergies with other monitoring programmes will complete the monitoring concept worksheet. A key output of the conceptual phase will be to document supporting decisions on which tools and methods will be used in the next phase of the monitoring programme, implementation. Working through the six questions will identify how monitoring will be conducted and potential synergies with the larger management programme or national network.

Implementation phase. The actual monitoring occurs during the implementation phase of the monitoring programme. This phase involves specific steps for performing field work and data collection. The necessary materials should be acquired, and a preliminary manual describing the details of the field work should be produced. When finalised, field workers should be trained according to the manual. After training, at least one test run should occur at an easily accessible site. Any changes should be recorded in the manual, and training and test run cycles should be repeated until the workflow functions as planned. Data collection can then begin at field sites through monitoring cycles that are repeated at appropriate intervals, depending on the indicators. Statistical analysis of the collected data should occur periodically during the implementation phase. Presentation of the results should be given transparently and in appropriate formats to different stakeholder groups. Findings should support adaptive management decisions, provide the basis for PAME reporting, and can be valuable for outreach.

Re-evaluation phase. The fourth phase is the re-evaluation phase. The timing of this phase is typically defined through funding cycles or reporting requirements to comply with biodiversity treaties. Findings are given to decision-makers to identify strengths and weaknesses of the biodiversity monitoring programme. Effective components can be transferred to other site monitoring programmes, or within national and regional biodiversity monitoring networks. Sharing findings can maximise synergies with other programmes because successful elements may serve as templates for further use in other protected areas and OECMs. The outcome of the re-evaluation phase is the decision to continue, adjust or terminate the monitoring activities.

Many general considerations are relevant to developing a biodiversity monitoring system and its many biodiversity monitoring programmes. The following points are applicable to most situations:

• Obligations: International conventions and policies (see chapter 6.1)

Biodiversity monitoring programmes in protected areas and OECMs should be guided by international agreements and must contribute in a meaningful way to national reporting requirements.

• Art of omission: Daring to simplify (see chapter 6.2)

To obtain high-quality data despite potential budgetary or resource restrictions, simplifying monitoring to the lowest number of indicators and technologies possible should be a key feature of a biodiversity monitoring programme.

• Biodiversity monitoring systems: Designing modular, multi-scale and multi-purpose monitoring systems (see chapter 6.3)

Biodiversity monitoring systems should utilise known effective methods and synergies to guarantee that the results are comparable between sites.

• Combining forms of knowledge (see chapter 6.4)

Indigenous and traditional ecological knowledge should be utilised in combination with scientific data for the best understanding of the state of biodiversity in the protected area or OECM.

• Continuity risks: Avoiding disruptions and gaps in data (see chapter 6.5)

Data are most valuable when they are supported by an appropriate statistical design.

• Detecting trends and correlations: The value of time series (see chapter 6.6)

Well-timed and regular collection of data provides strong evidence of trends of the selected indicators.

• Maintaining ecological balance: Establishing baselines and thresholds (see chapter 6.7)

The baseline condition of an indicator is required knowledge to develop meaningful thresholds of change, at which point management activities may be necessary for adaptive management.

• Setting up monitoring systems: Costs and outcomes (see chapter 6.8)

Biodiversity monitoring programme costs are highest at the beginning of the programme, whilst knowledge gain occurs in later cycles.

• Protected Area Management Effectiveness evaluation tools (see chapter 6.9)

Protected area management programmes are evaluated using PAME evaluation tools that are harmonised across protected area networks or National Biodiversity Strategies and Action Plans.

The processes to develop effective biodiversity monitoring programmes are outlined in the following chapters. The **preparatory phase is introduced in chapter 2**, followed by a detailed breakdown of the **conceptual phase in chapter 3**. The **implementation phase is addressed in chapter 4**. Key components of ongoing monitoring cycles are illustrated, with particular emphasis on data management and feedback to facilitate adaptive management (Caughlan & Oakley, 2001). Considerations of the **re-evaluation phase are addressed in chapter 5**. In re-evaluating the programme, reflection is given on appropriate ways to guide future monitoring cycles. In **chapter 6, greater detail is provided on the general considerations listed above. In chapter 7, a brief review of past and current tools for biodiversity monitoring is provided.** The guideline concludes in **chapter 8 with a synthesis and future outlook** of biodiversity monitoring in protected areas and OECMs.

With this road map, it is now time to develop site-specific biodiversity monitoring programmes that will contribute to meeting the overall protected area or OECM management objectives.



Stand of cacti surrounding epiphyte-covered tree *Eriotheca ruizii*, Refugio de Vida Silvestre Laquipampa, Lambayeque, Peru. © *Tobias Fremout*

CHAPTER 2.

Preparatory phase: Setting the frame

2 Preparatory phase

Setting up a biodiversity monitoring programme is a strategic decision-making process that will have long-term effects on protected area or OECM site management. Therefore, enough time should be spent in the preparatory phase to make sure that the programme is properly conceptualised. Site-specific goals and main conservation challenges should be reviewed through a basic site investigation if they are not present in the pre-existing management plan.

In the preparatory phase, all background materials on management of the protected area or OECM should be collected and analysed for their relevance in the subsequent phases. This will indicate the major site threats – and where they are most serious – prior to developing the supporting biodiversity monitoring programmes. Decisions affecting the design of a programme will reference the materials gathered in the preparatory phase. The key output of the preparatory phase is to produce a monitoring statement of purpose to identify and prioritise the main objectives of each monitoring programme in the biodiversity monitoring system (Figure 3).



Figure 3 Ontology of the preparatory phase Development of effective biodiversity monitoring programmes requires consideration of management objectives as components of the larger management plan. The previous establishment of a protected area or other effective area-based conservation measure (OECM) is the starting point for filling any knowledge gaps through a basic site investigation. Managers should become familiar with relevant outstanding features, ecological characteristics, threats and site conservation goals. This includes identifying frameworks such as the cycle of drivers, pressures, state, impacts and responses (see Figure 4). With all available information, a list of priority monitoring objectives will be described in a monitoring statement of purpose. Source: Compiled by the report authors

2.1 Conventions and standards

Many international conventions and national programmes have reporting obligations on species or habitats of particular value. These requirements should be the first considerations for developing or revising a biodiversity monitoring programme. Sources for this information include formal paperwork, previously surveyed indicators, appropriate monitoring intervals and legal agreements. The information will establish a binding basis for monitoring. To compensate for incomplete information, administrators may consult pre-existing standards and frameworks applicable to similar sites as a guide to develop site-specific approaches. This step is only necessary if the information is not already available in the protected area or OECM management plan.

2.2 Basic investigation and site assessment

The basic requirement for establishing a biodiversity monitoring programme is good knowledge of the baseline conditions of the protected area or OECM. Existing data, documents, Indigenous and traditional ecological knowledge, and descriptions should be assessed early on to identify the most important biodiversity assets and indicators of the site. If available, assessment should be based on the IUCN Red List of Threatened Species[™] (IUCN, 2022c) for species that have historically lived at the site. Data from previous programmes, national and regional catalogues, or other institutions may be helpful. For example, data of climatological or hydrological relevance may improve understanding of the pre-existing site conditions. Depending on the site, cultural or natural heritage considerations may also be important. Land use history may be revealed through examining archive satellite data, historical maps and other past records. Basic investigation and site assessment are only necessary if a management plan does not already provide this information.

2.3 Specifying conservation objectives

Desired conservation outcomes are usually listed either in the text of the nomination or in the legal document establishing the protected area or OECM. However, based on national and international standards and obligations, managers must have a comprehensive site overview of key species and habitats and their protection status. Indicators may be of biological, geological, ecological or cultural significance. Species monitoring can track abundance, distribution, diversity, conservation status and biological characteristics, as described by the Essential Biodiversity Variables developed by GEO BON (Box 2). A site will have elevated conservation responsibility for rare or endangered species and habitats if it meets the criteria for Key Biodiversity Area designation (IUCN, 2016). Comprehensive site planning is beyond the scope of regular management activities. If the management plan does not provide a detailed description of the conservation objectives, administrators should invest sufficient resources for comprehensive site planning, at least to the extent that is reasonably achievable.

Box 2

A global concept: Understanding indicators through the Essential Biodiversity Variables



By definition, biodiversity includes genes, species, traits, community composition and ecosystems. Data on one or more of these dimensions over time and space support biodiversity assessments. Information on how biodiversity changes in these environments is necessary for policy-making. In order to detect change, systematic biodiversity observations are collected using standard formats and methods, together with environmental monitoring. These observational data are moved to open databases. Ensuring that data are interoperable across databases will make efficient use of biodiversity information for guiding conservation and sustainable development strategies. Essential variables to understand climate, biodiversity and other environmental changes have already been developed (e.g. Essential Climate Variables, Essential Ocean Variables). The concept of Essential Biodiversity Variables (EBVs) was introduced to advance the collection, sharing and use of biodiversity information (Navarro et al., 2017; Pereira et al., 2013), providing a way to aggregate the many biodiversity observations collected through different methods such as in situ monitoring or remote sensing. EBVs can be visualised as biodiversity observations at one location over time, or in many locations, aggregated in a time series of maps.

Aggregation requires collecting biodiversity observations by people and groups, depositing raw data into databases using standard formats and metadata, and processing the data. This information helps to detect and model biodiversity change for science, policy and sustainable development applications. Completing the whole procedure is important for protected area and OECM management activities because the analysis shows changes in biodiversity across large spatial scales. The underlying drivers and pressures of biodiversity change can then be identified (Mace & Baillie, 2007) and modelled (Oliver et al., 2015). Validation of modelling can then feed into global and regional policy processes to explain observations, to improve forecasting of biodiversity change and to produce global assessment reports. EBVs are scalable, meaning the underlying observations can be used to represent different spatial or temporal resolutions required for the analysis of trends. For example, ecological community data collected at a location from different sampling events or methods can be combined into a single time series. The aggregated data may indicate the change in ecological communities across the region.

When combined with social or economic information from human or environmental pressures, EBVs can be used to identify indicators for biodiversity that reflect responses, for example change in the proportion of habitat in protected areas and ecosystem service benefits to humans. Essential Ecosystem Service Variables have been defined as a type of EBV to support the monitoring of ecosystem services (Balvanera et al., 2022).

Developing and applying EBVs requires local, national and international adoption of standard approaches to collect, store and share biodiversity and environmental observations. This is fundamental to address the pressing societal and economic needs of today and in the future.

Source: GEO BON, Canada. Picture: Animal collage. © Gernot Kunz

2.4 Management planning

Biodiversity monitoring is just one of many components allowing effective site-based management (Stephenson, 2019). Information from monitoring campaigns will help guide adaptive management decisions, resulting in improved conservation outcomes. An effective management plan must address social, cultural, economic and ecological factors (Jungmeier et al., 2013). These factors are best considered through including the perspectives of the full range of regional stakeholders. This involves understanding how burdens and benefits are distributed amongst local communities, clearly defining the management decision-making processes, and recognising the cultural identities of local and Indigenous groups (Zafra-Calvo et al., 2017). A protected area or OECM management plan is most effective when it includes collaboration with stakeholders to achieve goals of common interest (Karadeniz & Yenilmez Arpa, 2022b). A smart management plan will not only conserve biodiversity, but will also encourage local engagement. By including diverse values in management planning, governance of protected areas and OECMs can become transformative. This approach is necessary for sustainable management (Kelemen et al., 2023).

A well-considered management plan will involve Indigenous peoples and local communities in all phases of development and will include legal obligations, site characteristics, and local and regional conservation objectives. A management plan should already be in place in advance of establishing a new biodiversity monitoring programme (Box 3). It is important to have a clear picture of the main pressures, impacts and interactions on site-level biodiversity (Figure 4). Understanding these factors will guide management activities to address specific goals. This information will expedite the conceptual phase (see chapter 3). If some information is missing, additional assessment will be required.



Figure 4 The framework cycle of drivers, pressures, state, impacts and responses is the main tool for managers to determine overall management effectiveness at their sites. This example shows bush encroachment in a savanna ecosystem (see Box 3). Source: Compiled by the report authors

Box 3

Bush encroachment in an African savanna: An application of the framework cycle of drivers, pressures, state, impacts and responses in Kruger National Park



Since the pre-industrial era, global mean temperatures have increased by 0.8°C to 1.2°C. This global temperature increase is driven by human activity, and effects on ecosystems are already visible globally (Sala et al., 2000). The world has experienced increased frequency of extreme weather events, prolonged droughts, and heat. Global temperatures will continue to rise, with projected mean temperatures from 1.0°C to 3.5°C warmer by 2100 than the pre-industrial era (IPCC, 2021). Second only to land use change, climate change will have the greatest impact on biodiversity this century (Sala et al., 2000). Species across most ecosystems have already undergone range shifts and phenological changes (IPCC, 2022). Climate change has resulted in losses of critical habitat. In terrestrial ecosystems, warming of 1.5°C is expected to drive up to 14 per cent of species to extinction, with greater percentages at more extreme warming scenarios (IPCC, 2022). This represents a critical loss of biodiversity, even excluding direct impacts of human activities (Lenton et al., 2019).

Interactions between the biodiversity and climate crises are exemplified through ongoing bush encroachment in the savannas of Kruger National Park, South Africa. A major **driver** of bush encroachment is the globally high usage of fossil fuels. The increased atmospheric carbon dioxide (CO₂) levels are one of the pressures causing bush encroachment through CO, fertilisation that increases tree growth (Buitenwerf et al., 2012; Stevens et al., 2016). Large parts of the park are in a state of transition from a more open system characteristic of savannas to a more closed system. The resulting impacts include decreased grass abundance and biodiversity across the landscape, a lower carrying capacity for grazing animals, greater tree biomass, and lower fire frequency. In closed canopy systems many species become excluded, including cheetahs that need space to reach the speed to pursue prey, and vultures that need a certain level of open canopy to take flight (Bamford, Monadjem & Hardy, 2009). A possible management response includes applying targeted high-intensity controlled burning in some areas, allowing grasses to dominate in some regions and woody shrubs with old growth trees in others (Smit et al., 2016).

The role of protected areas and OECMs to mitigate the pressures caused by human activities is more important than ever. These pressures increase the vulnerability of ecosystems to climate change effects. More land must be conserved to allow species to respond to climate change, including movement of species through natural corridors in response to habitat loss or changes in climatic suitability. Managers must also develop stewardship plans to cope with climate change. Currently, protected area networks do not have enough coverage to ensure that species and ecosystems are resilient to human impacts and climate change. Conserving 30 per cent of land and limiting warming to 2°C could reduce species extinction risks by 50 per cent compared to a scenario where there is no global increase in conserved areas or control of climate change (Hannah et al., 2020).

Source: Kerry Grey, Stellenbosch University, South Africa. Picture: A closing canopy favours certain savanna species at the expense of others. © Clive Kaiser

2.5 Output: Statement of purpose

A statement of purpose should be produced as a brief explanation on the purpose and intent of the biodiversity monitoring programme. It should be focused on a narrow area of interest and should reference a prioritised list of indicator species or habitats. The statement of purpose should be in line with the overall vision and goals of the management plan and will serve to communicate the objectives of monitoring to employees, local residents, visitors and responsible authorities (Box 4). For clear direction, each monitoring programme should have its own statement of purpose, complete with relevant modular components. The statement of purpose will constitute the first element of the monitoring concept worksheet.

Box 4

South Georgia and South Sandwich Islands Marine Protected Area: Conservation objectives of a Marine Protected Area

The Southern Ocean is home to the South Georgia and South Sandwich Islands Marine Protected Area. This protected area was declared in 2012 and serves as a key research site to understand marine ecosystems. The islands are home to millions of seals and birds, whilst the surrounding waters are important for migratory whale species (Government of South Georgia & the South Sandwich Islands, 2016). The islands are uninhabited by people but face many pressures including climate change, tourism, fishing and introduction of invasive species.

The main objective of this protected area is to conserve marine biodiversity – with many specific objectives outlined for each of its management zones – and is supported by a comprehensive research and monitoring plan (Government of South Georgia & the South Sandwich Islands, 2021). The plan serves as a framework under which scientific research can be conducted given available funding and resources. Management activities are categorised under 10 research themes. Each theme has a clear statement of purpose with comprehensive lists of monitoring activities, research needs and relevant projects. The frequency of monitoring is prescribed for many activities. These themes thus represent the biodiversity monitoring system of the South Georgia and South Sandwich Islands Marine Protected Area.

Objecti Research theme	ର Conserve biodiversity, habitats, ecosystems	Resilience to climate change	Sustainable fisheries management	Protect from trawling	Protect predators	Protect unique feature or area	Manage human activities	Prioritised monitoring activities	Prioritised research needs
Oceanography and biogeochemistry	Х	Х						yes	yes
Pelagic ecosystems	Х	Х						yes	yes
Higher predators	Х				Х	Х		yes	yes
Benthic ecosystems	Х		Х	Х		Х			yes
Harvested fish	Х		Х			Х		yes	yes
Harvested krill			Х		Х			yes	yes
Impact of fisheries – benthic habitats			Х	Х		Х		yes	yes
Impact of fisheries – predator interactions	Х		Х		Х	Х		yes	yes
Climate change		Х						yes	yes
Other human impacts							X		

Source: Government of South Georgia & the South Sandwich Islands, 2021, adapted by the report authors

CHAPTER 3.

Conceptual phase: Designing the monitoring concept worksheet

3 Conceptual phase

After the overall goals and purposes of the biodiversity monitoring programme are defined, the conceptual phase begins (Figure 5). The monitoring concept worksheet is a tool to help develop the logic of a new programme (see Annex Figure 1). It is designed to identify and structure its primary objectives within the larger protected area or OECM management plan. It can be completed together with partners, external experts and stakeholders. The first element of the monitoring concept worksheet is the statement of purpose that was developed at the conclusion of the preparatory phase (see chapter 2.5). The following elements make up the majority of the monitoring concept worksheet, addressing the questions: Why prepare the monitoring programme? What will be monitored? Where will monitoring take place? When will monitoring be conducted? Who will be involved in the monitoring programme? and What are the required resources to conduct monitoring? The questions are designed to be answered in an iterative way. Most importantly, the answers to the questions should interlink with one another. Different versions of the monitoring programme - for example different cost frames, areas of interest or technological approaches - can be proposed for internal debate. Working through the monitoring concept worksheet will help justify the need for monitoring in the context of management objectives. The monitoring concept worksheet is a key resource to draft a field manual for implementation of the monitoring programme (see chapter 4.3), including evaluation of synergies that can be built into the larger biodiversity monitoring system and national networks.



Figure 5 Ontology of the conceptual phase Starting with the statement of purpose of the preparatory phase, the conceptual phase is completed through consideration of management goals, site characteristics and the human and financial resources available to the biodiversity monitoring programme. The analysis will show the optimal scope of the programme. Source: Compiled by the report authors

3.1 Why: The purpose of monitoring

When answering the question of 'why', the manager should specify the purpose of the results for site management. By doing so, different uses can be targeted, for example whether the results will be used locally, regionally, nationally or internationally. A point allocation system can help to prioritise the most important purposes for the biodiversity monitoring programme (Annex Table 1).

Helping to determine management effectiveness is one of the main points of a biodiversity monitoring programme (Hockings et al., 2008). First, a determination is made on what type of monitoring should be conducted, for example, as a baseline study, as a research project, for regional documentation or for management purposes (see Box 1). This in turn supports the development of appropriate management activities. Targeted monitoring informs managers and decision-makers on whether the desired outcomes have been achieved, guiding adaptive management. Different audiences should receive descriptions of site management in diverse contexts. Monitoring results can therefore be used for outreach, educational and instructional purposes.

3.2 What: Indicators in a biodiversity monitoring programme

A biodiversity monitoring programme typically provides information about biological indicators. An indicator is the entity that a monitoring programme works with: the object that is recorded, measured and documented in a time series (Figure 6). An indicator should be sensitive to change, characteristic for the site, and as easy as possible to sample or determine. In some cases, it may be effective and useful to monitor biological indicators along with meteorological, hydrological, physical or other abiotic indicators if they are easier to track. Ecological variables such as habitat size or quality may also be useful to estimate species populations. If the indicator is not available or hard to track, a substitute of abiotic or ecological variables can be used, so-called proxies. The main challenge of using a proxy is to ensure that it accurately represents the state of the conservation target (Table 1). Pressures such as land use change can also be monitored as proxy indicators for the biological community because certain species depend on the quality of their habitat for survival (Harris et al., 2021). In addition, there is the possibility to use aggregated or sum indicators such as the degree of human influence in an ecosystem, or ecosystem services (Grabherr et al., 1998).



Figure 6 Indicator selection will be sitespecific and based on biodiversity monitoring system objectives. The conservation target may be a biological indicator, an abiotic indicator or an aggregated indicator combining environmental information with ecological properties. *Source: Compiled by the report authors*
Table 1 Examples of proxy indicators for biodiversity monitoring

Some biotic indicators are difficult to monitor due to the cryptic nature of their life cycle, dependencies on abiotic factors or logistical reasons. These biological features can be tracked through monitoring closely associated proxy indicators.

Conservation target	Challenge of monitoring conservation target	Proxy indicator
<i>Rosalia alpina</i> , Alpine longhorn beetle: endangered species	Larvae live in old partially dead <i>Fagus sylvatica</i> (beech trees), the limiting ecosystem factor.	Dead or dying beech wood in large-scale surveys to deduce beetle conservation status
Calcareous fen containing <i>Cladium ma- riscus</i> , swamp sawgrass: priority habitat	The favourable conservation status of the habitat depends on the range of fluctuation of the water level.	Fluctuations of the water level can be measured with a data logger
Habitat that is difficult to access or reach	Survey of habitat is laborious or hazardous.	Remote sensing data for habitat-based metrics

Source: Compiled by the report authors

Because life is so diverse, monitoring must focus on the minimum number of indicators or else the programme will face resource limitations (Box 5). This reduction is a key process. A scorecard approach assumes that the priority for monitoring is determined by two attributes (Figure 7). First, the status of a key species or habitat is addressed through observation of an appropriate indicator (1st ranking). The second criterion relates to the difficulty in managing the species or habitat on-site (2nd ranking). The more management activities and resources that are needed to gain a favourable conservation status of key species or habitats, the more important it is to monitor change in the indicator conservation status. The next step is to check whether the indicator can be observed with reasonable effort. Effective management may involve taking note of what is happening in the surrounding environment because major changes may be occurring that do not affect the current indicator group. The monitoring programme should be designed to include additional indicators at a later time, if necessary. The decision on what to monitor will be strongly influenced by a third attribute, the resources required to conduct the monitoring programme (see chapter 3.6). This illustrates why it is important for staff to examine all details of the monitoring concept worksheet and revisit all sections after a first comprehensive discussion.



Figure 7 Selection of key indicators is ideally based on two attributes. The first component is the importance of the species or habitat to meet site management objectives. The second component is how challenging the species or habitat is to manage. In reality. selection of indicators may be limited by the resources available to the monitoring programme. Source: Compiled by the report authors

Box 5

Indicator selection: Biosphere Reserve Integrated Monitoring in the Nock Mountains of Austria



The UNESCO Man and the Biosphere (MAB) programme describes the concept of a biosphere reserve. All biosphere reserves should support three main functions: conservation; development; and logistical support (UNE-SCO, 2020).

The Salzburger Lungau & Kärntner Nockberge Biosphere Reserve was established in Austria in 2012 following a long participatory planning process. As a condition for its establishment, stakeholders requested that an integrated monitoring scheme already be in place to evaluate management activities, the so-called Biosphere Reserve Integrated Monitoring (BRIM) approach. For establishing BRIM^{Nockberge}, four pillars of monitoring were addressed: social, economic, ecological and management effectiveness. Within each pillar, indicators were selected based on their relevance, their availability, their sensitivity to management action, and consistency (Jungmeier et al., 2013). Using these criteria, a narrow set of 12 indicators was chosen out of more than 100 proposed indicators. The 12 indicators are evaluated annually through field monitoring, questionnaires or externally. Evaluation can be simply summarised through marking an arrow on the evaluation form indicating the direction of change of the indicator.

Ecological indicators represent the condition of the mountain environment, as measured by an umbrella species representing the animal community, and two additional ecological indicators. Economic indicators involve statistical analysis of local and visitor taxes, as well as description of local agricultural activities. Socio-cultural indicators are based on active stakeholder participation and the migration balance of regional inhabitants. Management indicators are based on the number of visitors to sanctioned events, number of press reports, and the number of research projects that occur in the biosphere reserve.

By using the short list of 12 indicators, evaluated annually, biosphere reserve managers can track the effectiveness of management activities. The BRIM^{Nockberge} approach was driven by stakeholder input, fulfilling the key aspects of the UNESCO MAB programme and serving as a model of a successful biosphere reserve integrated management approach.

Source: Michael Jungmeier, Carinthia University of Applied Sciences. Picture: View from the Nock Mountains. © Michael Jungmeier

3.3 Where: Scale of spatial features

As part of developing an effective biodiversity monitoring programme, the area of interest, sampling approaches and statistical analysis should be considered (Figure 8). Depending on the indicator, the area of interest can vary from the plot level $(mm^2 - m^2)$ up to the site level (several hectares) to the habitat scale and beyond (Figure 9). To demonstrate the impact of area-based management practices, it may be necessary to monitor beyond the area of interest, allowing comparisons. Different monitoring approaches and statistical tests will result in choosing the appropriate target organism. For example, monitoring the behaviours of mobile or migratory bird species will require a different spatial approach than monitoring the genetic diversity of fish species in lentic waters.



Figure 8 Spatial considerations of monitoring include area of interest, sampling format and statistical design. The area of interest may be the entire protected area or other effective area-based conservation measure (OECM), a portion of it, or it may extend beyond the site boundaries. The biological characteristics of the indicator determine the selection of the best sampling scheme, and the statistical design may depend on the objectives of the management plan. Source: Compiled by the report authors



spatial scale

Decisions about the sampling design must account for area-based features such as the spatial distribution of the indicator. Species and habitats are unevenly distributed across the landscape. Elevation, exposure, water regime, water depth and nutrient availability are some factors that restrict the distribution of species or habitats. Geomorphological features or coarse vegetation types – as revealed by remote sensing or satellite imagery (see chapter 7.3.3) – can often be correlated with key species or habitats in the area of interest where indicators are most likely present. It is important to perform some sampling outside the expected areas to confirm whether selected features match the expected distribution of the target. If necessary, voucher specimens can be collected outside a plot to minimise any disruption of the monitoring sites. Permissions must be secured prior to removal of biological resources from field sites.

Figure 9 The spatial scale of a biodiversity monitoring programme can range from a genetic to landscape level. This chart illustrates the spatial range of indicators that can be monitored in a monitoring programme. The box indicates the typical range of most monitoring programmes. *Source: Compiled by the report authors* The better the understanding of the ecological relationship between the indicator and its environment, the more accurate the prediction of its spatial distribution will be. Knowledge of the distribution and behavioural patterns of the indicator will help determine the minimum mapping unit or spatial scale that can be reliably analysed. This accuracy will help to reduce sample numbers. The selection and distribution of plot, transect or point designs can be random, stratified based on previous findings, or systematic (Elzinga, Salzer & Willoughby, 2019; Magurran, 2004) (see chapter 4.1). Using a polygon-based approach is of special importance for the survey of vegetation or land cover. Likewise, it is possible to set up a systematic area-based survey using a grid-based approach. In this case, the indicators are surveyed in plots along the grid. When comparing features across sites, the sampling units should be standardised and spatially independent for statistical purposes. When possible, plot design should be consistent across sites, particularly when they are part of the same national monitoring network.

3.4 When: Scale of temporal features

Because monitoring is an investigation over time, the temporal design of a monitoring programme is very important and must be defined in the context of the larger biodiversity monitoring system. Monitoring can take place over very different timescales. To answer the question on when to monitor, three key principles should be considered: timing of programme initiation; programme duration; and the interval between monitoring events (Figure 10).



Initiation timing of a monitoring programme must be determined in the conceptual phase. Monitoring can begin as a response to an ecosystem disturbance such as a catastrophic storm, coral bleaching event or flood (Obura et al., 2019). Starting a monitoring programme may occur before restoration efforts or new management strategies begin. This timing will document the baseline condition of the indicator, allowing before-and-after comparisons. Potential considerations of site accessibility may further guide initiation timing, for example if seasonality, potential non-target effects or cultural considerations might affect access to key sites.

The duration of a time series must be long enough to identify trends or changes in the status of the indicator as a response to management actions. If a monitoring programme is designed for a short-term timescale, it can produce valid results after only a few monitoring cycles (e.g. 3 to 5 cycles). In this case a predetermined end date can be set. Monitoring can also occur over medium-term (e.g. 5 to 25 years), long-term and indefinite time spans, depending on the management question and the variability of the indicator.

Trends of an indicator can only be accurately observed following many monitoring cycles. For example, one way for a species to be classified as 'Threatened' on the IUCN Red List is for its population to be in decline for at least 10 years or three generations, whichever is longer (IUCN Standards and Petitions Committee, 2022). For animal population studies, the minimum

duration of a monitoring programme depends on the species but in general must continue for a minimum of 16 years to confirm trends (White, 2019). In a practical sense, the timeframe of a monitoring programme will more likely be based on the available resources, rather than the biology of the indicator. Innovative programme funding – for example through foundation funding or tax measures – may support indefinite long-term monitoring of key species (Lindenmayer et al., 2012a). As required by the CBD, National Biodiversity Strategies and Action Plans may provide a strong justification to administrators for long-term, sufficient funding to maintain biodiversity monitoring programmes in protected areas and OECMs.

The third factor concerning timing of monitoring is the interval of the monitoring activity. The interval will depend on the phenological activity of the indicator and its variability over time (Figure 11). Certain indicators have short lifespans or are only briefly present in the ecosystem, whilst other indicators have a long-term presence. If possible, the beginning of a monitoring programme should feature frequent monitoring intervals, with reduced frequency at a later stage once trends and variability are known. More frequent intervals early in the programme allow managers to react quickly to any methodological or technical errors. Not all indicators will be appropriately surveyed at the same interval. Therefore, a modular approach should be applied in the case of monitoring multiple indicators. Biodiversity monitoring may focus on trends of species occurrence, distribution and dispersal, as well as on their drivers such as climate change or fragmentation of habitats. These features are generally visible over long periods of time.



timescale

Long-term, consistent and well-documented biodiversity monitoring programmes are the 'gold standard' because the value of monitoring accumulates over time. Long-term monitoring may continue for decades. This scope goes beyond a professional career and even the lifespan of institutions; therefore, turnover in personnel, institutional, legal or financial situations is guaranteed. To ensure continuity, these limitations require consideration prior to implementation of the biodiversity monitoring programme. Complete records of monitoring protocols and objectives are necessary to maintain continuity despite these changes over time.

Figure 11 Timescale of different biological processes Depending on the species or habitat, a biodiversity monitoring programme should occur at the relevant timescale of the indicator. Some species may have rapid life cycles that require closely spaced monitoring intervals, whilst other species may be present in the environment for decades or centuries. The box indicates the typical temporal frame of most indicators. Source: Compiled by the report authors

3.5 Who: Identifying actors and stakeholders

This step identifies the core team responsible for implementing the monitoring procedures, supporting staff, active contributors of ideas and resources, and stakeholders who should be informed about the processes and results (Figure 12). The core team will be directly involved in making on-site decisions. It may include the managers of the protected area or OECM, personnel who lead and conduct the field work, directly involved Indigenous community representatives, and local expert consultants.

4 Stakeholders to be informed	1 Core monitoring team, e.g.
3 Stakeholders to be involved	 Staff experts External experts Indigenous representatives
2 Supporting staff	2 Supporting staff, e.g.
1 Core monitoring team	External experts Citizen scientists External service providers
	3 Stakeholders to be involved, e.g.
	Associations
	Union groups
	4 Stakeholders to be informed, e.g.
	Landowners
	Visitors Policymakors

The actors involved in the monitoring activity will affect the scope and quality of data collection. Institutional help and external collaborators can be identified at this early stage. Permanent staff positions should be secured for monitoring. Supporting staff may include scientists, statisticians, GIS experts, IT experts, database managers, field crew leaders and other specialists. Early and ongoing co-design of the monitoring programme between involved scientists, governmental agencies, non-governmental organisations, and Indigenous peoples and local communities is essential. Indigenous peoples and local communities should be actively engaged in all stages of the design of a monitoring programme, from conceptualisation and field work to logistical support (Dyck, LeClair & Bockstael, 2022). Involving these key stakeholders at all steps of programme development will incorporate valuable traditional ecological knowledge, providing essential scientific and social support for the programme. A primary role of managers is to then translate traditional ecological knowledge and scientific findings into recommendations for policymakers (Thompson, Lantz & Ban, 2020). More generally, it is important to include people and interests who are sceptical about the management programme because conflicts or even vandalism can be prevented through involving and informing the full spectrum of potential stakeholders (Hodgkinson & Young, 2016). A blank stakeholder worksheet is provided to help managers determine the level of involvement of different stakeholder groups (Annex Figure 3).

The responsibilities of monitoring should be assigned to professionals and experts. Depending on programme objectives, some aspects of monitoring can be accomplished by volunteers including students, nature enthusiasts and citizen scientists. The participation of volunteers in monitoring requires a series of support measures for effective training; however, involving non-scientists has many benefits ranging from scientific output to education to programme visibility (Box 6).

Figure 12 Example stakeholder worksheet listing key participants in a biodiversity monitoring programme Managers and staff should work together to identify potential stakeholders for an overview of who may be involved in the monitoring programme and the scope of their participation. *Source: Compiled by the report authors*

Box 6

Integrating citizen science to estimate the Slovenian brown bear population: It counts to be involved



Brown bear (*Ursus arctos*) populations were nearly exterminated in western Europe following centuries of intolerance, but large populations remained in parts of eastern and central Europe. Populations from south-eastern Europe are a source for reintroduction efforts elsewhere on the continent. In the Dinaric Mountains of Slovenia, cull quotas increased dramatically beginning in the early 2000s, supported by high official population size estimates and an increased number of recorded human-bear conflicts. Population estimates lacked a scientific basis and varied widely, igniting a debate on how many bears actually lived in the region, rather than focusing on effective management to facilitate co-existence.

To provide a robust, scientific estimate of the brown bear population size, a mark-recapture study utilised non-invasive genetic sampling (Skrbinšek et al., 2019). Bear scat was sampled and genotyped to identify individuals. Each genotyped specimen was considered a 'capture' of an individual, providing a basis to model population abundance. This design required high-intensity sampling on a large spatial scale. Considering limited funding, a sampling strategy was conceived using a citizen-science network of volunteer stakeholders, mainly hunters. Recruitment of volunteers was possible due to the flagship status of the brown bear and was supported by the very good organisation of Slovenian hunters. The project was advertised through different media, including a Slovenian hunting magazine which allowed contact with the local hunting community. The main guiding principles were for the scientists to maintain a professional appearance

so that people would take the project seriously, and that participation would be as simple as possible for all volunteers. Organised workshops provided well-designed sampling materials and training on how to collect bear scat for genetic analysis. The scientists took care of material distribution, handling of samples, and kept in contact with representatives of all hunting clubs. In total, more than 1,000 volunteers collected 1,057 samples over a three-month period covering an area of approximately 6,000 km². At the project's end, all results were published in the hunting magazine with sincere acknowledgement of the volunteer contributions.

Findings indicated a Slovenian brown bear population of 380–460 bears providing the first robust, scientifically based population estimate. The estimate stopped the endless debates about the number of bears and opened the more important discussion about managing humanbear conflicts. It has had a profound and continuing impact on brown bear management in Slovenia.

A well-designed study, combined with a large citizen scientist volunteer force, can accomplish impressive results. Motivation is the key factor, and the rewards for participation must exceed the effort. After all, volunteers don't work for free, they are just not paid money.

Source: Tomaž Skrbinšek, DivjaLabs, University of Ljubljana. Picture: Brown bear family. © Miha Krofel. Flickr

3.6 Required resources: Identifying the resource pool

A biodiversity monitoring programme will be successfully implemented only with the support of an adequate resource base (Figure 13). First, a realistic estimation of financial costs should be made, considering the difference in costs between the establishment phase of the programme and continuation in subsequent monitoring cycles. Generally, the establishment phase will require the greatest financial resources (Lindenmayer et al., 2012b). Furthermore, an estimated 25–30 per cent of the programme budget should be provided for data management, assessment and reporting (Caughlan & Oakley, 2001). A simple worksheet is provided to help to determine the amount of financial resources required for sufficient data collection (Annex Table 2). The second component is the demand for human resources. If there is a mismatch in the availability of resources, an adaptation of the monitoring objectives, indicators or methods will be required. Identifying synergies among funding sources or pooling resources from multiple programmes could increase the resources that are available for implementation of the monitoring programme. In some cases, the manager might be able to access extra resources for a critical procedure, or find innovative ways to acquire resources. These resources could come in the form of in-kind support from local communities, schools or research institutions.



monitoring programme depends on the available staff, resources and knowledge. Two categories of resources must be considered. Financial resources and the available management infrastructure are required for an operational management base. Specialised human resources including technical skills, logistical support and field work are also necessary to carry out an efficient monitoring programme. Source: Compiled by the report authors

Figure 13 A biodiversity

3.7 Output: Defined scope of the biodiversity monitoring programme

Examining the six questions of 'why', 'what', 'where', 'when', 'who' and 'required resources' will complete the conceptual phase. The necessary research questions for establishing the biodiversity monitoring programme should now be identified, but they will not be worked out in detail. The balance of the monitoring concept worksheet will be completed in the following phases by outlining details on how the monitoring activities will be performed in practice, and through identifying potential synergies of the programme with other elements of the protected area or other effective area-based conservation measure management plan. The implementation phase, described next in chapter 4, will help the monitoring team to decide on the specific methodologies that will be used in field monitoring.

CHAPTER 4.

Implementation phase: Setting up the infrastructure

4 Implementation phase

The implementation phase addresses how monitoring will be conducted (Figure 14). Ethical and cultural topics must be considered from the beginning, and potential legal constraints need to be resolved (Checklist 1). At the start of the implementation phase, the sampling design and methods of monitoring will be determined, followed by the acquisition and configuration of tools. Training of field technicians should include all steps planned for the actual monitoring, from setting up monitoring tools, collecting, storing and analysing data, and finally, presenting the findings. Presentation is a key step for communicating with decision-makers and local stakeholders. It shows how the monitoring programme contributes to conservation of biodiversity so that it is broadly supported both socially and through policy decisions such as resource allocation. Protocols should be documented in a detailed preliminary manual.



Before workflows, tools and methods are implemented on a large scale, they must be tested. Test runs should be performed to ensure proper implementation of all steps in the preliminary manual (see chapter 4.4). Test runs are best performed at a test site where data collection and installation of tools can be easily achieved (Box 7). In addition to relevant natural features, appropriate infrastructure such as electricity, wireless communication systems or a laboratory may be necessary to support the analytical steps. A test site can further be used as a learning and training centre, and for awareness-raising and public relations. It can also be used jointly by different institutions, for example park management, scientific institutions, companies and schools.

Figure 14 Ontology of the implementation phase The findings of the conceptual phase are the starting point for implementing the biodiversity monitoring programme. Next come considerations of plot and methods design, followed by acquisition of required materials. A preliminary manual detailing the methodology is developed and is provided to the workforce for training. Following training, at least one test run should be conducted, involving setting up the tools in an accessible area, acquiring, storing and analysing the data, and concluding with a brief presentation of results. The test run provides a basis for evaluating the methodology and revising the preliminary manual. Following any changes, the workforce should be retrained and a new test run should be conducted. Once the process works as described in the manual, a protocol for monitoring the selected indicators is finalised for implementation as a pre-run in the field. Source: Compiled by the report authors

Box 7

Test and experimental sites: BioDivTecs Hub Carinthia, Austria



The Lakeside Science and Technology Park (Klagenfurt, Austria) is home to a university, numerous technology companies and research institutions, and an educational laboratory for schools. The Lendspitz-Maiernigg Natura 2000 area sits directly adjacent to Lakeside Park and is used for research and education. All nature protection regulations must be observed in the protected area. An international test site for monitoring technologies is currently being developed at this location.

Increasingly, biodiversity monitoring systems are sophisticated technical systems consisting of different components. This requires testing the methods and individual devices. A test site is an outdoor research area with high demands imposed on it. First of all, the relevant species and ecosystems must be well known. Good accessibility, sufficient technical and logistical infrastructure, and comprehensive reference data are also necessary prerequisites for effective test runs for these technologies. The consent of owners or official permits may also be necessary.

Under the name BioDivTecs Hub Carinthia, various

research, educational and nature conservation organisations and companies are currently developing test and experiment sites, providing an opportunity to test and certify the technologies. There are several sites, called FieldLabs, designed for different requirements. In addition, the sites are embedded in a network of international reference programmes including GLORIA and LTERnet.

The test site is designed for use by both the developers and users of biodiversity monitoring systems, as well as educational institutions, ecologists and public bodies. The technological readiness levels of individual tools are determined through hands-on work at this easily accessible site. These tests act as pre-runs of technological workflows, breaking one of the barriers of real-world biodiversity monitoring. The results provide examples of how technologies can be applied in other protected areas.

Source: Jennifer Insupp, Ilja Svetnik, Carinthia University of Applied Sciences, Austria. Picture: View of the Lakeside Science and Technology Park. © Ulf Scherling

4.1 Deciding on sample design and methods

Sampling designs and methods differ considerably depending on the site research objectives, indicators and habitats (Elzinga, Salzer & Willoughby, 2019; Feinsinger, 2001; Magurran, 2004). Using polygons allows an area-based survey of habitats across the whole protected area or under predefined conditions (e.g. an elevational gradient) or zones (e.g. biosphere reserve core zone). Depending on the desired resolution, this approach can be very costly or not suitable for the selected indicators. Therefore, the most efficient tools to accomplish monitoring objectives should be identified at this stage, and the capacity for their proper implementation must be verified. Assessment methodology should be carefully considered. This includes the number of plots per site, spatial layout, size and shape (Figure 15). Involving a statistician during the study design process will be necessary for meaningful analysis (Adams-Huet & Ahn, 2009). A combination of different plots – a so-called nested plot design – may be implemented during test runs in order to identify the most efficient approach for plot establishment. Each plot design has its benefits and drawbacks, and selection of plot design should be carefully considered (Table 2). When different approaches yield similar results, the most cost-effective methodology should be prioritised.



Figure 15 Example plot designs for biodiversity monitoring a) Surface mapping using polygons; b) quadrat; c) circle; d) line transect; e) belt transect; f) grid. See Table 2 for more information on plot designs. *Source: Compiled by the report authors*

With the exception of the areal survey (polygon), the selection of the trial areas, transects or points, their number, and especially their location, is of crucial importance. Some common ways to select a survey area include:

- **Random:** In the case of random sampling, a very high number of samples is usually necessary to obtain a reliable result.
- **Systematic:** Sampling is carried out according to a regularly placed predetermined layout system (e.g. every third plot of a grid).
- **Stratified:** On the basis of a previous detailed survey, areas with similar characteristics are clustered and used as a basis for a representative selection of areas (e.g. 10 per cent of plots within forested areas).
- **Expert-based:** Often the area selection is made by experts who select the plots based on their experience (e.g. expected breeding sites, known intervention areas).

In practice, plot selection is often influenced by very practical criteria (accessibility, infrastructure, etc.), which can lead to a considerable distortion of the results. For example, acoustic monitoring is an excellent method to determine bird community composition, but managers must be sure that they can acquire enough devices and ensure adequate training for personnel to install them, collect and analyse the data.

Table 2 Plot designs used in field-level biodiversity monitoring programmes. Some purposes, benefits and limitations of different plot designs are described along with an example of each.

Design	Purpose	Example of use	Benefits	Limitations
Surface mapping using polygon	Complete coverage of habitat units or catchment area	Remote sensing to im- prove habitat mapping (Radoux et al., 2019)	Surveys the full area of interest; valuable and necessary background information for all types of moni- toring; analysis of areal change over time possible	Combination of remote sensing and terrestrial mapping required; not all polygons may be accessible; delimi- tation depends on data resolution; polygons change over time (pheno- logy, land uses, etc.)
Quadrat and rec- tangle	Area-based technique for abundance and distribution	GLORIA plot design (Pauli et al., 2015)	Simple inexpensive design; can be structured into subplots; redundant demarcation possible; mainly for immobile indicators (plants, vegetation); single lost corner markers can be restored	Selection and location of plots decisive for results; optimal plot size and number to be identified; strong disturbance and edge effects; individual plots may be dispersed and difficult to (re-)find; not easy to locate in dense vegetation; subset of whole area
Circle	Area-based technique for abundance, distribution with minimum edge	Forest inventory (Packalen et al., 2023)	Simple inexpensive design; can be structured into segments; mainly for immobile indicators (plants, deadwood)	Selection and location of plots decisive for results; optimal plot size and number to be identified; strong disturbance and edge-effects; centre point required for reconstruction; edges of large plots are hard to indicate in the field; subset of whole area
Line transect	Habitat gradient or species distribution, documentation of ecotone	Spatial mapping of rare trees (Bäuerle & Noth- durft, 2011)	Line-based simple design; time efficient; needs a starting and an endpoint; analysis and interpre- tation of distribution patterns or gradients quite simple	Number and location of transects decisive for results; difficult to set up in dense vegetation and over large distances; areal distribution difficult to conclude; subset of whole area
Belt transect	Habitat gradient or species distribution, documentation of ecotone	Prairie restoration (Grant et al., 2004)	Simple area-based design; time-efficient; field work uncomplicated; analysis and interpretation of distribution patterns or gradients quite simple	Number and location of transects decisive for results; allows for more sophisticated analysis than line transect; areal distribution difficult to conclude; subset of whole area

Source: Compiled by the report authors

Well-designed methodologies should permit consistent data acquisition over a long-term timescale. Methodologies may be useful to measure species population parameters including abundance, distribution or genetic diversity (Elzinga, Salzer & Willoughby, 2019). Identifying how these parameters change over time is very often the objective of the biodiversity monitoring programme and will help guide adaptive management strategies. A good biodiversity monitoring programme will allow for modular expansion of monitoring designs to adapt to changing resources or needs. Most biodiversity data fall into four categories: species surveys, community surveys, habitat types and animal–habitat relationships (Jones, 1986; Elzinga, Salzer & Willoughby, 2019). State-of-the-art approaches to implement efficient and effective biodiversity monitoring programmes are shown in Table 3.

Table 3 State-of-the-art approaches to biodiversity monitoring

Suitability for protected area management			nd sensors	p, etc.)		ensing	S			jons, plot)			etc.)	i, nests, etc.)	ght, colour, etc.)
Very well suited	1	Objects of observation	vices aı	nera tra	iera tra sing	mote s	ng tool			d, polyç	(sures,	(tracks	ones, liç
Well suited	2														
Less suitable	3			can	can	sens e re	acki	Ś		ase	ased	cts)	nclo	tion	щo
Unsuitable	4		stic	evices (nid-rang	y and tra	r device:	nethods	(area-b	(grid-ba	(transe	(nets, e	bservat	its (pher
Not relevant	-		COL												
			Passive a	Optical d	Satellite	Close / n	Telemetr	Olfactory	Genetic r	Mapping	Mapping	Mapping	Trapping	Indirect o	Attractan
		Forests and shrublands	-	3	1	1	-	4	4	1	2	3	-	-	-
Landscapes,		Glaciers, mountains, rocky habitats	-	3	1	1	-	4	4	1	2	3	-	-	-
land cover,		Wetlands, rivers, water bodies	-	3	1	1	-	4	3	1	2	3	-	-	-
vegetation		Grasslands, savannas, deserts	-	3	2	1	-	4	4	1	2	3	-	-	-
		Urban areas, artificial habitats	-	3	2	1	-	4	4	1	2	3	-	-	-
		Funghi and lichen	-	3	4	3	-	4	1	4	1	1	-	-	-
		Microbes	-	-	4	4	-	4	1	4	1	2	-	-	-
🔶 😽		Plants and algae	-	2	3	2	-	4	1	4	1	1	-	-	-
< 💥		Mammals	2	1	4	2	1	3	1	4	2	2	3	1	1
Species and		Bats	1	3	4	4	2	4	1	4	2	3	2	2	3
populations		Birds	1	2	4	3	1	4	1	4	2	2	2	2	3
		Fish	3	2	4	2	3	4	1	4	2	4	2	4	3
		Reptiles	3	2	4	4	3	4	1	4	2	3	2	3	3
		Amphibians	2	2	4	4	3	4	1	4	2	3	2	4	3
		Insects	2	3	4	4	3	3	1	4	2	2	2	4	1
		Other invertebrates	3	3	4	4	3	3	1	4	2	2	2	4	2

Source: Compiled by the report authors

4.2 Acquisition and customisation of tools and materials

Once the field design is determined, the tools and materials may be acquired. When possible, standardised protocols from similar programmes should be used. Managers should factor in possible delivery time delays due to administrative processes or supply chain problems. When selecting tools or materials, managers should verify whether adequate staff are available to use and maintain the devices. For digital tools, access to a reliable power source must be considered, for example whether the batteries will last for the duration of time between site visitation, or whether a solar array or alternative power source may be appropriate. Some digital tools require an internet connection for full functionality. Field data storage issues should be considered in advance. Data processing requirements, including volume or compatibility with cloud-based technologies, should also be considered. Data transfer from devices should be tested prior to field work. Selected tools must be resistant to climatic conditions to prevent data loss. Calibration of data collection devices should be performed once materials are in hand. Often, additional hardware or other items are necessary to physically set up the tools in the field. For equipment that will be installed in the field, extra devices should be secured in case some devices become lost or damaged. Managers and field technicians should record their experiences customising the monitoring tools in detail in a preliminary field manual.

4.3 Elaboration of a field manual

All decisions made in the implementation phase should be recorded in a preliminary field manual. The preliminary manual is a 'living document' and should include a detailed section on the analysis, archiving and presentation of data. It can be revised and adapted as needed over the course of several test run cycles. The manual should be written in a modular way, allowing specialists to focus on the relevant components of the monitoring programme. Pre-existing field manuals can serve as templates to develop key sections. Possible topics may include field logistics, data quality control, and work expectations, among others (Checklist 2). Worker safety is particularly important, as the protected area or OECM administration may be liable for negligent actions or insufficient training. Field technicians should receive adequate training for using field equipment and handling emergencies (Checklist 3). For all field work, risk analysis should be conducted, and technicians should not work alone, especially in potentially hazardous conditions. Special consideration towards possible encounters with dangerous animals or hazardous situations should be given.

Data quality is of the highest importance. Data sheets should be carefully designed. Minimum standards for metadata must be presented and explained in the preliminary manual. This includes information on who collects or revises records, the exact locations, dates and times of records, which literature and methods are used to determine species, and other site-specific or programme-specific information. These 'hidden data' are essential for the interpretation of results. All data and metadata must be relevant and precise. Data that are found to be unusable in statistical analysis do not need to be collected in future cycles.

4.4 Conducting test runs

Two types of field trials should be completed in advance of the main field work. These are test runs and pre-runs of the monitoring workflows. Test runs take place in easily accessible locations and occur after completion of each draft of the preliminary field manual. Pre-runs take place under actual field conditions and occur after finalisation of the field manual.

Test runs will verify the proposed workflows of the monitoring programme. The value of test runs cannot be overstated because they identify the most efficient methodologies, optimising investments of time and resources. Test runs may identify features of unexpected value. They will demonstrate the most effective combination of methodologies to measure the indicators. If test runs cannot be performed, simulated data should be generated and analysed to determine whether the planned sampling and analyses are sufficient to draw conclusions.

Test runs should be set up at an easily accessible site. Tools should be calibrated and set up just like they would be in an actual field survey. Data should be collected, stored, analysed and presented. It is important to perform data analysis directly after each test run to identify any problems in the statistical design or methodology. Above all, field supervisors, technicians and other participants should discuss whether the preliminary manual is applied uniformly by everyone, or whether individual steps are unclear or incompletely developed. This feedback will direct the revision of the preliminary manual. If any changes are made, additional test runs should be conducted following the revision. Every step should be repeated. Metadata from one test run to the next should be linked for analysis. Reference data are required to calibrate the methodology. This is because comparative analysis may provide the basis to reduce plot numbers or indicators to an acceptable minimum.

At least one complete assessment of collected data should result from the test run cycles. If the criteria are fulfilled in the test run, it shows that selected proxies and indicators are adequate for monitoring. Once implementation functions as recorded in the preliminary manual, the methods of the biodiversity monitoring programme are documented in a finalised field manual.

The next step is to conduct pre-runs in the field. The pre-run is the first monitoring run in the field and the last chance to revise the manual prior to starting the ongoing monitoring cycles. Different field teams should conduct the monitoring protocols on the same plot as a test of methods and workflows. Pre-run data can identify significant correlations between the measured indicators, helping to optimise the most effective methodologies. Any changes to the protocol must be noted in detail for future monitoring cycles.

4.5 Ongoing monitoring cycles

Field implementation is the roll-out of the finalised field manual as developed through test runs. In the field implementation phase, the workflow is applied in actual monitoring sites. Data collection may now begin on a regular schedule. It is important to follow the field manual as closely as possible to secure the comparability of data with findings from all monitoring cycles. High quality control should be prioritised in the data acquisition process.

Using a more frequent monitoring interval at the beginning of the programme will identify potential or unexpected trends of the indicator that may inform early revision of the field manual. Ideally, a greater number of indicators and similar methodologies could be surveyed in the first monitoring period to identify the optimal combinations of resources. Redundant information can be identified through a detailed analysis of the results and allow more efficient resource use in the long term. It should also be considered whether it is necessary to collect all indicators on all plots, or whether a subset of plots is sufficient to answer a specific question. Unforeseen events may change the required approach. Anticipating these possibilities requires enough flexibility in the monitoring cycle to permit adaptive management whilst reducing the risk of data loss to an acceptable minimum. All changes in methodology must be documented so that changes in data collection can be accounted for when analysing data over multiple monitoring cycles.

4.5.1 Field work and data acquisition

Field work is a crucial factor in any biodiversity monitoring programme. Many challenges arise in field data collection, including correctly recording information and ensuring safety of field personnel.

Taxonomy is a common challenge for the field staff. For this reason, training by an experienced taxonomist may help to maintain consistent species identification in programme activities. In many protected areas and OECMs, Indigenous and local knowledge of species is key for proper identification of valuable indicators. This emphasises the value of including Indigenous peoples and local communities in data collection planning or involving Indigenous guides in the field. Preservation of voucher specimens, for example herbarium pressings or insect collections, will help to verify the data in the case of future taxonomic disputes. If the species pool is well known and baseline data already exist, photographs may be sufficient for verification and are preferable over physical removal of plants or animals. Special permits may be required to remove specimens from the environment. Depositing genetic barcodes from environmental DNA analysis (see chapter 7.3.6) is another potential way to verify field-based findings.

Human labour is increasingly supported by advanced technologies and equipment (Dalton et al., 2022). Safe field work and efficient data acquisition must be well planned (Checklist 4). For effective field work, a systematic approach should be uniformly applied to establish for instance all field plots, such as maintaining the same directional orientation in all plots, marking the same corner, or taking plot photographs from the same angle. The use of tablets or smartphones can help standardise data collection by providing a first quality check of the data directly in the field (Joly et al., 2018). If digital tools will be used, the skill set of the user must be kept in mind. One way to ensure consistent data collection is to use digital platforms such as QField that can be loaded onto smartphones or tablets (Nowak et al., 2020). A well-prepared digital data sheet will prompt field technicians to collect all required data. It will minimise errors through offering drop-down lists, plot photography, and by using graphical interfaces rather than text (Box 8). This approach will facilitate merging datasets at a later time. Using graphical interfaces will break language or literacy barriers, allowing trained field technicians to collect data despite linguistic background. If digital options are not realistic, hard copies of the relevant data and data sheets should be provided to the field data manager as soon as possible. The data manager can identify inconsistencies while still at the field site, providing an opportunity to correct the monitoring technique or data.

The primary technical goal of data acquisition is the production of high-quality datasets for long-term time series. Each monitoring cycle is an asset whose value increases with every additional cycle of data collection. To ensure data quality, staff training should be given at the beginning of each monitoring cycle. Ranger training programmes can be provided by a conservation organisation, whilst a number of IUCN PAPACO Massive Open Online Course offerings (https://mooc-conservation.org/) may also provide training services to a protected area or OECM monitoring programme. As with the pre-run step, repeated assessment of a small set of plots on the same day and comparison of results by different staff will help clarify individual variation. Techniques ranging from simple – such as repeated site visitation adding to historical records – to complex (e.g. data analysis for extrapolation, interpolation, scenario

modelling, trend detection, etc.) should be used to maximise the data. Similar data may be available from other monitoring activities or sites. Suitable data should be integrated into the analysis and interpretation of the monitoring results. It is almost always advisable to bring along descriptive information gathered from the previous monitoring cycle, for example photographs or GPS coordinates of the site. It is very important to mark the field in a way that ensures the spatial precision of sampling points for future surveys. Providing lists of species known to the area is advisable and should be included as a resource during training of field personnel (Halloy, Ibáñez & Yager, 2011). Field trials from a long-term monitoring programme revealed that the main source of error was random rather than systematic, suggesting that adequate field training is sufficient for consistent results (Futschik et al., 2020). Therefore, providing previous findings to field technicians is not recommended and could actually bias the findings.

Box 8

Working with SMART and CyberTracker: An example at Lake Tana, Ethiopia



The Spatial Monitoring and Reporting Tool (SMART) is an open-source software programme to help protected area and OECM managers handle and interpret conservation data. SMART is a desktop app that can be used in conjunction with mobile apps such as CyberTracker or SMART Mobile to enable remote data collection on a tablet or other mobile device. CyberTracker utilises an icon-based user interface to simplify data collection. Field technicians receive project-specific training to ensure that data are collected in a reliable way.

At Lake Tana Biosphere Reserve, Ethiopia, SMART is being implemented with CyberTracker in an effort to track the control of the water hyacinth (*Eichhornia crassipes*), an invasive aquatic plant species that out-competes all other species growing in the vicinity and poses a threat to aquatic biodiversity. Several control techniques are available, including chemical control, physical removal, biological control and multiple techniques in combination. To date, physical control techniques have been implemented at Lake Tana. Local farmers play an important role in removing the water hyacinth from the water. The quality of the work is observed by development agents who are required to create awareness about the adverse effects of the weed and report the daily volume of plants removed from the lake. The development agents are a natural choice to collect data using the CyberTracker app because their job is to oversee the weed control work. Collecting infestation data with CyberTracker takes no extra time in the field.

At Lake Tana, the first step of the SMART / CyberTracker workflow was to develop a desktop computer model for data collection using the SMART app. The model was then exported to CyberTracker as a checklist of possible conditions at a survey point. For water hyacinth infestation, different categories are allowed, such as presence / absence, levels of plant density, or per cent cover over a standardised area. Scientific researchers provide periodic training to the development agents on how to use CyberTracker. Following training, test runs are conducted to correct potential problems in the workflow. All steps are recorded in a field manual for future reference. After collection by trained local participants, data are transferred from the mobile device to the computer, and georeferenced data are exported in the office to SMART. Depending on the configuration of the mobile device, upload can occur via email, Bluetooth or directly to the cloud. Data from the Lake Tana SMART / CyberTracker workflow can be used to generate maps showing changes in the state of the water hyacinth infestation over time. This information can be evaluated to recommend possible changes to the adaptive management programme.

Source: Daniel Mengistu and Tirusew Ayisheshim Ebistu, Bahir Dar University, Ethiopia, and Melanie Erlacher, Carinthia University of Applied Sciences, Austria. Picture: Detection of the spread of water hyacinth (Eichhornia crassipes) through the use of remote sensing technologies. © Geospatial Data and Technology Center, Bahir Dar University

4.5.2 Storage, backup and archiving

Today's digital biodiversity monitoring devices are producing data at an unprecedented rate. These so-called 'big data' are now a component of modern biodiversity monitoring systems, allowing advanced statistical analysis that can transform evidenced-based decision-making (Strohbach et al., 2016). There are many benefits of remote data collection, but managers of protected areas and OECMs must decide how to address the many challenges of storing and using big data. It is therefore crucial for managers to implement a holistic data model for each biodiversity monitoring system. Individual programmes should develop datasets that conform to the data model. Big data management is divided into five steps: data collection, uploading, processing, analysing, and visualising the findings (Mazumdar et al., 2019).

For both traditional data and big data, the first step is collection. Once collected, data should undergo a quick, immediate analysis to detect gaps or inconsistencies that cannot be remedied at a later stage. The easiest way to do this, if resources allow, is by using smartphones or tablets to collect field data. Data can be synchronised onto different devices to avoid data loss. In the next step, managers need to identify how the data will be entered into the database. Data should be delivered in their simplest raw form into a permanent data repository so that they can be read years into the future. Some example formats include entering data in columns, as a document or spreadsheet, or as a time series. Database management systems are available for common formats (Mazumdar et al., 2019). Metadata that are attached to the raw data will enable the server to process the information and store it across cloud networks. Early consideration of which metadata will be included will improve data management.

Storage of data is usually independent of data analysis. To enable future analysis, it is important that raw data and metadata are stored not only in readable formats with unique identifiers of each dataset, but also that backup archiving mechanisms are in place. Metadata allow data from different programmes to be compared to one another, increasing their long-term value (Huettmann, 2009) (Box 9). Using protocols for data archiving and backup is very important. In some cases, automatic cloud-based data transfer will enable instant, transparent and open data access worldwide. Approaches using artificial intelligence are assisting the archiving process through automatically categorising new data (Colavizza et al., 2022). If resources allow, a physical printed copy of the data could be stored as an additional defence against loss. Considering long-term data accessibility, data rights and data security is important and should be built into the project from the early design stages. Technical concepts and data storage strategies are provided (Checklist 5).

Box 9

Unified approach to scientific monitoring in the mountains: Example of the GLORIA network



Biodiversity monitoring networks help managers understand how biodiversity behaves in certain biomes. The Global Observation Research Initiative in Alpine Environments, GLORIA, is an exemplary long-term scientific network. It is designed to record changes in plant species distributions due to climate changes in high mountain environments. The GLORIA network provides standardised data on species richness and abundance through long-term surveillance, describing the abiotic environment, useful biodiversity indicators, large-scale risks and potential conservation strategies. Mountain biomes are ideal for these objectives because they occur in all ecozones from the tropics to the polar regions, and plants in mountain zones are restricted by cold temperatures (Pauli et al., 2015).

Many GLORIA monitoring sites are located in protected areas. Parameters affecting biodiversity are investigated systematically at each site according to three activity levels as defined in the freely available GLORIA field manual (Pauli et al., 2015). Protected area managers can implement the standardised GLORIA methodologies and upload monitoring data to the central database whilst retaining data property rights. This allows local site data to be compared globally with data from the GLORIA network, increasing the power of the dataset to make inferences on drivers of local biodiversity change or threats facing the region. GLORIA is an example of a monitoring programme that is successful due to its uniform approach to the monitoring design at all sites. Specific criteria must be met in order to establish new sites. Field methodologies and survey intervals are established in a clear manual of protocols. Also, datasets must follow a certain structure and fulfil a given quality standard.

Project metadata are customisable. Before data collection begins, it is important to define the required metadata, as is the case within the GLORIA network. Generally certain elements should always be included: title, dataset reference date, topic of the dataset, language and an abstract, contact information in case of questions, and a time stamp (Huettmann, 2009). Metadata management tools (e.g. Metadata++, ExifTool) allow users to record relevant metadata, verify that files contain all necessary metadata, and decrease the risk of omitting important elements. The generous release of these invaluable datasets will allow a variety of ground-breaking, largescale analyses to be conducted, further leveraging the immense potential of locally collected biodiversity data.

Source: Klaus Steinbauer, Carinthia University of Applied Sciences, Austria. Picture: Monitoring plot according to the standardised GLORIA methodology. © Klaus Steinbauer

4.5.3 Data policy

Having a culture of open data access across protected area networks is an ideal situation. But, if not required by national or institutional policies, biodiversity data might not be shared. Reasons can be a sense of data ownership, national security concerns, poor coordination across institutions, and perceived disadvantages from international obligations (Liang & Gamarra, 2020). Paid technicians and researchers may face poor job security, a dangerous and demanding work environment, and may be expected to dedicate their own resources for data collection (de Lima et al., 2022). For citizen scientist volunteers, motivation to participate may be for social reasons, for learning about and connecting with the natural world, and to gain a sense of personal achievement (Ganzevoort et al., 2017). Clear communication with data collectors about the intended use of their data is key to maintaining motivation for current and future projects.

Biodiversity data produce improved knowledge on the species examined in monitoring programmes and scientific research. The data policy of the protected area or OECM should be clearly stated in the overall management plan. Generally, information and raw data are considered facts and may not have legal protection under copyright law (Egloff et al., 2016). If collected through public funding, biodiversity data may even belong to the public. Free universal access to data gives a sense of transparency to the data collection process, leading to wider acceptance of findings. For these reasons, biodiversity data and research results should be published in open access sources. Within this context, well-justified restriction of certain data should be accepted. Special circumstances include masking the location of endangered species, maintaining the privacy of a co-operator, or protecting Indigenous communities. To safeguard the rights and interests of Indigenous peoples and local communities, and to guarantee transparency in the data collection process, a number of data sharing principles must be followed (Box 10).

Data policies should outline who owns the data, whether it will be freely available in an online data repository, and how the data will be used. One established framework is to designate data using Creative Commons licenses (Culina et al., 2018). Different Creative Commons licenses provide a range of privileges and permissions of data access. Publishers may have programmes offering waivers to pay reduced publication fees for open access content. Managers should communicate directly with publishers to learn whether they qualify for such programmes.

Box 10

CARE and FAIR data sharing principles: Protecting the data sources

To explicitly consider Indigenous peoples' rights and interests, CARE Principles for Indigenous Data Governance focus on the people and purpose underlying the data (Carroll et al., 2020). CARE principles include **collective benefit** allowing Indigenous peoples to benefit from the data, **authority** of Indigenous peoples to control how they are represented in the data, **responsibility** of data managers to show how data will support Indigenous peoples, and **ethics** emphasising Indigenous peoples' rights and well-being.

Online deposition of metadata allows discoverability of project data, helping to compare results across programmes. Metadata describe the permanent record of observation. They are a requirement for modern standard data management processes meeting the FAIR data sharing principles (Wilkinson et al., 2016). When data are submitted to digital repositories, they should be assigned a **findable** persistent identifier and described with high-quality metadata. They should be open and freely **accessible**. To be **interoperable**, metadata should be described in a common and accepted language, with appropriate reference to additional metadata. To be **reusable**, metadata should meet discipline-related standards and have a clear data usage license describing any limits to reusing the data, such as defined in Creative Commons licenses (Kissling et al., 2018).

Source: Compiled by the report authors

4.5.4 Data analysis and evaluation

Data analysis techniques should be suitable for the hypotheses and research goals. Analysis of monitoring data should be scheduled with a statistician (Adams-Huet & Ahn, 2009). The procedures should be standardised using well-documented or scripted semi-automatic processes. Interpretation of findings can be improved with the help of a specialist for the target object, if available. Interpretation is the process of understanding data, trends and statistical tests (Cooperrider, 1986). It is the critical link between data collection and adaptive management. Interpretation should address the following questions: What do the data mean?

What is the cause of the trend? Are the data of good quality? Are the data sufficient to make decisions? What are the expected impacts of the proposed management? Answers to these questions will ideally be based on the scientific results but might require a 'best guess' opinion in order to continue with the decision-making process, particularly if the manager is faced with knowledge gaps. Notably, conclusions from data analysis might make sense only after multiple monitoring cycles have been completed.

4.5.5 Aggregation and integration into the management cycle

The conclusions of the biodiversity monitoring programme should support management decisions and reporting obligations. The 'classic' means of communication is through programme reports and scientific publications (Checklist 6). These formats of communication are typically very technical. However, management decisions rely on both technical and managerial experiences. As digital technologies improve, visualisation of trends is facilitated through online dashboards (Box 11). A site dashboard is designed to allow easy and regular observation of information by decision-makers. Visualisation is the next step to give value to the biodiversity data. Using dashboards is a useful form of data visualisation, giving particular value to the biodiversity data and making the complex processes and trends accessible in a simple form. Data can be visualised graphically to improve analysis. Appropriate visualisation of the data further contributes to the presentation of findings to various stakeholders. Presentations are a format that may generate stakeholder engagement. Presentations should be given early and often to stakeholders. Data should be presented in an open and transparent context so that the audience can freely reflect upon the findings.

Box 11

Dashboard view: The example of EarthRanger



EarthRanger is a data visualisation and analysis platform that gives conservationists the real-time information they need to keep wildlife, habitats and communities safe. The platform collects, integrates and displays all historical and available data. These data are combined with reports from the field to provide one unified view of collared wildlife, rangers and any other assets whether on land or sea. The applications of EarthRanger are diverse and are customisable to the primary needs of the area or organisation. Among them, the platform is used to track and study wildlife, coordinate ranger units and proactively mitigate human-wildlife conflict. Through active partnerships, features like instantaneous alerts and patrol management, and integrations with cutting-edge satellite services, EarthRanger enhances how organisations monitor vast areas in real-time.

The system is user-friendly, easily accessible via computer, tablet and smartphone, and ideal for use in the field without radios or satellite trackers. The platform is compatible with greater than 100 leading hardware devices, data services and platforms like SMART and Skylight, saving time and resources, and increasing impact.

EarthRanger is part of the Allen Institute for Artificial Intelligence (Al2), a non-profit institute created by the late Microsoft co-founder Paul G. Allen, with the mission of conducting high-impact artificial intelligence research and engineering in service of the common good.

Source: Jordan Steward, EarthRanger. Picture: A wildlife biologist uses EarthRanger to visualise the real-time movements of wildlife across a protected area. © Gorongosa National Park

Chaper 4 Implementation phase

Free and easy access to monitoring data ensures the legitimacy of the monitoring programme. A reasonable exception would be to exclude sensitive data from public access (Lunghi et al., 2019). The importance of open communication is critical, regardless of whether observations and trends are positive or negative. Standard algorithms should be used to avoid errors, to ensure comparable results, to help visualise the results, and to reach appropriate stakeholders. In this way, trends of the selected indicators can be visualised for scientists, administrators and the public. Sharing the findings within the monitoring network or with research institutions allows the possibility for wider comparisons across ecosystems and species. This is a crucial step to distinguish anomalies from actual trends. It may also justify the possible need for increased resources to achieve management objectives.



Monitoring vegetation using GLORIA methodology, Annapurna, Nepal. © Robbie Hart

CHAPTER 5.

Re-evaluation phase: Decision on adapting or terminating

5 Re-evaluation phase

The biodiversity monitoring system should be designed in a way to let managers and administrators reflect on the successes and shortcomings of individual programmes. It is recommended to review a biodiversity monitoring programme at a predetermined interval. Individual components should be reviewed at more frequent intervals in order to determine the best use of available resources. Completion of a monitoring cycle may lead to additional questions, and experiences should be shared across borders or within a protected area network. Technological advances, new funding schemes or other changes may allow a more efficient approach to conduct future monitoring, as identified through reflection.

Even simple changes in data collection methodology can lead to incompatibility between datasets and loss of statistical power. During re-evaluation, if the managers consider changing the field manual, possible effects on data analysis and presentation should be thoroughly assessed. If it is necessary to change the approach, a dataset comparing the old and new methods should be collected to assess the impact of the changes on data analysis and presentation.

Outcome: Monitoring approaches and goals should be re-evaluated after several monitoring cycles or following major changes to reporting obligations. Successes or shortcomings of the biodiversity monitoring system and its components can be communicated to administrative agencies and other stakeholders. A decision on continuing, modifying or terminating the biodiversity monitoring programme should be made. In the case of major revision, managers and stakeholders may need to go back to the conceptual phase to determine the scope of the monitoring programme based on available resources and site factors.



Ephemeral pond, William L. Finley National Wildlife Refuge, Oregon, USA. © Daniel Dalton

CHAPTER 6.

General considerations for effective monitoring

6 General considerations

With the framework for establishing effective biodiversity monitoring programmes in mind, this chapter describes an overview of basic considerations that will promote effective biodiversity monitoring. Knowledge of site-specific conditions and management objectives are required for effective planning of monitoring procedures.

6.1 Obligations: International conventions and policies

The main purpose of protected areas and OECMs is to conserve the unique biodiversity that is contained on site. Many international policies require or imply biodiversity monitoring and reporting efforts (Mitchell, 2003) (Box 12). Managers should have a clear understanding of which institutions, treaties, instruments and the corresponding mechanisms apply to their site reporting requirements. These may include the CBD, various reports for UNESCO, World Conservation Monitoring Centre, linkages to the IUCN Red List and IUCN Species Survival Commission, Living Planet Index, Natura 2000 network, and more (Table 4).

Box 12

Monitoring in Natural World Heritage sites: Japanese island ecosystems in focus



Natural World Heritage Sites are areas of high biodiversity that are listed by the World Heritage Convention, an international treaty that aims to permanently protect the most unique ecosystems on the planet. Sites are nominated based on their Outstanding Universal Value. This is defined in a way that emphasises a site's importance for the international community because these sites are globally unique and cannot be replaced (Bertzky et al., 2013). Natural World Heritage Sites are expected to stimulate local economies through international valorisation and branding. Through ecotourism, visitors are encouraged to take an interest in natural ecosystems, promoting conservation action. Once an area is registered as a Natural World Heritage Site, site-based conservation and sustainable use of biodiversity must be well planned. Real and potential threats to Natural World Heritage Sites include over-tourism, invasive alien species and the impact of climate change (Osipova et al., 2014; Osipova et al., 2017; Osipova et al., 2020). The intensification and increased risk of natural disasters due to climate change threaten not only the ecological attributes that support heritage values, but also the livelihoods of visitors and residents. To mitigate these threats, implementing monitoring systems and adaptive management activities are essential tools (Osipova et al., 2020). Local residents play a major role. Japan's Natural World Heritage Sites include a series of islands containing rare and endemic species of Outstanding Universal Value (Toyama et al., 2023). These island ecosystems are in danger of losing endemic species due to climate change, invasive species introductions and other human impacts (Working Group for Comprehensive Assessment of Biodiversity and Ecosystem Services, 2021). Discovering changes in biodiversity and endemism can be achieved through monitoring, helping managers understand how pressures contribute to ecological changes. This is essential knowledge for establishing a management system to mitigate threats. A scientific committee should be established in each Natural World Heritage Site to allow monitoring to support the corresponding National Biodiversity Strategy and Action Plan. In Japan, a policy is being considered that will improve sustainable development through the promotion of ecotourism. Using Nature-based Solutions, it is expected that the cobenefits of ecosystems will extend to local communities, balancing economic sustainability with conservation and utilisation (Study Group on the Next National Biodiversity Strategies and Action Plans, 2021).

Source: Yayoi Takeuchi, Senior Researcher, National Institute for Environmental Studies, Japan. Picture: Asarum fudsinoi. © Yayoi Takeuchi

Effective date	Name of convention / programme	Type and interval of reporting / monitoring
1971	Man and the Biosphere Programme (MAB)	Every 10 years (5-year interim reports): site-specific evaluation report to MAB Programme
1975	Convention on International Trade in Endangered Species of Wild Fauna and Flora (CITES)	Annually: national reports to the Secretariat on imports, exports, re-exports and introductions of specimens
1975	Ramsar Convention on Wetlands	Every 3 years: national reports to Conference of the Parties
1975	World Heritage Convention (WHC)	Every 6 years: report on site integrity to World Heritage Centre
1979	EU Birds Directive	Every 6 years: report on population size and trends of bird species
1983	Convention on Conservation of Migratory Species (CMS)	Every 3 years: national progress reports on implementation
1992	EU Habitats Directive	Every 6 years: conservation status and trends of species and habitats
1993	Convention on Biological Diversity (CBD)	Every 4 years: national reports to CBD
1996	Convention on the Protection, Management and Development of Marine and Coastal Environment of the Western Indian Ocean (Nairobi Convention)	Every 2 years to the Conference of the Parties for review
1998	Convention on the Protection of the Marine Environment of the North-East Atlantic (OSPAR Convention)	Every 2 years: assessments of the status of the OSPAR network of Marine Protected Areas
2016	UNESCO Global Geoparks	Every 4 years: site-specific revalidation report to UNESCO Global Geoparks

Table 4 List of main international biodiversity treaties and conventions

Source: Compiled by the report authors

A global standard on implementation of Nature-based Solutions is available (IUCN, 2020). Implementing Nature-based Solutions should align biodiversity conservation with sustainable development. Effective solutions will help realise ecosystem-based management decisions that address societal challenges whilst improving the state of biodiversity and human well-being. The global standard links eight criteria guiding the background principles of Nature-based Solutions with the assumptions required for effective management interventions (Table 5).
 Table 5 Criteria and assumptions for integrating Nature-based Solutions into management activities

Criteria for implementing effective Nature-based Solutions	Assumptions				
	The most urgent challenges are prioritised;				
1. Solutions target societal challenges.	Managers adequately understand the targeted challenges;				
	Outcomes of human well-being are benchmarked and periodically assessed.				
2. Spatial scale helps design implementation measures.	Economic, societal and ecosystem factors interact with each other;				
	Activities should be synergetic across sectors;				
	Risk assessment extends beyond the intervention site.				
	Evidence-based response to the state of the ecosystem and its main drivers;				
3. Solutions improve biodiversity and ecosystem integrity.	Minimum benchmarks established to help determine the effect of the measure;				
	Monitoring and evaluation plan examines unintended effects of the measure;				
	Strategies identified to improve ecosystem integrity and connectivity.				
	Benefits and costs of a measure are documented, including who pays and who benefits;				
4. Solutions are economically	Cost-effectiveness study determines level of support for implementation;				
viable.	Value of implementation is greater than possible alternatives;				
	Material, financial and human resources available for proper implementation.				
	All stakeholders can use a readily available feedback and resolution mechanism;				
5. Solutions assume inclusive, transparent and empowering	Participation founded on gender equality, respects the rights of Indigenous peoples, and does not depend on age or social status of participants;				
governance.	All affected stakeholders should be involved in the processes of the measure;				
	Joint decision-making should be possible across jurisdictional boundaries.				
 Solutions balance achieving their primary goal(s) and providing other benefits. 	Costs and benefits of different intervention scenarios are considered;				
	Rights, usage and access to land or resources are recognised and valued;				
	Safeguard systems are in place to avoid negative consequences of a measure.				
	Monitoring and evaluation of the measure against the baseline condition is regularly performed;				
7. Solutions are managed adaptively.	Monitoring and evaluation plan is used through the full course of the intervention measure;				
uduptivoly.	Learning opportunities from the intervention measure will contribute to adaptive management.				
	Design, implementation and findings from the measure should be shared for wider transformative changes;				
8. Solutions are sustainable and can be mainstreamed.	Measures must be aligned with other current regulations and policies, and incompatibility across sectors is documented to provide learning opportunities;				
	Measures contribute to targets from national and sub-national policies.				

Source: IUCN, 2020, adapted by the report authors

The CBD is a landmark agreement intended to promote the conservation of biodiversity, the sustainable use of biodiversity, and the fair and equitable sharing of the benefits derived from genetic resources (CBD, 1992). Since ratification in 1993, multiple annexes to the CBD have been implemented, including the Cartagena Protocol on Biosafety in 2003 and the Nagoya Protocol Access to Genetic Resources in 2014. Currently, in the UN Decade on Ecosystem Restoration 2021–2030, conservationists and policymakers are addressing the biodiversity crisis through the Kunming-Montreal Global Biodiversity Framework (GBF) (Nicholson et al., 2021). Target 2 of the GBF calls for effective restoration of 30 per cent of the area of degraded terrestrial and aquatic ecosystems.

Target 3 is a headline indicator of the GBF and contains the ' 30×30 ' goal. To meet this goal, 30 per cent of the Earth's surface should be placed under effective conservation by the year 2030. Integration of conserved sites in the larger landscape should occur with full recognition and respect for the territorial rights of Indigenous peoples and local communities. Coverage calculations and importance of sites involves calculations and mapping based on submitted reports by national governments (UNEP-WCMC, 2023b). The effectiveness component of Target 3 is still being defined but is focused primarily on creating positive outcomes for the conservation of biodiversity (CBD, 2023). We define a consistent monitoring strategy in this guideline that is intended to be a key tool to satisfy effectiveness requirements.

6.2 Art of omission: Daring to simplify

Monitoring programmes can suffer from collecting data on too many variables. Each dataset increases the complexity of data analysis. Because data collection and analysis require human effort, uninformative or poor-quality data represent a waste of time and resources. In a worst-case scenario, poor data may increase the risk of errors by introducing inconsistency in the dataset (Box 13). Mixing a variety of data formats further challenges comparisons across sites or over time. The narrowest set of indicators representing the status of the site should be selected (Butler et al., 2012). Indicators can include key habitats, species or proxy parameters that are tightly linked to the component of interest (Lindenmayer et al., 2015). Abiotic factors such as air quality, water quality or temperature may be monitored as proxies of biological elements in cases where the factor is well correlated with the biological feature of interest. Simplification of a monitoring programme may be informed through test analyses of simulated data to help identify the most informative indicators (Figure 16).



Figure 16 The effort required for monitoring system implementation and data management increases with large numbers of selected parameters. For efficiency and the greatest likelihood of obtaining time-series data, indicators and data collection techniques should be kept to the minimum required level. Source: Compiled by the report authors

Box 13

'Best of' mistakes in monitoring: Succession in a fallow field in Austria



The change in the plant community on abandoned agricultural fallow land was analysed within the framework of a 30-year monitoring programme in a conservation area near Liebenfels, Austria. Surveys were conducted using a very simple study design. Plant species and cover were recorded on 29 study plots (5×5 metres). Through data analysis, the scientific team found numerous errors.

Here is an excerpt of the monitoring mistakes. Some plots and transects could not be found from one survey to the next despite their locations being redundantly marked. Individual plot records, photos, files and documents could not be found. Available photos and documents were not labelled or could not be assigned in time and space and therefore could not be utilised. Individual data files could no longer be read because file formats became outdated. Species composition changed 'suddenly' with a change of scientific staff or new taxonomic designations. Phenological recordings at slightly different times of the year changed the result. Lack of metadata complicated the interpretation of the findings.

These examples lead to the following conclusions: 1. Each individual set of data is vulnerable; 2. Each indicator multiplies complexity and costs disproportionately; 3. Each irregularity reduces the quality of results disproportionately.

When the monitoring plan is more complex than necessary, the chances increase that some important data will be lost. Considering that mistakes will always be made in long-term programmes, it is better to design simply and realistically at the beginning. Monitoring is the art of omission.

Source: Michael Jungmeier & Melanie Erlacher, Carinthia University of Applied Sciences, Austria. Picture: View of the monitoring area. © Elisabeth Wiegele

6.3 Biodiversity monitoring systems: Designing modular, multi-scale and multi-purpose monitoring systems

Integrating different regional, national or transboundary management programmes is essential to create synergies and promote management effectiveness beyond the boundaries of protected areas and OECMs (Karadeniz & Yenilmez Arpa, 2022a). When designing a new biodiversity monitoring programme, data should be acquired from various sources to improve the concept and design, as well as to identify the appropriate required materials. In some cases, data will come from external partners. National programmes may already have valuable information available, such as assessments of key species that are relevant for protected area management. Effective modular components of a monitoring programme may be adapted for different regions or transferred to similar ecosystems (Dallas, 2021) (Box 14).

Box 14

From individual sites to a protected area system: The Brazilian Programa Monitora

Since the beginning of the 21st century, the number of protected areas and their coverage have more than doubled in Brazil, today numbering about 2,000 units covering more than 1.5 million km² (OECD, 2015). Programa Monitora was formalised in 2014 after several years of conceptualisation. The main goal is the evaluation of conservation effectiveness of the 334 protected areas managed by Instituto Chico Mendes de Conservação da Biodiversidade, Brazil. The programme also produces qualified information for management decisions at local and regional scales. Today, Programa Monitora contains three subprogrammes focusing on 10 types of ecosystems. Currently, 23 targets are being monitored, some of them corresponding to previous long-term monitoring initiatives (e.g. marine turtles), and others representing novel indicators (e.g. dragonflies in streams). Some targets are related to resource use, like

artisanal fisheries and Brazil nut harvesting, and all protocols deliver information about ecosystem health.

The programme was designed after considering successes and drawbacks of previous worldwide examples, thus adopting a set of integrated principles. Common targets and protocols allow comparison across regions. Targets are selected based on local environmental and social factors. The scale of sampling is optimised according to the national, regional or local scope of the subprogramme.

Protocols are simple enough to allow data collection by people without formal biological training, but effective because they are based on sound science. Protocols are modular to allow specific combinations for each site.



A participatory process utilising community-based volunteers is used in order to generate programme acceptance. All participants receive appropriate training. The national protected area network utilises a unified data management strategy, with a policy to provide open and immediate data access, showing transparency. Finally, qualified scientific debate around the findings allows widespread development of learning and knowledge.

The resulting programme is widely recognised by protected area managers, local communities, funding agencies and the scientific community. Programa Monitora strengthens links between institutions and people of different backgrounds, generating local results, national reports and scientific papers.

Programa Monitora - Português (Brasil)

(https://rb.gy/0naso8)

Source: Katia Torres Ribeiro, Instituto Chico Mendes de Conservação da Biodiversidade, Brazil. Picture: Butterflies are part of the Terrestrial Component of Programa Monitora. It is a group that arouses interest for many people, and through simple protocols it is possible to collect information about the health of forests. © Instituto Chico Mendes de Conservação da Biodiversidade

An indicator can be used to evaluate multiple monitoring obligations, or on the other hand multiple indicators can be used together for detailed area-based reporting. Ideally, components can be used in a modular way, meaning that certain indicators of past and existing programmes can be monitored in future programmes, or across multiple scales. A biodiversity monitoring system should include a monitoring programme on habitats (e.g. ecosystems and land cover), in addition to monitoring key species. Findings should be disseminated in various formats, from scientific reports to stakeholder outreach (Figure 17). This background allows integration of further monitoring programmes and interpretation of results based on actual management activities.



Figure 17 Data are brought together to determine a modular framework of new biodiversity monitoring programmes. Data allow identification of improved concepts, realistic capacities, and required equipment for monitoring, leading to informed decision-making, project or programme evaluation, and outreach to stakeholders. Source: Compiled by the report authors

6.4 Combining forms of knowledge

A high amount of the world's biodiversity is conserved in areas that were managed sustainably for generations by Indigenous peoples and local communities using traditional ecological knowledge (IUCN, 2022a). Protected areas and OECMs lie at the intersection of scientific and traditional ecological knowledge because sites are often managed by organisations that were not historically part of the area (Danielsen et al., 2009). Co-creating knowledge through linking traditional ecological knowledge and scientific approaches demonstrates how partnerships can improve conservation outcomes (Box 15). To ensure that a protected area or OECM management strategy will produce the best outcomes for biodiversity and traditional land uses, traditional ecological knowledge and scientific knowledge need to be viewed as equal contributors towards understanding nature (Thompson, Lantz & Ban, 2020). For the highest chance of success, managers must work towards developing strong relationships and trust with Indigenous community members. Utilising the input of Indigenous communities in all phases of the biodiversity monitoring programme is essential for equal power-sharing and developing trust. Engaging Indigenous peoples and local communities in genuine participatory processes will eliminate one-way flow of information (Borrini, Kothari & Oviedo, 2004). Assigning leadership roles to Indigenous group members will help to equalise the influence of scientific and Indigenous forms of knowledge (Thompson, Lantz & Ban, 2020). Empowering Indigenous or community members in decision-making processes will break down biases and allow community members to be confident that they control how their knowledge will be used and represented. Developing a mutual understanding of the cultural backgrounds of involved stakeholders will further contribute to increasing different forms of knowledge. In some cases, working with a cultural liaison advisor can help with this process. Potential technological barriers can be broken down through the use of digital tools such as the Spatial Monitoring and Reporting Tool (SMART) allowing a user to collect raw data based on icons.

Box 15

Forest restoration using traditional ecological knowledge in a transboundary biosphere reserve, Peru and Ecuador



The iconic tropical dry forests of the Bosques de Paz Transboundary Biosphere Reserve of north-western Peru and southern Ecuador are recognised as an important world biodiversity hotspot (Olson & Dinerstein, 1998). The biosphere reserve covers more than 1.5 million ha and contains multiple protected area designations within its boundaries, including Cerros de Amotape National Park of Peru that comprises much of the core zone of the biosphere reserve (UNESCO, 2021). The area has been greatly affected by human pressures in recent decades, such as livestock husbandry and extraction of natural resources. Today, conservation of the wilderness and restricted use areas of the dry forest Bosque Seco Denso de Montaña is a primary objective of the national park.

Restoration of special use and buffer zones of Cerros de Amotape National Park provides an opportunity for local contribution to management objectives. The park management plan calls for restoration of more than 60,000 ha of national park land to recover its environmental quality to a historically natural condition, or about 38 per cent of the national park (SERNANP, 2015). This goal meets the main objective of Target 2 of the Kunming-Montreal Global Biodiversity Framework, where: "by 2030 at least 30 per cent of areas of degraded terrestrial, inland water, and marine and coastal ecosystems are under effective restoration, in order to enhance biodiversity and ecosystem functions and services, ecological integrity and connectivity" (CBD, 2022).

Researchers interviewed stakeholders from communities located in and around Bosques de Paz Transboundary Biosphere Reserve on their perspectives for using native plant species for provisioning of ecosystem services (Fremout et al., 2021). They compiled more than 4,000 responses of local expert and non-expert community members on traits including the potential use, stress tolerance and perceived threat status of nearly 150 species. Available scientific knowledge was then compared to the traditional ecological knowledge of the stakeholders. Favourable comparisons were realised between traditional ecological knowledge and scientific knowledge for 97 per cent of the species for which scientific knowledge was available. About half of the species lacked scientific knowledge altogether. The findings showed that the close relationship of local communities to native plant species had allowed development of traditional ecological knowledge that could not be replaced by scientific knowledge (Fremout et al., 2021).

Traditional ecological knowledge provides a strong basis upon which the most useful species can be prioritised for restoration purposes. Utilising local knowledge resources further invites communities to contribute meaningfully to restoration projects, representing community members and leading to greater support and improved chances for successful long-term management.

Source: Compiled by the report authors. Picture: Local community members explaining the uses of the choloque tree, Department of Piura, Peru. © Tobias Fremout

6.5 Continuity risks: Avoiding disruptions and gaps in data

Robust statistical design must be considered in advance of data collection (Zuur et al., 2009). The value of a long-term monitoring programme is diminished if the data are collected at the wrong time or season or at insufficient intervals. If data are collected at an inadequate frequency, the interpretation of the time series may be fundamentally flawed (Figure 18). Contingency plans accounting for potential loss of resources or financial support should be developed in advance of any disruption. After collection, data should undergo a rapid preliminary analysis to identify any problems with the conceptual approach of the monitoring strategy or with its implementation. Consultation with a statistician can identify statistical tests that may be used with irregular datasets. Finally, a timeframe for data analysis should be designated.



Figure 18 Monitoring at an improper frequency may display faulty trends. Source: Compiled by the report authors

It is essential to keep in mind the long-term nature of biodiversity monitoring and to ensure continuity across institutional, technical or organisational changes. This is a significant challenge given the relatively long course of biological processes compared to rapid changes in monitoring resources and technologies. For instance, major changes in scientific theory, technology and human societies have occurred within the lifespan of a tree (Figure 19). Building on the experiences gained from previous programmes will improve the outcome of following efforts.



Figure 19 Timeline of key achievements advancing monitoring capacities Scientific discovery occurs slowly from the perspective of a human lifespan. The tree rings represent the idea that important discoveries occur rapidly on the scale of ecosystems. Over the past three centuries, our understanding of life has fundamentally changed. The way we monitor the life around us is undergoing a similar revolution today. *Source: Compiled by the report authors*

6.6 Detecting trends and correlations: The value of time series

Monitoring involves a regular and long-term process of data collection. This information provides a measure of whether management goals are being met. Biodiversity monitoring should provide objective measurements of the state of key ecological features. The value of monitoring increases the longer the protocols are in place because precise estimation of indicator status – and in some cases even the existence and direction of change – can be determined only with sufficient observations over time (Figure 20). Accurate and timely observations provide an important basis for decision-making, whilst faulty conclusions based on too few data points can harm the integrity of the management programme. For conservation or restoration purposes, objectives or impacts may be evaluated over time through monitoring (Box 16). Multiple sites implementing the same approaches will help to identify and compare long-term trends. Threats can be identified in long-term datasets through correlating changes in the state of a sensitive indicator. In this way, the state of the indicator can be monitored as a basis for adaptive management decisions (Westgate, Likens & Lindenmayer, 2013).



Figure 20 Long-term

monitoring provides increasingly accurate knowledge. In this example, the longterm cycles of the indicator become clear over time, leading to less uncertainty in future predictions. *Source: Compiled by the report authors*

Box 16

Monitoring and evaluation of environmental measures: Water usage in the south-east of the United States

The concept illustrated in Figure 18 and Figure 20 shows that regularly collected data will show trends on the subject. In the context of protected area management, available data from multiple monitoring cycles should therefore allow managers to predict future observations given no changes to an intervention. Conservation and management decisions are often limited due to one-time observations or at best limited pre- and post-intervention measurements (Adams, Barnes & Pressey, 2019).

To show the value of time series in an environmental context, we draw upon studies that used data in a randomised control trial to decrease household water use (Ferraro & Price, 2013). To test how much an experimental design with paired controls and limited pre- and post-intervention measurements can match the intervention's 'true' impact, the authors of the water use study ran a number of experiments. They found that where only one pre-intervention period was considered, serious challenges occurred in matching real-world findings to experimental estimates (Ferraro & Miranda, 2014). A follow-up study with multiple pre-treatment observations resulted in far better predictions (Ferraro & Miranda, 2017).

These studies show that when pre-intervention observations are unavailable, evaluating the intervention is challenging even with the best statistical approaches. Surveillance monitoring to detect trends over time will still be possible post-intervention but will be poorly correlated with the intervention itself. When pre-intervention observations are available, multiple post-intervention measurements will improve interpretation.

The take-away message is that investing in multiple pre- and post-treatment observations will improve the understanding of the changes in environmental outcomes and attributing those changes to the management interventions.

Source: Vanessa Adams, University of Tasmania

6.7 Maintaining ecological balance: Establishing baselines and thresholds

The ability of an ecosystem to return to a stable state following a disturbance is called resilience. The ecosystem can transition to a different stable state if a disturbance is strong enough to pass an ecological threshold, or tipping point. Depending on the disturbance, the threshold might be irreversible. In other cases, returning to the desired state would require extreme actions with unknown consequences.

Considering thresholds in a management plan can contribute to achieving a favourable conservation status of a species in a protected area or OECM. For evidence-based site management, critical thresholds play an essential role. A known minimum amount of habitat is required for certain animal species where below the threshold area, populations cannot thrive. If threshold responses can be identified and measured, adaptive management will become more efficient and easier to execute. It is crucial to understand the circumstances in which thresholds are likely to be crossed and the mechanisms behind threshold behaviour (Groffman et al., 2006).

Threshold values can be derived from a baseline survey or from historical findings. They can sometimes be developed by stakeholders as a management target (e.g. re-establishment of 50 km² of natural habitat). Sometimes, and unfortunately far too rarely, these thresholds are anchored in the management plan of a park. In any case, these thresholds must be taken into account in the orientation and calibration of the monitoring programme. If certain thresholds are exceeded or not reached, it is time for adaptive management actions to change.

6.8 Setting up monitoring systems: Costs and outcomes

"Monitoring is to record change" (Bayfield, 1997). Many sectors in modern societies are steered, evaluated and driven by monitoring indicators such as employment rates, stock market indices, demographic features, and so on. For decision-making, the identification of trends is much more important than obtaining simple snapshots of the indicator over time.

It is essential to understand the long-term costs and benefits of monitoring. The costs are highest at the beginning of a monitoring programme, whilst the benefits become apparent only after several monitoring cycles. The value of the data therefore increases over time (Figure 21). Funding and project continuity should be considered in a long-term context. Such an approach will maintain continuity of monitoring despite changes to personnel or funding programmes. Institutionalised support from national protected area networks and agencies is important because it can guarantee long-term programme funding. Figure 21 The initial stages of a monitoring programme come with high up-front costs. The required effort during initial cycles of a biodiversity monitoring programme must be evaluated against the value of knowledge that accumulates after several monitoring cycles. *Source: Compiled by the report authors*


6.9 Protected Area Management Effectiveness evaluation tools

In the Kunming-Montreal Global Biodiversity Framework, protected areas and OECMs are becoming more accountable for outcome-based management. A robust National Biodiversity Strategy and Action Plan (NBSAP) will provide a framework for managers to ensure that the objectives of the management plan work towards the national strategy. NBSAPs are required under Article 6 of the CBD (CBD, 1992). Site-specific planning methods are also used extensively in protected area management, such as the Conservation Measures Partnership's Open Standards for the Practice of Conservation and its software platform Miradi (CMP, 2020). These planning methods are designed to provide standardised approaches for implementing conservation projects and evaluating their outcomes, with a key focus on improving adaptive management.

Management effectiveness in protected areas and OECMs requires good decision-making, planning and implementation (Borrini, Kothari & Oviedo, 2004). PAME evaluations are necessary to understand the impact of management activities at the site level, as determined by sufficient and appropriate monitoring (Hockings et al., 2006). PAME evaluations usually use standardised forms that are completed by managers of protected areas and OECMs, with help from staff, stakeholders and policymakers. Since the 1990s, some of the most widely used PAME methodologies include the Management Effectiveness Tracking Tool (METT), the Rapid Assessment and Prioritisation of Protected Area Management (RAPPAM) and the Integrated Management Effectiveness Tool (IMET) (Bialowolski et al., 2022). To show the outcome of management interventions, protected area managers can provide periodic reports through using PAME evaluation tools (Box 17). If possible, managers should report on what the status of the protected area would be in the absence of the management activity, in other words use counterfactual thinking. Findings may support continuing the management actions, or may alternatively prompt a long-term strategic change in management. PAME evaluations should be recorded in the Global Database on Protected Area Management Effectiveness (http://tinyurl.com/yc2cr7bd).

Box 17



Integrated Management Effectiveness Tool (IMET): Implementation in central Africa

To help improve the management of protected areas and achieve conservation objectives, IMET provides an integrated framework to support planning, monitoring and evaluation processes.

IMET assesses protected area management effectiveness through analysing results of management activities. Actions that solve problems, minimise pressures or transform threats into opportunities can be continued or even applied to other protected areas. The ultimate goal of IMET is to integrate planning and evaluation to determine which management changes are necessary to meet the desired conditions derived from the long-term site objectives. If a protected area does not use good planning or a long-term strategy, IMET can be used as a dashboard to indicate which changes should be adopted and integrated in short-term and long-term planning.

In central Africa, the first IMET evaluations were implemented as a test in 2014 on a small number of pilot sites. In 2016, after a consolidation of the tool, a large campaign was organised by the Central African Forest Observatory (OFAC) with the support of the BIOPAMA programme. Institutional agreements were arranged between the Central African Forest Commission (COMIFAC) and the national agencies in charge of protected areas. As of 2018, many partners had integrated IMET into their work plans to respond to their diverse needs. Such needs include revision of management plans, development of monographs, implementation of monitoring evaluation systems, evaluation of management effectiveness, mobilisation of financial resources, and more.

To facilitate its application in different countries, the use of IMET is supported in central Africa by a team of national coaches. These experts are trained in administering IMET evaluations and analysing the results. Coaches are mobilised by OFAC or national agencies as needed for field implementation of the tool. Today, the network of IMET coaches in central Africa numbers 42 experts across 8 COMIFAC countries.

Today more than 140 IMET exercises have been carried out in 77 central African protected areas. The findings have already permitted OFAC to produce national and regional reports on the management effectiveness of protected areas in central Africa.

IMET is accessible via the OFAC website at the following link (in French): https://www.observatoire-comifac.net/monitoring_ system/imet

Source: Florence Palla, Donald Jomha Djossi and Quentin Jungers, Project RIOFAC in support of OFAC/COMIFAC, Central Africa. Picture: African elephant with tracking collar in a protected area. © Vanessa Berger



Flowering Sempervivum montanum in the Nock Mountains, Austria. © Daniel Dalton

CHAPTER 7.

A review of methods and technologies to implement efficient and effective biodiversity monitoring programmes

7 A review of methods and technologies to implement efficient and effective biodiversity monitoring programmes

Collection of biodiversity data requires significant human effort. Many traditional data collection approaches are becoming insufficient to match today's requirements for protected areas and OECMs. However, over the past few decades advanced technologies have been developed that support effective biodiversity monitoring campaigns (Table 6). These modern approaches may enhance public interest in support of conservation actions (Verma, van der Wal & Fischer, 2016), and these technologies continue to improve (Lahoz-Monfort & Magrath, 2021).

In conservation applications, digital sensors, for instance, can convey threats to wildlife in near-real-time, may promote staff and visitor safety, and can enhance visitor experiences (Hodgkinson & Young, 2016). Users may initially be wary of new technologies, thereby limiting their adoption. But once they become familiar, the technologies may become the new standard, breaking previous barriers (Weiser, 1999). Because of high up-front costs and special user requirements, automated sensors are not always a perfect solution. Therefore, a combination of automated and traditional methodologies will be important to meet future monitoring objectives (Stephenson, 2020).

Table 6 Overview of different techniques for biodiversity monitoring with examples, benefits and limitations

Technique	Example	Benefits	Limitations	
Acoustic devices and sensors	Acoustic recorder	Less restricted by activity of targeted orga- nism; suitable for long-term data collection; non-intrusive method; portable; verification by multiple experts possible	Sound / call must be unique for species identification; many battery powered; large volumes of data are produced; limited detection range; background noise lowers data quality; vandalism	
Optical devices	Camera trap	Data can be uploaded onto cloud-based network; less restricted by activity of targeted organism; suitable for long-term data collection; non-intrusive method; portable; verification by multiple experts possible	Species ID not possible for all species on picture (e.g. insects); limited field view; triggering issues; many battery powered; large amounts of data are produced; vandalism; privacy issues	
Remote sensing	Drone-based monitoring	Great value for visualisation of landscape; high resolution comparison of habitats; temporally flexible; cost-effective; accessibility of rough terrain	Ground-truthing necessary for precise data; short flight time; professional technician needed; large volumes of data are produced; regulatory compliance; privacy concerns; collision and disturbance risk	
Telemetry and tracking tools	GPS collar	Long-term method to survey species move- ment; real-time data enable early warning system of threat; data-driven conservation strategy	Capturing animals may alter behaviour; device retrieval; limited battery life; lack of generalisability; data accura- cy and availability limited in rural areas	
Olfactory devices	Pheromone detection	Non-invasive; find sources of pheromones or act as a source / attractant for targeted organisms	State-of-the-art sensors still in experimental stage; electronic noses in use especially in chemical analysis and safety-related issues mostly not calibrated for nature conservation; limited sensor range; hard to locate source of pheromones	
Genetic methods	Environmental DNA sampling	DNA can be obtained from environmental samples (water, soil, air) as well as other organic matter such as feathers, dung, hair etc.; non-invasive; specific species detection; overview of species of taxa; analysis of large- scale surveys	Some expertise in sterile sampling to be acquired; specialised equipment and laboratory work required; data analysis and management might be challenging; spatial and temporal variability; sample can degradate or get contaminated; lack of reference data	

Source: Compiled by the report authors

7.1 Early bird or late adopter: Drivers and barriers of technology deployment

Adoption of a new innovation is influenced by many factors, including the type of innovation, how and for whom it is designed, user acceptance and cultural contexts in which the innovation is used (Mascia & Mills, 2018; Rogers, 1995). In conservation, a new practice will only be adopted if it is expected to help managers and field staff achieve their goals (Pannell et al., 2006).

In protected area and OECM management, the use of digital tools is poised to revolutionise biodiversity monitoring (Joppa, 2015). Often, different target organisms require unique sets of tools and methodologies to effectively monitor their status, and many expert-based classic approaches are suitable to monitor species. But there are many positive aspects to using new approaches for biodiversity monitoring. Foremost, by using remote data collection the presence of an expert is no longer required for field work. High-tech devices can provide a simplified workflow whilst allowing rapid collection of greater amounts of data than traditional observerbased approaches (Arts, van der Wal & Adams, 2015). Objective and standardised automated data collection may allow improved data comparability across protected area network sites. Such devices reduce the human impact of monitoring by minimising the required number of site visits, allowing documentation of timid or elusive wildlife species without the presence of field workers. Knowledge exchange and enforcement activities can be improved through the use of online dashboards powered by Internet of Things technologies. Internet of Things describes a network of sensors that are connected to the internet and can communicate with each other and send the data to a defined location. Digital workflows lead to more costeffective monitoring programmes in the long term because many technologies allow for a reduction in the number of personnel field days, freeing up resources for other purposes. Large biodiversity monitoring networks may also have the purchasing power to acquire devices at bulk prices, whilst institutions can provide incentives for implementing technologies through research funding, educational programmes and promotional opportunities. Partnering with research programmes may allow protected areas or OECMs to be early adopters of new technologies.

Despite the benefits, many barriers to digitalisation exist. It is challenging to determine the true value of a new technology because its effectiveness is often reported in the best light (Arts, van der Wal & Adams, 2015). Costs of digital technologies remain an obstacle for many management programmes, particularly in the developing world (Stephenson, 2020). High-tech innovations may fail to capture the cultural elements of a site or may exclude untrained groups (Cole & McCallion, 2018). Practitioners may be sceptical or distrustful of new approaches, instead favouring techniques that are more familiar to them. Data analytical infrastructure and reliability of the technology may be poor, particularly in remote areas. The large volumes of collected data introduce potential problems of data management. Continual improvement of technologies is a natural process; however, potential future changes in the workflow may create uncertainty for managers around adopting new tools (Ferrari et al., 2022).

The labour savings and quality improvement of using automated devices may easily make up the cost difference within a few cycles of use. Identifying how a digital workflow has operated in other sites can help managers adapt a tool to their own needs. Yet, high-tech solutions may not be realistic in many areas of the world or may not fully replace long-standing methodologies for certain indicators.

7.2 Toolkit: Overview of the indicators

Biodiversity monitoring has historically relied on trained field personnel to collect data, often in paper formats. Today, the need for monitoring outpaces the availability of trained staff. Restrictions on staff time may limit the scope of data acquisition, and storage of raw data in the form of paper records is difficult to archive. Field work is often implemented in habitats that are occupied by wildlife, thus posing a threat to the safety of observers and animals alike, or may change the very behaviours that are the focus of the monitoring effort (Verma, van der Wal & Fischer, 2016). Combining multiple methods simultaneously using specialised equipment may be an effective solution but is logistically challenging (Prosekov et al., 2020). However, traditional approaches will continue to make important contributions to biodiversity monitoring. This is because conventional data collection methods are often robust and fail-safe, cheap and easily implemented compared to high-tech methods.

Many traditional approaches to biodiversity monitoring are labour-intensive activities that target a specific group of organisms. In this chapter, a brief review is provided of the commonly applied monitoring methods for specific monitoring targets.

- plant communities;
- flying vertebrates;
- ground-dwelling vertebrates;
- arthropods;
- soil fauna;
- freshwater organisms;
- marine organisms.

7.2.1 Plant communities

Vegetation monitoring combines the identification of individual plant species with a measure of abundance or species diversity across a spatial area and over time. Timing of a monitoring cycle is key because surveying at different phenological stages can change the interpretation of findings (Pauli et al., 2015). Plants are mostly immobile; therefore, permanent plots with fixed positions offer the possibility to detect fine-scale vegetation changes over time (de Bello et al., 2020). The selection of an appropriate survey method is dependent on the management questions and the required precision and accuracy (Schulz, Bechtold & Zarnoch, 2009).

Simple presence / absence vegetation surveys can detect trends in a plant community. Abundance measurements may provide more precise estimations of change. Many suitable methods are used to estimate abundance (Braun-Blanquet, 1964), including visual plant cover estimation, point-frequency counts and subplot-frequency counts (Bråkenhielm & Qinghong, 1995). Using abundance classes has the advantage of allowing rapid quantification, especially for larger areas (Ricotta & Feoli, 2013). Detection of change in abundance over time is not very accurate compared to other measurements (Irvine & Rodhouse, 2010).

Selecting the size of sample plots in vegetation monitoring depends on the vegetation type and the target indicator. Knowledge of the indicator and its habitats will help managers determine the minimum mapping unit within the area of interest. Monitoring plots in open areas and low-lying vegetation, such as tundra, use plot sizes of about 1 m², whilst dense forest plots usually require larger plot sizes of several hundreds or thousands of square metres, as in the case of tropical forest monitoring (e.g. Picard et al., 2010). The shape of a monitoring plot additionally varies from quadrats (e.g. in grassland vegetation) to circles, which are often used in forest monitoring (Paul, Kimberley & Beets, 2019). Square- or rectangle-shaped plots are preferable when cover of many plant species should be estimated, as the straight boundaries can be easily marked by corner posts or measuring tapes (Elzinga, Salzer & Willoughby, 2019). Circular plots are useful when the number or attributes of large plants such as shrubs and trees should be assessed because only the centre point needs to be marked and the distance to the individual has to be measured. Linear transects are a useful option in areas where there is a sharp gradient of species (Pauli et al., 2015).

7.2.2 Flying vertebrates

Birds and bats can be identified through visual recognition, vocal identification and through indirect evidence. For visual recognition, birds and bats may be captured using mist nets that are placed in flight paths or tree canopies. Captured birds can be fitted with small leg bands that are marked with a unique identifying code. If a banded bird is recaptured, scientists can determine movement based on where or when the animal was previously handled. Harp traps are modified mist nets used to capture bats (Hoffmann et al., 2010). Captured bats can be fitted with semi-permanent light-tags (Buchler, 1976) or small radio transmitters (Timofieieva et al., 2019) to track movement behaviours. Permits are generally required to capture birds and bats, both for the animals' and the handlers' safety.

Less invasive techniques can be used to determine bird and bat activity. These techniques require a basic level of field training to be confident in the findings. Transect walks and visual point counts at roosting or nesting sites are standard approaches (Schieck, 1997; Stahlschmidt & Brühl, 2012). Vocal identification occurs through expert recognition of calls in the field, in combination with territory or spot mapping. Field playback of bird species calls is also used for monitoring because animals often respond to calls (Gregory, Gibbons & Donald, 2004). Modern state-of-the art sound recording devices are now used across the globe. So-called passive acoustic monitoring uses removable memory cards, and devices can also be linked to a cloud-based data repository. Computer algorithms can automatically determine species by their calls. Experts can then verify the species identities in the office. Metadata gathered during

recording events provide information on the audio device, time of day and the seasonality of species.

Another form of non-invasive monitoring of birds and bats is the collection of physical samples such as spoor prints, faecal pellets, hair or feathers. The biological samples may allow food web analysis. Genetic techniques are now available to support analysis of physical collections.

7.2.3 Ground-dwelling vertebrates

Medium and large animals can be tracked visually or through telemetric approaches (Fuller & Fuller, 2012). Movement behaviours of large animals can be mapped through the use of small aircraft or drones (Prosekov et al., 2020), through tagging of individual animals (Verma, van der Wal & Fischer, 2016) or through following spoor marks (Pirie, Thomas & Fellowes, 2016). Signs such as dung or fur can provide DNA for indirect evidence of animals, including food web associations (Skrbinšek et al., 2019). One common method for long-term large animal observation is to capture the animals in box traps or pens, fit them with radio collars or GPS tags and then release them for tracking. This approach requires professional expertise to minimise animal stress.

Small animals can be captured through live trapping. Trapping is used for species identification, analysis of diet and to determine the level of parasitism, among other applications. The technique can also be applied for collection of DNA (Hoffmann et al., 2010). This approach is particularly effective when traps are placed on a transect line that is situated along a drift fence. A major drawback of live trapping is the stress that can be placed on the animals. Field technicians must frequently check the traps so that captured animals do not starve or become dehydrated.

The use of camera traps is a passive way to document species (Rovero & Kays, 2021). Visual evidence of the animal is obtained through wildlife cameras. Camera trapping is now a well-established monitoring technique for warm-blooded animals. Time-lapse techniques and innovations in motion detection are poised to allow camera traps to document cold-blooded and small animals (Rovero et al., 2013). Acoustic sensors can also be deployed to survey for the presence of animals in the landscape. Computer algorithms may automatically identify species documented from camera traps and acoustic sensors, but expert verification is still required. Metadata from digital collections are automatically stored, giving information on the phenology and diurnal activity of species.

7.2.4 Arthropods

Field monitoring of insects and other arthropods requires multiple techniques to describe the diversity of even one taxonomic group in the environment (Magurran, 2004). Common approaches include using attractants such as light, colour, plant volatiles or pheromones that lure insects into collection containers. Non-attractant methods include sticky traps, flight interception traps and immunomarking-recapture techniques (Grootaert et al., 2010; Hagler, 2019). Artificial nesting aids can be effective for inventorying species of bees, wasps and spiders. Active collection techniques include using aspirators or vacuums, sweep nets, hand collection or visual observation. Passive techniques include using coloured pan traps, pitfall traps, emergence traps and Malaise traps. An advantage of physically collecting insects is that specimens can be sent to expert taxonomists for identification. Physical samples are also necessary for genetic identification. A main disadvantage is the lethal nature of many collection techniques. In some cases, photographs from the field may aid in the identification of insects. Images must clearly show the distinguishing features of the organism, and therefore a basic level of taxonomic knowledge is required for useful photographs. Many programmes using machine learning for photographic verification of arthropods are now in operation.

7.2.5 Soil fauna

Soil fauna can be collected with Berlese funnels and through direct searches of the environment. The diversity of soil fauna is influenced by environmental factors, for example the amount and age of coarse woody debris on the forest floor (Siitonen, 2001). Pitfall traps are commonly used to collect soil-dwelling arthropods. Baits can be added to pitfall traps to increase attractivity of the trap to a target species, and screens can be added to prevent by-catch of non-target animals. Pitfall traps are also used for collecting arthropods in caves, rock crevices and other hard-to-reach habitats (Nitzu et al., 2010). Taxonomists can identify physical specimens, whilst genetic analysis of soil samples can provide evidence of soil community composition (Oliverio et al., 2018). Genetic analysis can be outsourced to specialised third-party laboratories.

7.2.6 Freshwater organisms

Aquatic species are observed primarily through collection. Netting is the primary technique for sampling standing waters such as lakes, ponds and reservoirs. Different net types target certain types of animals. Deploying the appropriate mesh size will help capture target organisms and decrease by-catch (Stark et al., 2001). After removal from water, invertebrates and microfauna are sorted by sieves and preserved in ethanol. Collected specimens can be sorted morphologically using a hand lens in the field, or with stereomicroscopes in the laboratory. Grab samplers and traps baited with meat or dairy are effective ways to sample fish. Many techniques allow catch-and-release of larger organisms.

In freshwater ecosystems that cannot be fully accessed on foot, collection methods include electrofishing and sink nets. Algae can be sampled by planktonic nets. Biofilms may be sampled directly by skimming objects from the water surface. After collection, species are usually directly identified in the field or in the laboratory based on their morphology. In addition, species may be detected and identified with genetic methods. DNA can be obtained directly from target organisms, from bulk samples of invertebrates, or from DNA in the environment. Techniques for collecting environmental DNA (eDNA) have greatly improved in recent years.

When sampling multiple water bodies for comparison, measurement of water temperature, water current, pH, conductivity and oxygen saturation should ideally accompany the biological collections. These abiotic parameters can be measured using specific devices. In addition, water pollutants can be measured, including nutrient levels, heavy metals, pesticides and pharmaceuticals. Based on the collected organisms, water quality may be correlated with anthropogenic pressures (Dallas, 2021; Poikane et al., 2016).

7.2.7 Marine organisms

Monitoring marine ecosystems is more demanding and expensive in comparison to freshwater methods due to the unique challenges of the marine landscape (Cato et al., 2006). As with terrestrial ecosystems, no single set of monitoring approaches is sufficient to describe the full range of organisms present in a marine ecosystem. To guide data collection on the most important indicators, panels of experts at the Global Ocean Observing System have developed Essential Ocean Variables (EOVs). These EOVs are a set of physical, biochemical and biological measurements that are used as a guide to collect data on the most important indicators of marine ecosystems (Canonico et al., 2019). Methods used for monitoring the biological EOVs include satellite remote sensing, plankton net tows and trawl net surveys, metagenomics, acoustic telemetry, and visual or video underwater surveys (Muller-Karger et al., 2018). The use of passive acoustic monitoring on marine animals is applied to survey abundance and migration of large marine animals (Malinka et al., 2018).

7.3 Technology-based approaches for biodiversity monitoring

In recent years, development of advanced sensor-based devices has dramatically improved capacities for biodiversity monitoring. Examples include acoustic sensors for passive acoustic monitoring; optical and passive infrared sensors for camera trapping; Earth observation sensors for remote sensing; and GPS sensors for wildlife telemetry (Lahoz-Monfort & Magrath, 2021). The following sensor-based technologies are discussed:

- acoustic devices and sensors;
- optical devices;
- remote sensing;
- telemetry and tracking tools;
- olfactory devices;
- genetic techniques.

7.3.1 Acoustic devices and sensors

Recent improvements in passive acoustic monitoring now allow identification of birds, bats, amphibians, insects and even chainsaws or gunshots in the environment (Darras et al., 2019; Sethi et al., 2020; Sugai et al., 2019). Modern acoustic devices are small, portable and are designed to blend into the environment. Acoustic devices do not restrict the activity of the target organism. They can detect sounds within the range of human hearing, as well as sounds

that cannot be heard by people (e.g. ultrasonic frequencies for detection of bats, geophones for detection of marine animals). Devices can be programmed to function continuously, can be activated through incoming sound, or can be programmed to be active at specific times (Lahoz-Monfort & Magrath, 2021). In aquatic ecosystems, acoustic telemetry is advancing as a primary technique to track the behaviour of animals. This technique uses small tags that are attached to an animal, and a stationary receiver that collects data on the animal's movement (Matley et al., 2022).

Optimal distribution of audio sensors in terrestrial landscapes depends on the targeted animal community, landscape barriers, weather conditions and the equipment used for detection. The signal-to-noise ratio is the main factor determining the range at which a microphone can detect a sound. High signal-to-noise ratios allow greater detection of the target animal (Darras et al., 2020). In general, traps spaced less than 10 m from one another will effectively capture most sounds in an area-wide arrangement, with louder sounds detectable at greater distances (Hill et al., 2018; Sethi et al., 2020).

Networks of audio devices typically generate several hundred trap-days of data per deployment period. This large volume of data requires sophisticated procedures for data analysis. With improvements in computer technologies, passive acoustic monitoring is now assisted through automated species identification. Many open-source programmes and proprietary software are available for acoustic data analysis (Priyadarshani, Marsland & Castro, 2018). Artificial intelligence algorithms help to identify calls based on characteristics of the sound profile, visualised as a spectrogram. Some programmes can group all readings of a similar pattern together. So-called cluster analysis eliminates non-target sounds, reducing the size of the dataset and helping an expert to verify the findings.

7.3.2 Optical devices

Biodiversity assessment using camera traps has undergone sophisticated changes since the technique's beginnings in the 1890s (Box 18) (Kucera & Barrett, 2011). Modern camera trapping uses passive infrared sensors that are installed onto digital cameras. When an animal passes in front of a sensor, the camera is triggered to take a photograph or a short video. Data files are downloaded from internal memory cards or can be uploaded directly onto a cloud-based network (Wearn & Glover-Kapfer, 2017). Many brands of trail camera are commercially available. Trail cameras may have customisable features allowing users to change settings including the sensitivity of the sensor and image resolution. However, there is a clear trade-off of quality between lower-cost and more expensive devices (Newey et al., 2015).

Camera trap arrays are normally placed in a grid pattern but can be placed opportunistically, depending on the goals of the monitoring programme. If possible, camera stations should be located at least 2,000 m away from one another. Cameras should be placed within 500 m of the predetermined grid coordinate. This spacing ensures independent data collection (Abrams et al., 2018). For mammals, cameras should be installed to have a focal distance between 1.2–4 m. A total of 1,000 or more camera trap-days is sufficient for most studies (Rovero et al., 2013). Therefore, a camera trap array of 50 cameras needs to be in place for only about three weeks to document the biological community. The phenology of the target animal will affect decisions on when to begin the trapping period (Dalton et al., 2022).

There are several limitations to camera trapping. Camera traps detect moving animals. Depending on their route past the sensor, not all animals will be captured in the images. This is called 'imperfect detection' and must be considered during the sampling design stage. An open access journal article by Burton et al. (2015) provides further references on field design, including targeting uniquely patterned animals, unmarked animals, occupancy modelling, multi-species surveys, and incomplete coverage of the area of interest. Additionally, because passive infrared sensors work based on body heat of the target animals, standard camera trapping is not very effective at documenting amphibians, reptiles or invertebrates. Insects can be documented using time-lapse photography (Collett & Fisher, 2017). Other types of motion sensors have been used for remote detection of small or cold-blooded animals (Hobbs & Brehme, 2017). As with acoustic recording devices, monitoring using camera traps generates large volumes of data that require sophisticated computer analysis for interpretation (Norouzzadeh et al., 2018).

In marine systems, camera and video technologies are used for continuous, long-term monitoring of aquatic organisms. Video cameras and drop mount cameras can be mounted on remotely operated vehicles and autonomous underwater vehicles. These vehicles are equipped with additional sensors to gather information about the benthic environment. In shallow waters, divers can also collect photos of species and the aquatic environment (Danovaro et al., 2016).

Box 18



High-tech approaches: Camera trapping for biodiversity monitoring

Camera trapping is the use of motion-sensitive cameras to record images or videos of animals passing in front of them. Today this technology represents the preferred method for several types of wildlife studies (Steenweg et al., 2016), with growing scope and applications in conservation science and practice (Rovero & Kays, 2021). Medium-to-large (i.e. above 500 g), ground-dwelling mammals remain the optimal target.

Camera traps have many advantages, including the ability of trail cameras to work remotely, their costefficiency, and ability to collect large amounts of data. The TEAM project illustrates these advantages. TEAM is a network of 17 sites of tropical forest protected areas that have collected standardised camera trapping data using a strict sampling design since the mid-to-late 2000s (Rovero & Ahumada, 2017). The design consists of arrays of 60 sampling sites that are surveyed annually for a minimum of 30 days. Results have provided temporal trends of wildlife populations at an unprecedented resolution (e.g. Beaudrot et al., 2016). TEAM has also been instrumental in developing the Wildlife Picture Index (O'Brien et al., 2010). This index uses data from camera traps to detect changes in the presence of different species and the number of species in a community. This metric is officially recognised as an indicator by the Convention on Biological Diversity for Sustainable Development Goal 15 and other targets along with the Living Planet Index and the Red List Index (https://shorturl.at/cdB35). In the current digital era, the

most prominent developments have been in tools and software to store, manage, annotate and analyse the data. Multiple solutions are available to process the images (Young, Rode-Margono & Amin, 2018), including stand-alone desktop tools (CPW Photo Warehouse), cloud-based solutions (eMammal, Wildlife Insights, Conservation AI) and crowdsourcing to identify animals (Zooniverse). Artificial intelligence tools are being developed to automatically identify species in images (He et al., 2016; Norouzzadeh et al., 2018), as well as including dashboards to visualise findings. These approaches are now being integrated into data management platforms such as Wildlife Insights. These advances help improve biodiversity monitoring, as practitioners and protected area managers now have the means to master the entire process from data collection to reporting the findings.

Camera trapping continues to have a bright future in biodiversity monitoring, although some limitations still exist including patchwork regulations by public management agencies or private landowners, and challenges related to providing open access to photographic libraries.

Source: Francesco Rovero, University of Florence, Italy. Picture: Camera traps allow managers to get up close to wild animals in their natural habitats. © Instituto Chico Mendes de Conservação da Biodiversidade

7.3.3 Remote sensing: Applying geo-information, geo-statistics and geo-modelling tools

Earth observation refers to gathering information about the Earth's surface using remote sensing tools, without being in physical contact with the ground or water. This approach is ideal for habitat mapping of terrestrial and marine areas. Aerial vehicles can be equipped with customised packages of sensors to monitor animals and plants. Drone-based monitoring, coupled with artificial intelligence approaches, represents a new avenue of animal tracking and is particularly promising in terms of geospatial analysis (Prosekov et al., 2020). Satellites, drones and aeroplanes can capture remote sensing data using wavelength sensors or ranging sensors. Wavelength sensors determine reflectivity of the surface and include visual RGB, near infrared, thermal vision, multispectral and hyperspectral sensors. Ranging sensors such as LiDAR or ultrasonic sensors emit pulses of energy that bounce off an object's surface and return to the device. Ranging sensors provide a 3-D digital reconstruction of the object. Low-flying drones or aeroplanes can be equipped with high-resolution onboard sensors. These arrays can provide greater accuracy and detail than satellite imagery (Martínez-López et al., 2021).

Because wavelength sensors can capture spectra that are not visible to the human eye, they offer a different way to visualise the landscape based on canopy patterns. Modern data mining programmes can process spectral information to identify landscape trends (Bailey et al., 2017). This is called geostatistical analysis. Geo-modelling techniques can provide information on abiotic variables, can outline habitat borders and topographic gradients, and can identify plant phenology (Martínez-López et al., 2021).

Earth observation-based approaches may permit using the same dataset to compare habitats in different protected areas and OECMs. They may also help interpret changes in the environment based on changing climates and land use (Carilla et al., 2013). Data going back in time are often useful to gain an understanding of the environment before a major disturbance. Historical series of satellite data may allow a retrospective analysis of landscape changes over time (Szantoi et al., 2016). Pre-satellite aerial photographs can provide similar information and are available for many areas of the world (Seimon et al., 2017). Satellite- and UAV-based sensors are being designed to harmonise data collection allowing comparison of Essential Biodiversity Variables between sites and networks (Skidmore et al., 2021). Earth observation technologies are currently undergoing a rapid evolution. The technologies are well positioned to create new opportunities for biodiversity monitoring.

Remote sensing is often outsourced to professional technicians because of the required expertise, the large range of applications and the high expense. The appropriate sensors for the monitoring programme must be identified in advance of data collection. Ground-truthing of remote sensing data is required for modelling or validating Earth observation data (Danovaro et al., 2016). If validated, use of satellite or laser scanning data provides an efficient alternative to field monitoring, particularly in remote or inaccessible areas (Box 19).

Box 19

High-tech approaches: Airborne laser scanning for forest biodiversity assessment in an Austrian national park



LiDAR is a laser scanning approach that uses pulsed laser light to determine the 3-D structure of a landscape. Forests in steep, mountainous terrain face unique pressures that shape their biodiversity. One pressure is the impact of avalanches on forest structure. In Gesäuse National Park, Austria, a forested area that experienced an extreme avalanche was assessed using LiDAR scanning data from 2010 and 2020 (Berger, Kirchmeir & Hirschmugl, 2021). First, a digital terrain model was generated from the 2010 dataset, allowing a baseline measurement of topography. Next, digital surface models of the vegetation were extracted from both datasets. The difference in vegetation height was used to model forest stand regeneration. Based on this result, different areas were assigned to represent growth, knocked down and destroyed trees. The site was ground-truthed to confirm the apparent cause of vegetation change. Trees that had

fallen were facing downhill, following the direction of the avalanche track.

Overall, areas of tree growth were concentrated in pockets that are protected from avalanches by topography. Large stands of trees offer protection against avalanches, limiting the impact on forest structure, whilst small stands of trees are vulnerable to being completely destroyed. Frequently affected areas tend to be grassland habitats with only young trees. The dynamics of the avalanche track create a natural mosaic landscape that increases the diversity of the forest vegetation.

Source: Vanessa Berger, Carinthia University of Applied Sciences, Austria. Picture: Trees knocked down by an avalanche in Gesäuse National Park. © Vanessa Berger

7.3.4 Telemetry and tracking tools

In recent years, miniaturisation of devices has expanded the applications of wildlife tracking using telemetry. Using commercially available microprocessors such as Arduino units, lightweight GPS modules can now be custom-built for small- to medium-sized animals (Cain & Cross, 2018). Experimental transceivers may even effectively track insects of conservation value including beetles, crickets, bees, dragonflies (Kissling, Pattemore & Hagen, 2014) and butterflies (Wang et al., 2015). This is an area where sensor-based technology is poised to improve. Telemetry tools have many benefits for tracking movement of species, particularly in remote areas. Before using telemetry tools in the field, a number of factors must be considered. First, managers should be certain about which type of tracking device is most appropriate for their purposes. Capturing wild animals will cause stress that could change their behaviour. The logistics of pursuing the marked animals should be considered, including recapture for data collection and device retrieval. Further, the range of motion of animals is dependent on the individual. A review of radio telemetric approaches for tracking carnivores is given in Fuller and Fuller (2012).

7.3.5 Olfactory devices

Plants and animals produce unique signatures of volatile organic carbons that differ between healthy or stressed individuals. Volatiles can be determined in the laboratory through a number of traditional techniques. These approaches are impractical for most biodiversity monitoring programmes. State-of-the-art olfactory sensors, or electronic noses, are under development for field use. So-called GC-IMS sensors can collect plant volatiles in the field for up to four hours without external power (Wägele et al., 2022), but are relatively large and therefore limited in where they can be installed. Other hand-held devices are still in an experimental state for conservation but are poised to allow rapid field assessment of plant health without highly technical expertise. One promising device is the C-320 e-nose that is already used for many diverse applications including food quality assessment, agriculture and forestry, and biomedical applications, among others (Karakaya, Ulucan & Turkan, 2020). Machine learning is used to recognise signature patterns of the volatiles. In the coming years, improved pattern recognition, in combination with miniaturisation of the hardware, will support the wider use of e-noses directly in the field (Cui et al., 2018).

7.3.6 Application of genetic methods in biodiversity monitoring

Genetic approaches are promising techniques for biodiversity assessment in protected areas and OECMs (Pascher, Švara & Jungmeier, 2022). These techniques can detect single or multiple species based on the sequences of DNA in a sample. DNA can be obtained directly from individuals, bulk samples containing multiple organisms, or environmental samples including water, soil and air (Carvalho et al., 2019).

Nowadays, several guidelines are available, presenting the best approaches for species detection and identification based on genetic approaches (e.g. Bruce et al., 2021; Minamoto et al., 2021; Pawlowski et al., 2020). Community assessment can be performed using

metabarcoding approaches (Lim et al., 2022; Taberlet et al., 2012b). Metabarcoding is extremely fast and applicable to species detection and assessment of environmental quality (Aylagas et al., 2016; Ruppert, Kline & Rahman, 2019). The processing of environmental samples and metabarcoding will usually be outsourced to expert laboratories because the technique requires specialised expertise and equipment (Figure 22).



Collecting DNA directly from the environment can significantly improve monitoring efficiency and success by detecting species that are difficult to detect using traditional approaches (Taberlet et al., 2012a). One major advantage of eDNA collection is that sampling depends relatively little on the developmental stages of the sampled groups (Takeuchi et al., 2019). For reliable determination of species presence, a good temporal and spatial sampling plan needs to be developed (Erickson, Merkes & Mize, 2019) which makes consultation with experts necessary. With appropriate training, sample collection can be performed by non-experts (Biggs et al., 2015). Sampling of eDNA can be standardised, for instance the same volume of the medium can be collected and standardised equipment (e.g. filters) can be used. The exact collection methods will depend on the environment in which sampling will occur, but the analytical steps are largely identical between terrestrial, freshwater and marine ecosystems.

Certain limitations of eDNA sampling need to be considered to avoid incorrect species detection. Assessment of species abundance is still challenging and cannot be directly estimated from sequence number because the amount of DNA from each sample may vary considerably due to sample and analytical biases (Bista et al., 2018). In certain environments, samples of eDNA may be transported over long distances (Deiner & Altermatt, 2014). Windy weather conditions may also transport genetic materials (Lim et al., 2022). DNA may be preserved in soils and sediments for several days to years, potentially adding uncertainty to the findings (Foucher et al., 2020). Due to possible sample contamination and DNA degradation, using sterile DNA-free equipment and correct storage and transportation are essential. Correct sequence detection largely depends on the completeness of barcode reference databases. Well-established databases include Barcode of Life, DNA DataBank of Japan, the European Nucleotide Archive, and NCBI GenBank (Lawniczak et al., 2022). Many additional national databases are coming online, including the Norwegian Barcode of Life, German Barcode of Life and Austrian Barcode of Life (Weigand et al., 2019).

7.4 Analysis: Applying advanced computer technologies for big data

The development of digital technologies in conservation has resulted in a fast-growing volume of data. These data have great potential to revolutionise scientific understanding of the world's biodiversity. In the context of modern computing, data can be used in machine learning approaches, analysed through data mining, shared to generate more knowledge, and safely archived for future use.

7.4.1 Online data platforms, automated recognition and data logging

Online platforms are becoming more important than ever before for the management of big data in conservation. Real-time tracking tools have come into widespread conservation use with the rise of digitalisation. Tools such as Global Forest Watch can help recognise changes in vegetation and landscape structures. In citizen science networks, automated recognition software is now mainstream, enabling apps such as iNaturalist to have the capacity to document species on an enormous scale. Species identification apps are based on a

Figure 22 Biodiversity assessment using genetic methods follows a uniform workflow. Samples are collected in the field and returned to the laboratory. Then, DNA is extracted from the samples, amplified and sequenced to detect target organisms. The result is a list of DNA sequences that were present in the samples. Using bioinformatic analysis, species can be determined based on sequence assignment to the reference databases. Source: Compiled by the report authors

combination of artificial intelligence algorithms and crowdsourced for expert verification (Silvertown et al., 2015). Some examples include Flora Incognita, iNaturalist, iSpot and many more (Table 7). Data entry programmes like QField and CyberTracker are valuable for collecting and organising large-scale monitoring data in the field. The number of available apps will continue to expand in the future, and the value of high-tech conservation solutions will increasingly depend on capacities to handle the large volumes of data (Lahoz-Monfort & Magrath, 2021).

Торіс	Name	Website	
Biodiversity	eMammal	https://emammal.si.edu/	
image archiving	Wildlife Insights	https://www.wildlifeinsights.org/	
	BirdTrack	https://shorturl.at/bctz6	
	eBird	https://ebird.org/home	
Citizen Science platform	iNaturalist	https://www.inaturalist.org/	
	Merlin	https://merlin.allaboutbirds.org/	
	Zooniverse	https://www.zooniverse.org/	
	Barcode of Life	https://www.boldsystems.org/	
Constis seguence	Basic Local Alignment Search Tool	https://blast.ncbi.nlm.nih.gov/Blast.cgi	
repository	European Nucleotide Archive	https://www.ebi.ac.uk/ena/	
	DNA DataBank of Japan	https://www.ddbj.nig.ac.jp/index-e.html	
	NCBI GenBank	https://www.ncbi.nlm.nih.gov/genbank/	
	CyberTracker	https://cybertracker.org/	
Mobile data collection	Open Data Kit	https://getodk.org/	
	QField	https://qfield.org/	
	Copernicus Sentinel (ESA)	https://shorturl.at/hltEQ	
Satallita imagany	Global Forest Watch	https://www.globalforestwatch.org/	
Satellite illiagery	Landsat (NASA)	https://landsat.gsfc.nasa.gov/	
	MODIS (NASA)	https://modis.gsfc.nasa.gov/	
	Flora Incognita	https://floraincognita.com/	
	iNaturalist	https://www.inaturalist.org/	
Species	iSpot	https://www.ispotnature.org/	
identification	Pl@ntNet	https://identify.plantnet.org/	
	Plant.id	https://plant.id/	
	PlantSnap	https://www.plantsnap.com/	

Table 7 A selection of apps and data logging tools for modern biodiversity assessment

Source: Compiled by the report authors

7.4.2 Machine learning

A decades-old branch of computer science, machine learning can be applied to many data science problems. Many different types of learning can occur, with separate algorithms behind each type (Christin, Hervet & Lecomte, 2019). A task is assigned to the computer, for example recognising an animal in a photograph. The algorithm learns through 'supervised learning'. In all cases, large datasets produce better algorithms because more experience is gained (Alzubi, Nayyar & Kumar, 2018). In practice, machine learning uses part of a dataset as training data, and the remainder of the dataset for model testing. The performance of the algorithm is evaluated by an expert, and the computer algorithm uses the feedback to later help complete a similar task.

Deep learning is a particular type of machine learning that uses either 'supervised learning' or 'unsupervised learning'. Currently, most algorithms use one of many types of supervised

learning. Unsupervised learning does not require training data and can function even if data points are missing (Christin, Hervet & Lecomte, 2019), eliminating the requirement for labelled data. Unsupervised learning is therefore used to find underlying patterns or groupings in a dataset without explicit guidance. Deep learning can be used for identification and classification of digital signals, behavioural studies, population monitoring, ecological modelling and conservation management. Whilst the setup of a deep learning framework is complex, its ability to analyse very large datasets is unmatched. Partnering with computer scientists will allow application of machine learning and will also establish synergies benefiting protected areas, OECMs and collaborating institutions alike (Carey et al., 2019).

7.4.3 Data mining

Whilst machine learning is 'learning-driven', data mining is 'discovery-driven' (Tuysuzoglu, Birant & Pala, 2018). Data mining is an approach for biodiversity assessment that brings together information on species or habitats from multiple sources. Data mining has strong application when the knowledge of a system is low but large volumes of data exist. It is therefore exploratory in its application. The main use for most data mining methods is to discover patterns within a dataset, even when the parameters are unknown (Box 20). Methods are suitable for analysis of large datasets, or for datasets with many predictor variables. Discovered correlations should be confirmed through advanced statistics (Hochachka et al., 2007). For biodiversity monitoring, historical data can be combined with more recent information to help determine how species diversity has changed over time and space. Cause-and-effect relationships do not have to be known in advance for data mining to identify interactions. Differences in spatial resolution of different programmes usually make data mining appropriate for large-scale comparisons rather than for local features.

Managing and interpreting large sets of data are significant challenges. A drawback of data mining is that managers cannot be assured of equal data quality between studies because of methodological differences. Another challenge is the development of good algorithms to describe the feature being assessed (O'Sullivan et al., 2010). Because data mining is driven by algorithms, no two programmes will produce identical results (Hochachka et al., 2007). Therefore, a combination of independent classifiers, each detecting different patterns in the data, may indicate the best way to predict the next results (Tuysuzoglu, Birant & Pala, 2018).

Box 20



High-tech approaches: Terrestrial laser scanning for forest biodiversity assessment

Terrestrial laser scanning is a fully automated technique that can generate point clouds of more than 1 million data points in 3-D space, capturing the physical structure of a forest. In Ireland, afforestation of the landscape is occurring due to increased cultivation of plantation forests. Plantation forests constitute a potentially significant change in the biological community of Irish forests, but the exact effects are unknown. A data mining system was applied to terrestrial laser scanning point cloud data to address whether biodiversity measurements could be predicted based on forest structure. Findings would help the country comply with international regulations for biodiversity.

First, arthropod and bird species diversity counts were conducted in plantations and native forests. Terrestrial laser scanning point clouds were then generated at all sites. Point clouds identified the number and position of trees, their height and stem diameter at 10 cm height intervals. To determine relationships between species richness and the woodland parameters, five statistical procedures were applied using the open-source data mining system Weka.

Through data mining, it was found that the number of beetle species could be accurately predicted across 37 forest locations based on forest stand age and the average tree stem diameter. Using this approach, future forest inventories may be conducted using only terrestrial laser scanning, an automated procedure that will require less expertise to conduct the field work (O'Sullivan et al., 2010).

Source: Compiled by the report authors. Picture: Rendition of a forest after terrestrial laser scanning. © Vanessa Berger



Laser scanning in Rohrach Nature Reserve, Vorarlberg, Austria. © Vanessa Berger

CHAPTER 8.

Synthesis: A new age of biodiversity monitoring

8 Synthesis: A new age of biodiversity monitoring

As national and international biodiversity reporting requirements become more rigorous, managers of protected areas and OECMs are becoming increasingly accountable for outcome-based management activities. This publication proposes a comprehensive framework and workflow for developing biodiversity monitoring programmes in protected areas and OECMs. The framework promotes robust investigation of the state of relevant indicators, supporting effective biodiversity monitoring systems and adaptive management. It is written for managers of marine, coastal, terrestrial and freshwater protected areas and OECMs.

The ambitious call to action of the Kunming-Montreal Global Biodiversity Framework and other international agreements highlights the importance of an internationally recognised biodiversity monitoring framework that can be applied in protected areas, OECMs, and national or international monitoring networks. The monitoring framework supports consistency in the decision-making processes for which good methodological and technical preparation is mandatory. Long-term and well-structured monitoring programmes should be considered the 'gold standard' because they become more valuable with every additional dataset. Longterm monitoring provides greater understanding of the natural world with each passing monitoring cycle. Many monitoring cycles are required before the benefits of the programme become apparent. It is therefore important to invest enough resources at the beginning of a programme for the purpose of gaining knowledge in later cycles. Effective and continued communication between managers, decision-makers and local stakeholders is required to sustain a monitoring programme in the long term.

A unified approach to biodiversity monitoring should be applied to ensure valid and comparable datasets. New programmes must be reliable and standardised, modular and applicable to different situations, and should adequately reflect the current state of biodiversity at the site. Monitoring should focus on the minimum number of indicators to provide reliable data, or else the programme will risk becoming overwhelmed with too many features and too few resources to monitor them. Today, these efforts may be supported by the implementation of novel technologies and approaches that provide significant transformational impact on how biodiversity monitoring can be achieved. Managers must evaluate the ever-growing list of tools that are now available, find ways to incorporate historical data with current and future data, and implement realistic procedures to document the state of biodiversity at their sites.

This publication introduces managers to this new paradigm of monitoring, providing guidance on how to most effectively utilise the available resources. A four-phase process is advised, including the preparatory phase, the conceptual phase, the implementation phase and finally the re-evaluation phase. Whilst the guideline focuses primarily on the six main questions of the conceptual phase and a seventh question about implementation, all four phases are equally relevant for managers. These phases operate as a cycle, where reevaluation provides the basis for renewal of the framework. The output of the first phase – where site characteristics, monitoring requirements and information gaps are identified – is to develop a statement of purpose including a priority list of monitoring targets. Next, a realistic scope for a biodiversity monitoring programme is determined through a series of six questions. Then, the actual monitoring processes are performed, first through a series of test runs and pre-runs, and then in the field. Finally, potential outcomes of the reevaluation phase are to maintain the programme as originally described, to improve the programme through modernising the techniques or other modifications, or to dissolve the programme.

This publication further provides a review of traditional approaches compared to state-of-the-art technologies that may contribute to collection and management of biodiversity data. Requirements for biodiversity monitoring in protected areas and OECMs have undergone significant changes in recent years. Current innovations make it possible for managers to keep up with these requirements. To carry out monitoring on the scale required today and into the future, a combination of high-tech solutions and traditional techniques needs to be put into practice. Integration of suitable methods and technologies will support improved management effectiveness.

Global biodiversity is facing enormous pressure, but there is good reason for optimism. Networks of protected areas and OECMs are being supported by cohesive national and international policy decisions including the current Kunming-Montreal Global Biodiversity Framework. The biodiversity monitoring framework presented in this publication puts managers in a strong position to face today's challenges of conserving biodiversity in a standardised way. Successful implementation of the framework will make resource use more efficient, prevent data loss and lead to enhanced protected area management effectiveness.

References

Abrams, J., Axtner, J., Bhagwat, T., Mohamed, A., Nguyen, A., Niedballa, J., . . . Wilting, A. (2018). Studying terrestrial mammals in tropical rainforests: A user guide for camera-trapping and environmental DNA. Berlin, Germany: Leibniz-IZW.

Adams, V. M., Barnes, M. & Pressey, R. L. (2019). Shortfalls in conservation evidence: Moving from ecological effects of interventions to policy evaluation. *One Earth*, 1(1), 62–75. https://doi.org/10.1016/j.oneear.2019.08.017

Adams-Huet, B. & Ahn, C. (2009). Bridging clinical investigators and statisticians: Writing the statistical methodology for a research proposal. *Journal of Investigative Medicine*, *57*(8), 818–824. https://doi.org/10.2310/JIM.0b013e3181c2996c

Alzubi, J., Nayyar, A. & Kumar, A. (2018). Machine learning from theory to algorithms: An overview. *Journal of Physics: Conference Series*, 1142, 12012. https://doi.org/10.1088/1742-6596/1142/1/012012

Arts, K., van der Wal, R. & Adams, W. M. (2015). Digital technology and the conservation of nature. *Ambio, 44 Suppl 4*, 661–673. https://doi.org/10.1007/s13280-015-0705-1

Aylagas, E., Borja, Á., Irigoien, X. & Rodríguez-Ezpeleta, N. (2016). Benchmarking DNA metabarcoding for biodiversity-based monitoring and assessment. *Frontiers in Marine Science*, *3*, 96. https://doi.org/10.3389/fmars.2016.00096

Bailey, J. J., Boyd, D. S., Hjort, J., Lavers, C. P. & Field, R. (2017). Modelling native and alien vascular plant species richness: At which scales is geodiversity most relevant? *Global Ecology and Biogeography*, 26(7), 763–776. https://doi.org/10.1111/geb.12574

Balvanera, P., Brauman, K. A., Cord, A. F., Drakou, E. G., Geijzendorffer, I. R., Karp, D. S., . . . Schröter, M. (2022). Essential ecosystem service variables for monitoring progress towards sustainability. *Current Opinion in Environmental Sustainability, 54*, 101152. https://doi.org/10.1016/j.cosust.2022.101152

Bamford, A. J., Monadjem, A. & Hardy, I. C. W. (2009). An effect of vegetation structure on carcass exploitation by vultures in an African savanna. Ostrich, 80(3), 135–137. https://doi.org/10.2989/OSTRICH.2009.80.3.2.965

Bäuerle, H. & Nothdurft, A. (2011). Spatial modeling of habitat trees based on line transect sampling and point pattern reconstruction. *Canadian Journal of Forest Research*, *41*(4), 715–727. https://doi.org/10.1139/x11-004

Bayfield, N. G. (1997). Approaches to monitoring for nature conservation in Scotland. In *Monitoring for Nature Conservation: Vol. 22* (pp. 23–27). Vienna: Umweltbundesamt. https://www.zobodat.at/pdf/UBA_CP-022_0001-0079.pdf

Beaudrot, L., Ahumada, J. A., O'Brien, T., Alvarez-Loayza, P., Boekee, K., Campos-Arceiz, A., . . . Andelman, S. J. (2016). Standardized assessment of biodiversity trends in tropical forest protected areas: The end is not in sight. *PLoS Biology*, *14*(1), e1002357. https://doi.org/10.1371/journal.pbio.1002357

Berger, V., Kirchmeir, H. & Hirschmugl, M. (2021). Application-oriented approach to monitoring the dynamics of avalanche tracks using conventional forest inventory parameters and Lidar-based change detection. In *Proceedings of the SilviLaser Conference 2021*, 353–355. TU Wien. https://doi.org/10.34726/wim.2034

Bertzky, B., Shi, Y., Hughes, A., Engels, B., Ali, M. K. & Badman, T. (2013). Terrestrial biodiversity and the World Heritage List: Identifying broad gaps and potential candidate sites for inclusion in the Natural World Heritage network. Gland, Switzerland and Cambridge, UK: IUCN and UNEP-WCMC.

Bialowolski, P., Rakotobe, D., Marelli, A., Roggeri, P. & Paolini, C. (2022). Use of the IMET tool in the evaluation of protected area management effectiveness in Central Africa. *Journal of Environmental Management*, 326(Pt A), 116680. https://doi.org/10.1016/j.jenvman.2022.116680 Biggs, J., Ewald, N., Valentini, A., Gaboriaud, C., Dejean, T., Griffiths, R. A., . . . Dunn, F. (2015). Using eDNA to develop a national citizen science-based monitoring programme for the great crested newt (*Triturus cristatus*). *Biological Conservation*, 183, 19–28. https://doi.org/10.1016/j.biocon.2014.11.029

Bista, I., Carvalho, G. R., Tang, M., Walsh, K., Zhou, X., Hajibabaei, M., . . . Creer, S. (2018). Performance of amplicon and shotgun sequencing for accurate biomass estimation in invertebrate community samples. *Molecular Ecology Resources*, 18(5), 1020–1034. https://doi.org/10.1111/1755-0998.12888

Borrini, G., Kothari, A. & Oviedo, G. (2004). Indigenous and local communities and protected areas: Towards equity and enhanced conservation: guidance on policy and practice for co-managed protected areas and community conserved areas. Best practice protected area guidelines series: no. 11. Gland, Switzerland and Cambridge, UK: IUCN.

Borrini-Feyerabend, G., Dudley, N., Jaeger, T., Lassen, B., Broome, N., Phillips, A. & Sandwith, T. (2013). Governance of protected areas: From understanding to action. Best practice protected area guidelines series: No. 20. Gland, Switzerland: IUCN. https:// portals.iucn.org/library/node/29138

Bråkenhielm, S. & Qinghong, L. (1995). Comparison of field methods in vegetation monitoring. *Water, Air, and Soil Pollution, 79*, 75–87. https://doi.org/10.1007/BF01100431

Braun-Blanquet, J. (1964). *Pflanzensoziologie: Grundzüge der Vegetationskunde*. Vienna: Springer Vienna. Retrieved from https://doi.org/10.1007/978-3-7091-8110-2

Brondizio, E., Diaz, S., Settele, J., Ngo, H. T., Gueze, M., Aumeeruddy-Thomas, Y., . . . Jaureguiberry, P. (2019). Chapter 1 Assessing a planet in transformation: Rationale and approach of the IPBES Global Assessment on Biodiversity and Ecosystem Services. In E. Brondizio, S. Diaz and J. Settele (Eds.), *IPBES Global Assessment Report on Biodiversity and Ecosystem Services*. Zenodo. https://www.doi.org/10.5281/zenodo.5517203

Bruce, K., Blackman, R., Bourlat, S. J., Hellström, A. M., Bakker, J., Bista, I., . . . Deiner, K. (2021). *A practical guide to DNA-based methods for biodiversity assessment*. Pensoft Publishers. https://doi.org/10.3897/ab.e68634

Buchler, E. R. (1976). A chemiluminescent tag for tracking bats and other small nocturnal animals. *Journal of Mammalogy*, *57*(1), 173–176. https://doi.org/10.2307/1379522

Buitenwerf, R., Bond, W. J., Stevens, N. & Trollope, W. S. W. (2012). Increased tree densities in South African savannas: >50 years of data suggests CO₂ as a driver. *Global Change Biology*, 18(2), 675–684. https://doi.org/10.1111/j.1365-2486.2011.02561.x

Burton, A. C., Neilson, E., Moreira, D., Ladle, A., Steenweg, R., Fisher, J. T., . . . Boutin, S. (2015). Review: Wildlife camera trapping: a review and recommendations for linking surveys to ecological processes. *The Journal of Applied Ecology*, *52*(3), 675–685. https://doi.org/10.1111/1365-2664.12432

Butler, S. J., Freckleton, R. P., Renwick, A. R. & Norris, K. (2012). An objective, niche-based approach to indicator species selection. *Methods in Ecology and Evolution*, 3(2), 317–326. https://doi.org/10.1111/j.2041-210X.2011.00173.x

Cain, P. W. & Cross, M. D. (2018). An open-source hardware GPS data logger for wildlife radio-telemetry studies: A case study using Eastern box turtles. *HardwareX*, *3*, 82–90. https://doi.org/10.1016/j.ohx.2018.02.002

Canonico, G., Buttigieg, P. L., Montes, E., Muller-Karger, F. E., Stepien, C., Wright, D., . . . Murton, B. (2019). Global observational needs and resources for marine biodiversity. *Frontiers in Marine Science*, 6. https://doi.org/10.3389/fmars.2019.00367 Carey, C. C., Ward, N. K., Farrell, K. J., Lofton, M. E., Krinos, A. I., McClure, R. P., . . . Arzberger, P. (2019). Enhancing collaboration between ecologists and computer scientists: Lessons learned and recommendations forward. *Ecosphere*, *10*(5). https://doi.org/10.1002/ecs2.2753

Carilla, J., Grau, H. R., Paolini, L. & Mariano, M. (2013). Lake fluctuations, plant productivity, and long-term variability in highelevation tropical Andean ecosystems. *Arctic, Antarctic, and Alpine Research, 45*(2), 179–189. https://doi.org/10.1657/1938-4246-45.2.179

Carroll, S. R., Garba, I., Figueroa-Rodríguez, O. L., Holbrook, J., Lovett, R., Materechera, S., . . . Hudson, M. (2020). The CARE Principles for Indigenous Data Governance. *Data Science Journal, 19.* https://doi.org/10.5334/dsj-2020-043

Carvalho, S., Aylagas, E., Villalobos, R., Kattan, Y., Berumen, M. & Pearman, J. K. (2019). Beyond the visual: Using metabarcoding to characterize the hidden reef cryptobiome. *Proceedings of the Royal Society B: Biological Sciences, 286*(1896), 20182697. https://doi.org/10.1098/rspb.2018.2697

Cato, D., McCauley, R., Rogers, T. & Noad, M. (2006). Passive acoustics for monitoring marine animals – progress and challenges. In *Proceedings of ACOUSTICS 2006*, 453–460. Christchurch, New Zealand. Retrieved from https://t.ly/X-KDE

Caughlan, L. & Oakley, K. L. (2001). Cost considerations for longterm ecological monitoring. *Ecological Indicators*, 1(2), 123–134. https://doi.org/10.1016/S1470-160X(01)00015-2

Convention on Biological Diversity (CBD) (1992). The Convention on Biological Diversity: CBD. Retrieved from https://www.cbd.int/ doc/legal/cbd-en.pdf

CBD (2020). Aichi biodiversity targets. Retrieved from https://www.cbd.int/sp/targets/

CBD (2022). Decision adopted by the Conference of the Parties to the Convention on Biological Diversity 15/4. Kunming-Montreal Global Biodiversity Framework. Retrieved from https://www.cbd.int/doc/decisions/cop-15/cop-15-dec-04-en.pdf

CBD (2023). Kunming-Montreal Global Biodiversity Framework: Target 3. 30 per cent of areas are effectively conserved. Retrieved from https://www.cbd.int/gbf/targets/3/

Christin, S., Hervet, É. & Lecomte, N. (2019). Applications for deep learning in ecology. *Methods in Ecology and Evolution*, *10*(10), 1632–1644. https://doi.org/10.1111/2041-210X.13256

Conservation Measures Partnership (CMP) (2020). Conservation Standards 4.0: Open standards for the practice of conservation. Retrieved from Conservation Measures Partnership website: https://t.ly/EAcXp

Colavizza, G., Blanke, T., Jeurgens, C. & Noordegraaf, J. (2022). Archives and Al: An overview of current debates and future perspectives. *Journal on Computing and Cultural Heritage, 15*(1), 1–15. https://doi.org/10.1145/3479010

Cole, A. & McCallion, A. (2018). Special issue on information technology and systems: Towards an understanding of barriers to the Māori cultural adoption and use of information technology and systems in the expression of kaitiakitanga (Korero Māori No. 14). Tauranga, New Zealand.

Collett, R. A. & Fisher, D. O. (2017). Time-lapse camera trapping as an alternative to pitfall trapping for estimating activity of leaf litter arthropods. *Ecology and Evolution*, 7(18), 7527–7533. https://doi.org/10.1002/ece3.3275

Cooperrider, A. Y. (1986). Evaluation and interpretation. In A. Y. Cooperrider, R. J. Boyd & H. R. Stuart (Eds.), *Inventory and monitoring of wildlife habitat* (pp. 777–783). Denver, CO: Bureau of Land Management Service Center.

Cui, S., Ling, P., Zhu, H. & Keener, H. M. (2018). Plant pest detection using an artificial nose system: A review. *Sensors, 18*(2). https://doi.org/10.3390/s18020378 Culina, A., Baglioni, M., Crowther, T. W., Visser, M. E., Woutersen-Windhouwer, S. & Manghi, P. (2018). Navigating the unfolding open data landscape in ecology and evolution. *Nature Ecology & Evolution*, 2(3), 420–426. https://doi.org/10.1038/s41559-017-0458-2

Dallas, H. F. (2021). Rapid bioassessment protocols using aquatic macroinvertebrates in Africa – considerations for regional adaptation of existing biotic indices. *Frontiers in Water, 3*. https://doi.org/10.3389/frwa.2021.628227

Dalton, D. T., Pascher, K., Berger, V., Steinbauer, K. & Jungmeier, M. (2022). Novel technologies and their application for protected area management: A supporting approach in biodiversity monitoring. In M. Nazip Suratman (Ed.), *Protected Area Management – Recent Advances*. IntechOpen. https://doi.org/10.5772/intechopen.99889

Danielsen, F., Burgess, N. D., Balmford, A., Donald, P. F., Funder, M., Jones, J. P. G., . . . Yonten, D. (2009). Local participation in natural resource monitoring: A characterization of approaches. *Conservation Biology*, 23(1), 31–42. https://doi.org/10.1111/j.1523-1739.2008.01063.x

Danovaro, R., Carugati, L., Berzano, M., Cahill, A. E., Carvalho, S., Chenuil, A., . . . Borja, A. (2016). Implementing and innovating marine monitoring approaches for assessing marine environmental status. *Frontiers in Marine Science*, 3. https://doi.org/10.3389/fmars.2016.00213

Darras, K., Batáry, P., Furnas, B. J., Grass, I., Mulyani, Y. A. & Tscharntke, T. (2019). Autonomous sound recording outperforms human observation for sampling birds: A systematic map and user guide. *Ecological Applications*, 29(6), e01954. https://doi.org/10.1002/eap.1954

Darras, K. F. A., Deppe, F., Fabian, Y., Kartono, A. P., Angulo, A., Kolbrek, B., . . . Prawiradilaga, D. M. (2020). High microphone signal-to-noise ratio enhances acoustic sampling of wildlife. *PeerJ*, 8, e9955. https://doi.org/10.7717/peerj.9955

de Bello, F., Valencia, E., Ward, D. & Hallett, L. (2020). Why we still need permanent plots for vegetation science. Journal of Vegetation Science, 31(5), 679–685. https://doi.org/10.1111/jvs.12928

de Lima, R. A. F., Phillips, O. L., Duque, A., Tello, J. S., Davies, S. J., de Oliveira, A. A., . . . Vásquez, R. (2022). Making forest data fair and open. *Nature Ecology & Evolution*, 6(6), 656–658. https://doi.org/10.1038/s41559-022-01738-7

Deiner, K. & Altermatt, F. (2014). Transport distance of invertebrate environmental DNA in a natural river. *PLoS ONE, 9*(2), e88786. https://doi.org/10.1371/journal.pone.0088786

Dudley, N. (2008). *Guidelines for applying protected area management categories*. Gland, Switzerland: IUCN.

Dyck, T., LeClair, J. & Bockstael, E. (2022). Literature review: Braiding Indigenous knowledge and science in environmental assessments and reporting (Project no. PCABIS-22). Integral Ecology Group.

Egloff, W., Agosti, D., Patterson, D., Hoffmann, A., Mietchen, D., Kishor, P. & Penev, L. (2016). Data policy recommendations for biodiversity data. EU BON Project Report. *Research Ideas and Outcomes, 2.* https://doi.org/10.3897/rio.2.e8458

Elzinga, C. L., Salzer, D. W. & Willoughby, J. W. (2019). *Measuring & monitoring plant populations* (Bureau of Land Management Papers No. BLM/RS/ST-98/005+1730). Denver, CO. Retrieved from US Bureau of Land Management website: https://t.ly/BvHA_

Erickson, R. A., Merkes, C. M. & Mize, E. L. (2019). Sampling designs for landscape-level eDNA monitoring programs. *Integrated Environmental Assessment and Management*, 15(5), 760–771. https://doi.org/10.1002/ieam.4155

Feinsinger, P. (2001). *Designing field studies for biodiversity conservation*. Washington, D.C.: Island Press.

Ferrari, A., Bacco, M., Gaber, K., Jedlitschka, A., Hess, S., Kaipainen, J., . . . Brunori, G. (2022). Drivers, barriers and impacts of digitalisation in rural areas from the viewpoint of experts. *Information and Software Technology, 145*, 106816. https://doi.org/10.1016/j.infsof.2021.106816 Ferraro, P. J. & Miranda, J. J. (2014). The performance of nonexperimental designs in the evaluation of environmental programs: A design-replication study using a large-scale randomized experiment as a benchmark. *Journal of Economic Behavior & Organization, 107*, 344–365. https://doi.org/10.1016/j.jebo.2014.03.008

- Ferraro, P. J. & Miranda, J. J. (2017). Panel data designs and estimators as substitutes for randomized controlled trials in the evaluation of public programs. *Journal of the Association of Environmental and Resource Economists, 4*(1), 281–317. https://doi.org/10.1086/689868
- Ferraro, P. J. & Price, M. K. (2013). Using nonpecuniary strategies to influence behavior: Evidence from a large-scale field experiment. *Review of Economics and Statistics*, 95(1), 64–73. https://doi.org/10.1162/REST_a_00344
- Foucher, A., Evrard, O., Ficetola, G. F., Gielly, L., Poulain, J., Giguet-Covex, C., . . . Poulenard, J. (2020). Persistence of environmental DNA in cultivated soils: Implication of this memory effect for reconstructing the dynamics of land use and cover changes. *Scientific Reports*, 10(1), 10502. https://doi.org/10.1038/s41598-020-67452-1
- Fremout, T., Gutiérrez-Miranda, C. E., Briers, S., Marcelo-Peña, J. L., Cueva-Ortiz, E., Linares-Palomino, R., . . . Thomas, E. (2021). The value of local ecological knowledge to guide tree species selection in tropical dry forest restoration. *Restoration Ecology*, 29(4). https://doi.org/10.1111/rec.13347
- Fuller, M. R. & Fuller, T. K. (2012). Radio-telemetry equipment and applications for carnivores. In L. Boitani & R. A. Powell (Eds.), *Carnivore ecology and conservation: A handbook of techniques* (pp. 152–168). Oxford University Press. https://doi.org/10.1093/acprof:oso/9780199558520.003.0007
- Futschik, A., Winkler, M., Steinbauer, K., Lamprecht, A., Rumpf, S. B., Barančok, P., . . . Pauli, H. (2020). Disentangling observer error and climate change effects in long-term monitoring of alpine plant species composition and cover. *Journal of Vegetation Science*, 31(1), 14–25. https://doi.org/10.1111/jvs.12822
- Ganzevoort, W., van den Born, R. J. G., Halffman, W. & Turnhout, S. (2017). Sharing biodiversity data: Citizen scientists' concerns and motivations. *Biodiversity and Conservation, 26*(12), 2821–2837. https://doi.org/10.1007/s10531-017-1391-z
- Government of South Georgia & the South Sandwich Islands (2016). Biodiversity action plan for South Georgia & the South Sandwich Islands 2016–2020. Stanley, Falkland Islands. Retrieved from Government of South Georgia & the South Sandwich Islands website:https://t.ly/2Nv8b
- Government of South Georgia & the South Sandwich Islands (2021). South Georgia and South Sandwich Islands marine protected area research and monitoring plan. Stanley, Falkland Islands. Retrieved from Government of South Georgia & the South Sandwich Islands website: https://t.ly/4uN9l
- Grabherr, G., Koch, G., Kirchmeir, H. & Reiter, K. (1998). *Hemerobie* österreichischer Waldökosysteme. Veröffentlichungen des Österreichischen MaB-Programms: Bd. 17. Innsbruck: Universitätsverlag Wagner. https://www.cipra.org/de/publikationen/2288
- Grant, T. A., Madden, E. M., Murphy, R. K., Smith, K. A. & Nenneman, M. P. (2004). Monitoring native prairie vegetation: The belt transect method. *Ecological Restoration*, 22(2), 106–111. https://doi.org/10.3368/er.22.2.106
- Gregory, R. D., Gibbons, D. W. & Donald, P. F. (2004). Bird census and survey techniques. In W. J. Sutherland, I. Newton & R. Green (Eds.), *Bird Ecology and Conservation* (pp. 17–56). Oxford University Press.

https://doi.org/10.1093/acprof:oso/9780198520863.003.0002

Groffman, P. M., Baron, J. S., Blett, T., Gold, A. J., Goodman, I., Gunderson, L. H., . . . Wiens, J. (2006). Ecological thresholds: The key to successful environmental management or an important concept with no practical application? *Ecosystems*, 9(1), 1–13. https://doi.org/10.1007/s10021-003-0142-z

- Grootaert, P., Pollet, M., Dekoninck, W. & van Achterbert, C. (2010). Sampling insects: General techniques, strategies and remarks. In J. Eymann, J. Degreef, C. Häuser, J. Monje, Y. Samyn & D. VandenSpiegel (Eds.), *Manual on field recording techniques and protocols for all taxa biodiversity inventories and monitoring* (pp. 377–399). ABC Taxa.
- Hagler, J. R. (2019). Super mark it! A review of the protein immunomarking technique. *Annals of the Entomological Society of America, 112*(3), 200-210. https://doi.org/10.1093/aesa/say039
- Halloy, S., Ibáñez, M. & Yager, K. (2011). Puntos y áreas flexibles (PAF) para inventarios rápidos del estado de biodiversidad. *Ecología en Bolivia, 46*(1), 46–56.
- Hannah, L., Roehrdanz, P. R., Marquet, P. A., Enquist, B. J., Midgley, G., Foden, W., . . . Svenning, J.-C. (2020). 30% land conservation and climate action reduces tropical extinction risk by more than 50%. *Ecography*, 43(7), 943–953. https://doi.org/10.1111/ecog.05166
- Harris, M., Woodcock, P., Wright, E., Britton, A., Matear, L., Smith, A., . . . Cheffings, C. (2021). *Biodiversity indicators review – international climate finance evidence project*. Peterborough. Retrieved from JNCC website: https://rb.gy/83q68a
- He, K., Zhang, X., Ren, S. & Sun, J. (2016). Deep residual learning for image recognition. In 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR) (pp. 770–778). IEEE. https://doi.org/10.1109/CVPR.2016.90
- Hill, A. P., Prince, P., Piña Covarrubias, E., Doncaster, C. P., Snaddon, J. L. & Rogers, A. (2018). AudioMoth: Evaluation of a smart open acoustic device for monitoring biodiversity and the environment. *Methods in Ecology and Evolution*, 9(5), 1199–1211. https://doi.org/10.1111/2041-210X.12955
- Hobbs, M. T. & Brehme, C. S. (2017). An improved camera trap for amphibians, reptiles, small mammals, and large invertebrates. *PLoS ONE, 12*(10), e0185026. https://doi.org/10.1371/journal.pone.0185026
- Hochachka, W. M., Caruana, R., Fink, D., Munson, A. R., Riedewald, M., Sorokina, D. & Kelling, S. (2007). Data-mining discovery of pattern and process in ecological systems. *Journal of Wildlife Management*, *71*(7), 2427–2437. https://doi.org/10.2193/2006-503
- Hockings, M., James, R., Stolton, S., Dudley, N., Mathur, V., Makombo, J., . . . Parrish, J. (2008). Enhancing our Heritage toolkit: Assessing management effectiveness of natural World Heritage sites. Paris, France: UNESCO. Retrieved from https://rb.gy/h32gqx
- Hockings, M., Leverington, F. & Cook, C. (2015). Protected area management effectiveness. In G. L. Worboys, M. Lockwood, A. Kothari, S. Feary & I. Pulsford (Eds.), *Protected Area Governance and Management* (pp. 889–928). Canberra: ANU Press. https:// doi.org/10.22459/PAGM.04.2015
- Hockings, M., Stolton, S., Leverington, F., Dudley, N. & Courrau, J. (2006). Evaluating effectiveness: A framework for assessing management effectiveness of protected areas (2nd ed.). Gland, Switzerland: IUCN. https://doi.org/10.2305/IUCN.CH.2006.PAG.14.en
- Hodgkinson, S. & Young, D. (2016). *The Internet of Things for Pendjari National Park: A phased innovative technology investment strategy to improve management effectiveness.* Second report of IUCN Mission to Pendjari National Park.
- Hoffmann, A., Decher, J., Rovero, F., Schaer, J., Voigt, C. & Wibbelt,
 G. (2010). Field methods and techniques for monitoring mammals.
 In J. Eymann, J. Degreef, C. Häuser, J. Monje, Y. Samyn & D.
 VandenSpiegel (Eds.), *Manual on field recording techniques and protocols for all Taxa Biodiversity Inventories and Monitoring* (pp. 482–529). ABC Taxa.
- Huettmann, F. (2009). The global need for, and appreciation of, high-quality metadata in biodiversity database work. In E. Spehn & C. Körner (Eds.), *Data Mining for Global Trends in Mountain Biodiversity* (pp. 25–28). Boca Raton: CRC Press. https://doi.org/10.1201/9781420083705.ch4

- Intergovernmental Panel on Climate Change (IPCC) (2021). *Climate change 2021: the physical science basis*. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK, New York, NY: IPCC.
- IPCC (2022). Climate change 2022: impacts, adaptation and vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK, New York, NY: IPCC.
- Irvine, K. M. & Rodhouse, T. J. (2010). Power analysis for trend in ordinal cover classes: Implications for long-term vegetation monitoring. *Journal of Vegetation Science*, *21*(6), 1152–1161. https://doi.org/10.1111/j.1654-1103.2010.01214.x
- International Union for Conservation of Nature (IUCN) (2016). *A global standard for the identification of key biodiversity areas, version 1.0.* First edition. Gland, Switzerland: IUCN. https://doi.org/10.2305/iucn.ch.2020.08.en
- IUCN (2020). Guidance for using the IUCN Global Standard for Nature-based Solutions. A user-friendly framework for the verification, design and scaling up of Nature-based Solutions. First edition. Gland, Switzerland: IUCN. https://doi.org/10.2305/IUCN.CH.2020.09.en
- IUCN (2022a). Application of Indigenous & Local Knowledge (ILK) in IUCN Red List assessments: White paper. Version 1. Retrieved from IUCN SSC Red List Committee; IUCN CEESP-SSC Sustainable Use & Livelihoods Specialist Group Steering Committee website: https://www.iucnredlist.org/resources/ilk
- IUCN (2022b). IUCN glossary of definitions. https://shorturl.at/hJRSV
- IUCN (2022c). The IUCN Red List of Threatened Species. Retrieved from https://www.iucnredlist.org/
- IUCN Standards and Petitions Committee (2022). *Guidelines for using the IUCN Red List categories and criteria*. Retrieved from https://www.iucnredlist.org/documents/RedListGuidelines.pdf
- Joly, A., Vrochidis, S., Karatzas, K., Karppinen, A. & Bonnet, P. (2018). Multimedia tools and applications for environmental & biodiversity informatics. Cham, Switzerland: Springer International Publishing. https://doi.org/10.1007/978-3-319-76445-0
- Jonas, H. D., MacKinnon, K., Marnewick, D. & Wood, P. (2023). Sitelevel tool for identifying other effective area-based conservation measures (OECMs): first edition. Gland, Switzerland: IUCN. https://doi.org/10.2305/WZJH1425
- Jones, K. B. (1986). Data types. In A. Y. Cooperrider, R. J. Boyd & H. R. Stuart (Eds.), *Inventory and monitoring of wildlife habitat* (pp. 11–28). Denver, CO: Bureau of Land Management Service Center.
- Joppa, L. N. (2015). Technology for nature conservation: An industry perspective. *Ambio*, 44(4), 522–526. https://doi.org/10.1007/s13280-015-0702-4
- Jungmeier, M., Köstl, T., Lange, S. & Bliem, M. (2013). The art of omission: BRIM^{Nockberge} – designing a biosphere reserve integrated monitoring for the Carinthian part of the Biosphere Reserve Salzburger Lungau & Kärntner Nockberge. eco.mont, 5(2), 15–22. https://doi.org/10.1553/ecomont-5-2s15
- Karadeniz, N. & Yenilmez Arpa, N. (2022a). Guidelines for assessing the management effectiveness of protected areas. Ankara, Turkey: FAO, MAF. https://doi.org/10.4060/cb8349en
- Karadeniz, N. & Yenilmez Arpa, N. (2022b). Guidelines for engaging stakeholders in managing protected areas. Ankara, Turkey: FAO. https://doi.org/10.4060/cb8347en
- Karakaya, D., Ulucan, O. & Turkan, M. (2020). Electronic nose and its applications: A survey. *International Journal of Automation and Computing*, *17*(2), 179–209. https://doi.org/10.1007/s11633-019-1212-9
- Kelemen, E., Subramanian, S. M., de Vos, A., Amaruzaman, S., Porter-Bolland, L., Islar, M., . . . Zólyomi, Á. (2023). Signposts on the road toward transformative governance: How a stronger focus on diverse values can enhance environmental policies. *Current Opinion in Environmental Sustainability, 64*, 101351. https://doi.org/10.1016/j.cosust.2023.101351

- Kelleher, G. (1999). Guidelines for marine protected areas. Best practice protected area guidelines series: no. 3. Gland, Switzerland and Cambridge, UK: IUCN.
- Kissling, W. D., Ahumada, J. A., Bowser, A., Fernandez, M., Fernández, N., García, E. A., . . . Hardisty, A. R. (2018). Building essential biodiversity variables (EBVs) of species distribution and abundance at a global scale. *Biological Reviews of the Cambridge Philosophical Society*, 93(1), 600–625. https://doi.org/10.1111/brv.12359
- Kissling, W. D., Pattemore, D. E. & Hagen, M. (2014). Challenges and prospects in the telemetry of insects. *Biological Reviews of the Cambridge Philosophical Society*, 89(3), 511–530. https://doi.org/10.1111/brv.12065
- Kucera, T. E. & Barrett, R. H. (2011). A history of camera trapping. In A. F. O'Connell, J. D. Nichols & K. U. Karanth (Eds.), *Camera traps in animal ecology* (pp. 9–26). Tokyo: Springer Japan. https://doi.org/10.1007/978-4-431-99495-4_2
- Lacher, I. (2018). Systematic conservation planning in the Anthropocene. In *Encyclopedia of the Anthropocene* (pp. 461–469). Elsevier. https://doi.org/10.1016/B978-0-12-809665-9.10290-3
- Lahoz-Monfort, J. J. & Magrath, M. J. L. (2021). A comprehensive overview of technologies for species and habitat monitoring and conservation. *BioScience*, *71*(10), 1038–1062. https://doi.org/10.1093/biosci/biab073
- Lawniczak, M. K. N., Durbin, R., Flicek, P., Lindblad-Toh, K., Wei, X., Archibald, J. M., . . . Richards, S. (2022). Standards recommendations for the Earth BioGenome Project. *Proceedings* of the National Academy of Sciences of the United States of America, 119(4). https://doi.org/10.1073/pnas.2115639118
- Lenton, T. M., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W. & Schellnhuber, H. J. (2019). Climate tipping points – too risky to bet against. *Nature*, 575(7784), 592–595. https://doi.org/10.1038/d41586-019-03595-0
- Liang, J. & Gamarra, J. G. P. (2020). The importance of sharing global forest data in a world of crises. *Scientific Data*, 7(1), 424. https://doi.org/10.1038/s41597-020-00766-x
- Lim, M. C. W., Seimon, A., Nightingale, B., Xu, C. C. Y., Halloy, S. R. P., Solon, A. J., . . . Seimon, T. A. (2022). Estimating biodiversity across the tree of life on Mount Everest's southern flank with environmental DNA. *iScience*, 25(9), 104848. https://doi.org/10.1016/j.isci.2022.104848
- Lindenmayer, D., Pierson, J., Barton, P., Beger, M., Branquinho, C., Calhoun, A., . . . Westgate, M. (2015). A new framework for selecting environmental surrogates. *The Science of the Total Environment*, 538, 1029–1038. https://doi.org/10.1016/j.scitotenv.2015.08.056
- Lindenmayer, D. B., Gibbons, P., Bourke, M. A. X., Burgman, M., Dickman, C. R., Ferrier, S., . . . Zerger, A. (2012a). Improving biodiversity monitoring. *Austral Ecology*, *37*(3), 285–294. https://doi.org/10.1111/j.1442-9993.2011.02314.x
- Lindenmayer, D. B., Zammit, C., Attwood, S. J., Burns, E., Shepherd, C. L., Kay, G. & Wood, J. (2012b). A novel and cost-effective monitoring approach for outcomes in an Australian biodiversity conservation incentive program. *PLoS ONE*, 7(12), e50872. https://doi.org/10.1371/journal.pone.0050872
- Lockwood, M. (2010). Good governance for terrestrial protected areas: A framework, principles and performance outcomes. *Journal of Environmental Management*, *91*(3), 754–766. https://doi.org/10.1016/j.jenvman.2009.10.005
- Lunghi, E., Corti, C., Manenti, R. & Ficetola, G. F. (2019). Consider species specialism when publishing datasets. *Nature Ecology & Evolution*, 3(3), 319. https://doi.org/10.1038/s41559-019-0803-8
- Mace, G. M. & Baillie, J. E. M. (2007). The 2010 biodiversity indicators: Challenges for science and policy. *Conservation Biology, 21*(6), 1406–1413. https://doi.org/10.1111/j.1523-1739.2007.00830.x
- Magurran, A. E. (2004). *Measuring biological diversity*. Oxford, UK: Blackwell.

- Malinka, C. E., Gillespie, D. M., Macaulay, J. D., Joy, R. & Sparling, C. E. (2018). First in situ passive acoustic monitoring for marine mammals during operation of a tidal turbine in Ramsey Sound, Wales. *Marine Ecology Progress Series*, 590, 247–266. https://doi.org/10.3354/meps12467
- Martínez-López, J., Bertzky, B., Willcock, S., Robuchon, M., Almagro, M., Delli, G. & Dubois, G. (2021). Remote sensing methods for the biophysical characterization of protected areas globally: Challenges and opportunities. *ISPRS International Journal of Geo-Information*, 10(6), 384. https://doi.org/10.3390/ijgi10060384
- Mascia, M. B. & Mills, M. (2018). When conservation goes viral: The diffusion of innovative biodiversity conservation policies and practices. *Conservation Letters*, *11*(3), e12442. https://doi.org/10.1111/conl.12442
- Matley, J. K., Klinard, N. V., Barbosa Martins, A. P., Aarestrup, K., Aspillaga, E., Cooke, S. J., . . . Fisk, A. T. (2022). Global trends in aquatic animal tracking with acoustic telemetry. *Trends in Ecology* & *Evolution*, 37(1), 79–94. https://doi.org/10.1016/j.tree.2021.09.001
- Mazumdar, S., Seybold, D., Kritikos, K. & Verginadis, Y. (2019). A survey on data storage and placement methodologies for Cloud-Big Data ecosystem. *Journal of Big Data*, 6(1). https://doi.org/10.1186/s40537-019-0178-3
- Minamoto, T., Miya, M., Sado, T., Seino, S., Doi, H., Kondoh, M., . . . Uchii, K. (2021). An illustrated manual for environmental DNA research: Water sampling guidelines and experimental protocols. *Environmental DNA*, 3(1), 8–13. https://doi.org/10.1002/edn3.121
- Mitchell, R. B. (2003). International environmental agreements: A survey of their features, formation, and effects. *Annual Review of Environment and Resources*, 28(1), 429–461. https://doi.org/10.1146/annurev.energy.28.050302.105603
- Muller-Karger, F. E., Miloslavich, P., Bax, N. J., Simmons, S., Costello, M. J., Sousa Pinto, I., . . . Geller, G. (2018). Advancing marine biological observations and data requirements of the complementary essential ocean variables (EOVs) and essential biodiversity variables (EBVs) frameworks. *Frontiers in Marine Science*, 5. https://doi.org/10.3389/fmars.2018.00211
- Navarro, L. M., Fernández, N., Guerra, C., Guralnick, R., Kissling, W. D., Londoño, M. C., . . . Pereira, H. M. (2017). Monitoring biodiversity change through effective global coordination. *Current Opinion in Environmental Sustainability*, 29, 158–169. https://doi.org/10.1016/j.cosust.2018.02.005
- Newey, S., Davidson, P., Nazir, S., Fairhurst, G., Verdicchio, F., Irvine, R. J. & van der Wal, R. (2015). Limitations of recreational camera traps for wildlife management and conservation research: A practitioner's perspective. *Ambio, 44 Suppl 4*, 624–635. https://doi.org/10.1007/s13280-015-0713-1
- Nicholson, E., Watermeyer, K. E., Rowland, J. A., Sato, C. F., Stevenson, S. L., Andrade, A., . . . Watson, J. E. M. (2021). Scientific foundations for an ecosystem goal, milestones and indicators for the post-2020 global biodiversity framework. *Nature Ecology & Evolution*, 5(10), 1338–1349. https://doi.org/10.1038/s41559-021-01538-5
- Niemeijer, D. & de Groot, R. S. (2008). A conceptual framework for selecting environmental indicator sets. *Ecological Indicators*, 8(1), 14–25. https://doi.org/10.1016/j.ecolind.2006.11.012
- Nitzu, E., Nae, A., Giurginca, A. & Popa, I. (2010). Invertebrate communities from the mesovoid shallow substratum of the Carpatho-Euxinic area: Eco-faunistic and zoogeographic analysis. *Travaux de l'Institute de Spéologie "Émile Racovitza", 49*, 41–79
- Norouzzadeh, M. S., Nguyen, A., Kosmala, M., Swanson, A., Palmer, M. S., Packer, C. & Clune, J. (2018). Automatically identifying, counting, and describing wild animals in camera-trap images with deep learning. *Proceedings of the National Academy of Sciences* of the United States of America, 115(25), E5716-E5725. https://doi.org/10.1073/pnas.1719367115
- Nowak, M. M., Dziób, K., Ludwisiak, Ł. & Chmiel, J. (2020). Mobile GIS applications for environmental field surveys: A state of the art. *Global Ecology and Conservation, 23*, e01089. https://doi.org/10.1016/j.gecco.2020.e01089

- O'Brien, T. G., Baillie, J. E. M., Krueger, L. & Cuke, M. (2010). The Wildlife Picture Index: monitoring top trophic levels. *Animal Conservation*, *13*(4), 335–343. https://doi.org/10.1111/j.1469-1795.2010.00357.x
- Obura, D. O., Aeby, G., Amornthammarong, N., Appeltans, W., Bax, N., Bishop, J., . . . Wongbusarakum, S. (2019). Coral reef monitoring, reef assessment technologies, and ecosystem-based management. *Frontiers in Marine Science*, 6. https://doi.org/10.3389/fmars.2019.00580
- Organisation for Economic Co-operation and Development (OECD) (2015). *OECD environmental performance reviews: Brazil 2015*. Paris: OECD Publishing. https://doi.org/10.1787/9789264240094-en
- Oliver, T. H., Heard, M. S., Isaac, N. J. B., Roy, D. B., Procter, D., Eigenbrod, F., . . . Bullock, J. M. (2015). Biodiversity and resilience of ecosystem functions. *Trends in Ecology & Evolution*, 30(11), 673–684. https://doi.org/10.1016/j.tree.2015.08.009
- Oliverio, A. M., Gan, H., Wickings, K. & Fierer, N. (2018). A DNA metabarcoding approach to characterize soil arthropod communities. *Soil Biology and Biochemistry, 125*, 37–43. https://doi.org/10.1016/j.soilbio.2018.06.026
- Olson, D. M. & Dinerstein, E. (1998). The Global 200: A representation approach to conserving the Earth's most biologically valuable ecoregions. *Conservation Biology*, *12*(3), 502–515. https://doi.org/10.1046/j.1523-1739.1998.012003502.x
- Osipova, E., Emslie-Smith, M., Osti, M., Murai, M., Åberg, U. & Shadie, P. (2020). *IUCN World Heritage Outlook 3: A conservation* assessment of all natural World Heritage sites. Gland, Switzerland: IUCN. https://doi.org/10.2305/IUCN.CH.2020.16.en
- Osipova, E., Shadie, P., Zwahlen, C., Osti, M., Shi, Y., Kormos, C., . . . Badman, T. (2017). *IUCN World Heritage Outlook 2: A conservation assessment of all natural World Heritage sites*. Gland, Switzerland: IUCN. https://doi.org/10.2305/IUCN.CH.2017.17.en
- Osipova, E., Wilson, L., Blaney, R., Shi, Y., Fancourt, M., Strubel, M., . . . Verschuuren, B. (2014). *IUCN World Heritage Outlook* 2014: A conservation assessment of all natural World Heritage sites. Gland, Switzerland: IUCN. https://portals.iucn.org/library/ node/44889
- O'Sullivan, B., Keady, S., Keane, E., Irwin, S. & O'Halloran, J. (2010).
 Data mining for biodiversity prediction in forests. In J. Breuker, N. Guarino, J. N. Kok, J. Liu, R. López de Mántaras, R. Mizoguchi, . . . N. Zhong (Eds.), *Frontiers in Artificial Intelligence and Applications: Vol. 215.* ECAI 2010 (pp. 289–294). Amsterdam: IOS Press. https://doi.org/10.3233/978-1-60750-606-5-289
- Packalen, P., Strunk, J., Maltamo, M. & Myllymäki, M. (2023). Circular or square plots in ALS-based forest inventories—does it matter? *Forestry*, 96(1), 49–61. https://doi.org/10.1093/forestry/cpac032
- Pannell, D. J., Marshall, G. R., Barr, N., Curtis, A., Vanclay, F. & Wilkinson, R. (2006). Understanding and promoting adoption of conservation practices by rural landholders. *Australian Journal of Experimental Agriculture, 46*(11), 1407. https://doi.org/10.1071/EA05037
- Pascher, K., Švara, V. & Jungmeier, M. (2022). Environmental DNAbased methods in biodiversity monitoring of protected areas: Application range, limitations, and needs. *Diversity*, *14*(6), 463. https://doi.org/10.3390/d14060463
- Paul, T. S. H., Kimberley, M. O. & Beets, P. N. (2019). Thinking outside the square: Evidence that plot shape and layout in forest inventories can bias estimates of stand metrics. *Methods in Ecology and Evolution*, 10(3), 381–388. https://doi.org/10.1111/2041-210X.13113
- Pauli, H., Gottfried, M., Lamprecht, A., Niessner, S., Rumpf, S., Winkler, M., . . . Grabherr, G. (2015). *The GLORIA field manual* – standard multi-summit approach, supplementary methods and extra approaches (5th edition). Vienna, Austria: GLORIA-Coordination, Austrian Academy of Sciences & University of Natural Resources and Life Sciences.

Pawlowski, J., Apothéloz-Perret-Gentil, L., Mächler, E. & Altermatt, F. (2020). Environmental DNA applications for biomonitoring and bioassessment in aquatic ecosystems. Bern: Federal Office for the Environment. https://doi.org/10.5167/UZH-187800

Pereira, H. M., Ferrier, S., Walters, M., Geller, G. N., Jongman, R. H. G., Scholes, R. J., . . . Wegmann, M. (2013). Essential biodiversity variables. *Science*, 339(6117), 277–278. https://doi.org/10.1126/science.1229931

Picard, N., Magnussen, S., Banak, L. N., Namkosserena, S. & Yalibanda, Y. (2010). Permanent sample plots for natural tropical forests: A rationale with special emphasis on Central Africa. *Environmental Monitoring and Assessment, 164*(1-4), 279–295. https://doi.org/10.1007/s10661-009-0892-y

Pirie, T. J., Thomas, R. L. & Fellowes, M. D. E. (2016). Limitations to recording larger mammalian predators in savannah using camera traps and spoor. *Wildlife Biology*, 22(1), 13–21. https://doi.org/10.2981/wlb.00129

Poikane, S., Johnson, R. K., Sandin, L., Schartau, A. K., Solimini, A. G., Urbanič, G., . . . Böhmer, J. (2016). Benthic macroinvertebrates in lake ecological assessment: A review of methods, intercalibration and practical recommendations. *The Science of the Total Environment, 543*(Pt A), 123–134. https://doi.org/10.1016/j.scitotenv.2015.11.021

Priyadarshani, N., Marsland, S. & Castro, I. (2018). Automated birdsong recognition in complex acoustic environments: A review. *Journal of Avian Biology, 49*(5), jav-01447. https://doi.org/10.1111/jav.01447

Prosekov, A., Kuznetsov, A., Rada, A. & Ivanova, S. (2020). Methods for monitoring large terrestrial animals in the wild. *Forests*, *11*(8), 808. https://doi.org/10.3390/f11080808

Radoux, J., Bourdouxhe, A., Coos, W., Dufrêne, M. & Defourny, P. (2019). Improving ecotope segmentation by combining topographic and spectral data. *Remote Sensing*, *11*(3), 354. https://doi.org/10.3390/rs11030354

Ricotta, C. & Feoli, E. (2013). Does ordinal cover estimation offer reliable quality data structures in vegetation ecological studies? *Folia Geobotanica*, 48(4), 437–447. https://doi.org/10.1007/s12224-013-9152-6

Roberge, J.-M. & Angelstam, P. (2004). Usefulness of the umbrella species concept as a conservation tool. *Conservation Biology*, 18(1), 76–85. https://doi.org/10.1111/j.1523-1739.2004.00450.x

Rogers, E. M. (1995). *Diffusion of innovations* (4th ed.). New York, London: Free Press.

Rovero, F. & Ahumada, J. (2017). The Tropical Ecology, Assessment and Monitoring (TEAM) network: An early warning system for tropical rain forests. *The Science of the Total Environment*, *574*, 914–923. https://doi.org/10.1016/j.scitotenv.2016.09.146

Rovero, F. & Kays, R. (2021). Camera trapping for conservation. In S. A. Wich & A. K. Piel (Eds.), *Conservation Technology* (pp. 79–104). Oxford: Oxford University Press. https://doi.org/10.1093/oso/9780198850243.003.0005

Rovero, F., Zimmermann, F., Berzi, D. & Meek, P. (2013). "Which camera trap type and how many do I need?" A review of camera features and study designs for a range of wildlife research applications. *Hystrix, 24*(2), 148–156. https://doi.org/10.4404/hystrix-24.2-8789

Ruppert, K. M., Kline, R. J. & Rahman, M. S. (2019). Past, present, and future perspectives of environmental DNA (eDNA) metabarcoding: A systematic review in methods, monitoring, and applications of global eDNA. *Global Ecology and Conservation*, *17*, e00547. https://doi.org/10.1016/j.gecco.2019.e00547

Sala, O. E., Chapin, F. S., Armesto, J. J., Berlow, E., Bloomfield, J., Dirzo, R., . . . Wall, D. H. (2000). Global biodiversity scenarios for the year 2100. *Science*, 287(5459), 1770–1774. https://doi.org/10.1126/science.287.5459.1770

Schieck, J. (1997). Biased detection of bird vocalizations affects comparisons of bird abundance among forested habitats. *The Condor*, *99*(1), 179–190. https://doi.org/10.2307/1370236

Schulz, B. K., Bechtold, W. A. & Zarnoch, S. J. (2009). Sampling and estimation procedures for the vegetation diversity and structure indicator. (General Technical Report PNW-GTR-781). United States Department of Agriculture, Forest Service, Pacific Northwest Research Station, Portland. https://www.fs.usda.gov/pnw/pubs/pnw_gtr781.pdf

Seimon, T. A., Seimon, A., Yager, K., Reider, K., Delgado, A., Sowell, P., . . . Halloy, S. (2017). Long-term monitoring of tropical alpine habitat change, Andean anurans, and chytrid fungus in the Cordillera Vilcanota, Peru: Results from a decade of study. *Ecology and Evolution*, 7(5), 1527–1540. https://doi.org/10.1002/ece3.2779

SERNANP (2015). *Plan maestro del Parque Nacional Cerros de Amotape*. Lima. Retrieved from Servicio Nacional de Áreas Naturales Protegidas por el Estado website: https://shorturl.at/epqV3

Sethi, S. S., Jones, N. S., Fulcher, B. D., Picinali, L., Clink, D. J., Klinck, H., . . . Ewers, R. M. (2020). Characterizing soundscapes across diverse ecosystems using a universal acoustic feature set. *Proceedings of the National Academy of Sciences of the United States of America*, 117(29), 17049–17055. https://doi.org/10.1073/pnas.2004702117

Siitonen, J. (2001). Forest management, coarse woody debris and saproxylic organisms: Fennoscandian boreal forests as an example. In B. G. Jonsson & N. Kruys (Eds.), *Ecological Bulletins, no. 49. Ecology of woody debris in boreal forests* (pp. 11–41). Oxford: Blackwell Science.

Silvertown, J., Harvey, M., Greenwood, R., Dodd, M., Rosewell, J., Rebelo, T., . . . McConway, K. (2015). Crowdsourcing the identification of organisms: A case-study of iSpot. *ZooKeys*, (480), 125–146. https://doi.org/10.3897/zookeys.480.8803

Skidmore, A. K., Coops, N. C., Neinavaz, E., Ali, A., Schaepman, M. E., Paganini, M., . . . Wingate, V. (2021). Priority list of biodiversity metrics to observe from space. *Nature Ecology & Evolution*, 5(7), 896–906. https://doi.org/10.1038/s41559-021-01451-x

Skrbinšek, T., Luštrik, R., Majić-Skrbinšek, A., Potočnik, H., Kljun, F., Jelenčič, M., . . . Trontelj, P. (2019). From science to practice: Genetic estimate of brown bear population size in Slovenia and how it influenced bear management. *European Journal of Wildlife Research*, 65(2). https://doi.org/10.1007/s10344-019-1265-7

Smit, I. P. J., Asner, G. P., Govender, N., Vaughn, N. R. & van Wilgen, B. W. (2016). An examination of the potential efficacy of highintensity fires for reversing woody encroachment in savannas. *Journal of Applied Ecology*, 53(5), 1623–1633. https://doi.org/10.1111/1365-2664.12738

Stahlschmidt, P. & Brühl, C. A. (2012). Bats as bioindicators – the need of a standardized method for acoustic bat activity surveys. *Methods in Ecology and Evolution*, 3(3), 503–508. https://doi.org/10.1111/j.2041-210X.2012.00188.x

Stark, J. D., Boothroyd, I. K. G., Harding, J. S., Maxted, J. R. & Scarsbrook, M. R. (2001). Protocols for sampling macroinvertebrates in wadeable streams: New Zealand Macroinvertebrate Working Group Report No. 1. Prepared for Ministry for the Environment. Retrieved from Engineering New Zealand/Water NZ Rivers Group website: https://rb.gy/4ynz18

Steenweg, R., Whittington, J., Hebblewhite, M., Forshner, A., Johnston, B., Petersen, D., . . . Lukacs, P. M. (2016). Camerabased occupancy monitoring at large scales: Power to detect trends in grizzly bears across the Canadian Rockies. *Biological Conservation, 201*, 192–200. https://doi.org/10.1016/j.biocon.2016.06.020

Stephenson, P. J. (2019). The Holy Grail of biodiversity conservation management: Monitoring impact in projects and project portfolios. *Perspectives in Ecology and Conservation*, *17*(4), 182–192. https://doi.org/10.1016/j.pecon.2019.11.003

Stephenson, P. J. (2020). Technological advances in biodiversity monitoring: Applicability, opportunities and challenges. *Current Opinion in Environmental Sustainability*, 45, 36–41. https://doi.org/10.1016/j.cosust.2020.08.005 Stevens, N., Erasmus, B. F. N., Archibald, S. & Bond, W. J. (2016). Woody encroachment over 70 years in South African savannahs: Overgrazing, global change or extinction aftershock? *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 371(1703). https://doi.org/10.1098/rstb.2015.0437

Strohbach, M., Daubert, J., Ravkin, H. & Lischka, M. (2016). Big data storage. In J. M. Cavanillas, E. Curry & W. Wahlster (Eds.), *New horizons for a data-driven economy* (pp. 119–141). Cham, Switzerland: Springer International Publishing. https://doi.org/10.1007/978-3-319-21569-3_7

Study Group on the Next National Biodiversity Strategies and Action Plans (2021). *Report of the Study Group on the Next National Biodiversity Strategies and Action Plans*. Retrieved from Nature Conservation Bureau, Ministry of the Environment, Government of Japan, Tokyo. Retrieved from Biodiversity Center of Japan. https://shorturl.at/fCRZ3

Sugai, L. S. M., Silva, T. S. F., Ribeiro, J. W. & Llusia, D. (2019). Terrestrial passive acoustic monitoring: Review and perspectives. *BioScience*, 69(1), 15–25. https://doi.org/10.1093/biosci/biy147

Szantoi, Z., Brink, A., Buchanan, G., Bastin, L., Lupi, A., Simonetti, D., . . . Davy, J. (2016). A simple remote sensing based information system for monitoring sites of conservation importance. *Remote Sensing in Ecology and Conservation, 2*(1), 16–24. https://doi.org/10.1002/rse2.14

Taberlet, P., Coissac, E., Hajibabaei, M. & Rieseberg, L. H. (2012a). Environmental DNA. *Molecular Ecology, 21*(8), 1789–1793. https://doi.org/10.1111/j.1365-294X.2012.05542.x

Taberlet, P., Coissac, E., Pompanon, F., Brochmann, C. & Willerslev, E. (2012b). Towards next-generation biodiversity assessment using DNA metabarcoding. *Molecular Ecology*, 21(8), 2045–2050. https://doi.org/10.1111/j.1365-294X.2012.05470.x

Takeuchi, A., Iijima, T., Kakuzen, W., Watanabe, S., Yamada, Y., Okamura, A., . . . Tsukamoto, K. (2019). Release of eDNA by different life history stages and during spawning activities of laboratory-reared Japanese eels for interpretation of oceanic survey data. *Scientific Reports*, 9(1), 6074. https://doi.org/10.1038/s41598-019-42641-9

Thompson, K.-L., Lantz, T. C. & Ban, N. C. (2020). A review of Indigenous knowledge and participation in environmental monitoring. *Ecology and Society*, 25(2), 10. https://doi.org/10.5751/ES-11503-250210

Timofieieva, O., Hukov, V., Kovalov, V. & Vlaschenko, A. (2019). How to make bat's life easier: A new method of chemiluminescent light tags attachment. *Barbastella*, 12(1). https://rb.gy/qsnebt

Toyama, H., Tagane, S., Aiba, S., Ugawa, S., Suzuki, E., Yamazaki, K., . . . Takeuchi, Y. (2023). High plant diversity and characteristic plant community structure in broad-leaved evergreen forests on Amami-Oshima and Tokunoshima Islands, Japan's newest natural World Heritage Site. *Ecological Research*, 38(3), 403–419. https://doi.org/10.1111/1440-1703.12381

Tuysuzoglu, G., Birant, D. & Pala, A. (2018). Ensemble methods in environmental data mining. In C. Thomas (Ed.), *Data Mining*. InTechOpen. https://doi.org/10.5772/intechopen.74393

United Nations Environment Programme-World Conservation Monitoring Centre (UNEP-WCMC) (2023a). Explore the world's protected areas: December 2023 update of the WDPA and WD-OECM. Retrieved from https://shorturl.at/auEPR

UNEP-WCMC (2023b). Factsheet – Indicators for the Post 2020 Global Biodiversity Framework: Indicator Repository. Retrieved from https://www.post-2020indicators.org/metadata/headline/3-1

United Nations Educational, Scientific and Cultural Organization (UNESCO) (2020). *Statutory framework of the World Network of Biosphere Reserves* [Brochure]. Paris: UNESCO. https://unesdoc.unesco.org/ark:/48223/pf0000373378

UNESCO (2021). Technical guidelines for biosphere reserves. Paris: UNESCO. Retrieved from UNESCO website: https://en.unesco.org/sites/default/files/tgbr_en.pdf UNESCO (2022). UNESCO in brief. Paris: UNESCO. Retrieved from https://www.unesco.org/en/brief

Verma, A., van der Wal, R. & Fischer, A. (2016). Imagining wildlife: New technologies and animal censuses, maps and museums. *Geoforum*, 75, 75–86. https://doi.org/10.1016/j.geoforum.2016.07.002

Wägele, J. W., Bodesheim, P., Bourlat, S. J., Denzler, J., Diepenbroek, M., Fonseca, V., . . . Wildermann, S. (2022). Towards a multisensor station for automated biodiversity monitoring. *Basic* and Applied Ecology, 59, 105–138. https://doi.org/10.1016/j.baae.2022.01.003

Wang, B., Brown, D., Gao, Y. & La Salle, J. (2015). MARCH: Multiscale-arch-height description for mobile retrieval of leaf images. *Information Sciences*, 302, 132–148. https://doi.org/10.1016/j.ins.2014.07.028

Wearn, O. & Glover-Kapfer, P. (2017). Camera-trapping for conservation: A guide to best-practices. Woking, UK: WWF-UK. https://doi.org/10.13140/RG.2.2.23409.17767

Weaver, P. (2010). Understanding programs and projects – Oh, there's a difference! PMI Global Congress 2010-Asia Pacific, Melbourne, Victoria, Australia. Newtown Square, PA. Retrieved from Project Management Institute website: https://shorturl.at/uBFY6

Weigand, H., Beermann, A. J., Čiampor, F., Costa, F. O., Csabai, Z., Duarte, S., . . . Ekrem, T. (2019). DNA barcode reference libraries for the monitoring of aquatic biota in Europe: Gap-analysis and recommendations for future work. *The Science of the Total Environment*, 678, 499–524. https://doi.org/10.1016/j.scitotenv.2019.04.247

Weiser, M. (1999). The computer for the 21st century. *ACM SIGMOBILE Mobile Computing and Communications Review*, *3*(3), 3–11. https://doi.org/10.1145/329124.329126

Westgate, M. J., Likens, G. E. & Lindenmayer, D. B. (2013). Adaptive management of biological systems: A review. *Biological Conservation*, 158, 128–139. h ttps://doi.org/10.1016/j.biocon.2012.08.016

White, E. R. (2019). Minimum time required to detect population trends: The need for long-term monitoring programs. *BioScience*, 69(1), 40–46. https://doi.org/10.1093/biosci/biy144

Whyte, K. P. (2013). On the role of traditional ecological knowledge as a collaborative concept: A philosophical study. *Ecological Processes, 2*(1). https://doi.org/10.1186/2192-1709-2-7

Wilkinson, M. D., Dumontier, M., Aalbersberg, I. J. J., Appleton, G., Axton, M., Baak, A., . . . Mons, B. (2016). The FAIR guiding principles for scientific data management and stewardship. *Scientific Data*, *3*, 160018. https://doi.org/10.1038/sdata.2016.18

Working Group for Comprehensive Assessment of Biodiversity and Ecosystem Services (2021). *Japan biodiversity outlook 3:* 2021 report of comprehensive assessment of biodiversity and ecosystem services in Japan. [Summary for Policymakers]. Tokyo. Retrieved from Nature Conservation Bureau, Ministry of the Environment website: https://shorturl.at/uEJSU

Young, S., Rode-Margono, J. & Amin, R. (2018). Software to facilitate and streamline camera trap data management: A review. *Ecology* and Evolution, 8(19), 9947–9957. https://doi.org/10.1002/ece3.4464

Zafra-Calvo, N., Pascual, U., Brockington, D., Coolsaet, B., Cortes-Vazquez, J. A., Gross-Camp, N., . . . Burgess, N. D. (2017). Towards an indicator system to assess equitable management in protected areas. *Biological Conservation*, *211*, 134–141. https://doi.org/10.1016/j.biocon.2017.05.014

Zhang, J., Xiong, K., Liu, Z. & He, L. (2022). Research progress and knowledge system of world heritage tourism: A bibliometric analysis. *Heritage Science*, *10*(1), 42. https://doi.org/10.1186/s40494-022-00654-0h

Zuur, A., Ieno, E. N., Walker, N., Saveliev, A. A. & Smith, G. M. (2009). Mixed effects models and extensions in ecology with R Statistics for biology and health. New York, NY: Springer New York. Retrieved from https://link.springer.com/book/10.1007/978-0-387-87458-6

A framework for monitoring biodiversity in PAs and OECMs | 76

Annexes

What resources are required for the monitoring programme?		
Where will monitoring take place?	e place?	ı synergies can be used?
What will be monitored?	When will monitoring tak	Which
Why establish a monitoring programme?	re monitoring programme?	ing programme be implemented?
Statement of purpose	Who is involved in th	How will the monitor

Annex Figure 1 Blank monitoring concept worksheet The monitoring concept worksheet should be filled out during development of a biodiversity monitoring programme to identify how it will be conducted. *Source: Compiled by the report authors*



sedentary species. Through identification This question can be answered through Monitoring of entire habitats will require a different approach than monitoring of What is the spatial scale of the

- monitoring technique?
- habitat type, location)?
- Does the monitoring programme use an area- or plot-based design?
 - How are the plots distributed, e.g. transect line, randomly, grid, stratified sample?
- What is the minimum mapping unit or the resolution of the spatial data?

Where will monitoring take place?

of the spatial resolution of the target of observation, a catalogue of suitable single species. Highly mobile species will require a different approach than detailed assessment of the site. options will become apparent

- How is the area of interest defined (e.g.

- erns about the biodiversity monitoring system? , cultural, or ethical Are there social
 - Are legal constraints expected?
 - Is a timetable already in place?
- At what point should the manual be finalised?

Which synergies can be used?

- Are existing baseline datasets available?
 Are there existing projects (internal / external) that may be beneficial to the planned monitoring (e.g. methodology, resources, data processing)?

What resources are required for the monitoring programme?

There are two main components to answer the monitoring cycles. Generally, the establishmen costs between the establishment phase of the question. A realistic estimate of financial costs should be made, considering the difference of monitoring programme (plot establishment programme and continuation in subsequent What is the estimated budget to set up a resources. The second component is the demand for human resources. phase will demand the greatest financial

How many teams and human resources are software)? needed?

monitoring devices, IT infrastructure, data

- Does the organisation have administrative staff, permanent field workers, or seasonal technicians available for the programme?
 - Will the available funding and human resources be sufficient to effectively
- availability in subsequent monitoring cycles complete the monitoring process, and will be guaran
- Are there human resources for analysis and reporting of long-term monitoring results or must this be outsourced?
 - Can supplemental resources be mobilised if needed?

Annex Figure 2 Cheat sheet for monitoring concept worksheet Source: Compiled by the report authors



Annex Figure 3 Blank stakeholder worksheet

The stakeholder worksheet should be filled out during the conceptual phase to identify potential stakeholders. *Source: Compiled by the report authors*

Annex Table 1 Point system to rate biodiversity monitoring programme priorities

The management team of the protected area or OECM distributes a fixed number of points amongst the categories. For the best evaluation, all points must be distributed. The sum of points across rows indicates the relative purpose of the monitoring programme by different stakeholders. The sum of points across columns indicates the relative scale on which the programme is operational.

Category	Use of results	Local	Regional	National	International	Total
Purpose of monitoring programme	Planning (management)	0-5	0-5	0-5	0-5	
	Evaluation (management)	0-5	0-5	0-5	0-5	
	Governance (reporting)	0-5	0-5	0-5	0-5	
	Activation (stakeholder contribution)	0-5	0-5	0-5	0-5	
Knowledge	Public relations (stakeholder outreach of information)	0-5	0-5	0-5	0-5	
transfer	Science (research)	0-5	0-5	0-5	0-5	
	Education (awareness)	0-5	0-5	0-5	0-5	
	Total					

Source: Compiled by the report authors

Annex Table 2 Cost worksheet describing the human resources, travel and material expenses for data collection

Category	Cost category	Required resources	Sum available resources	Gaps / Comments
Human resources	staff	working days x per diem		
	volunteer	working days x per diem + incentives		
	external experts	working days x daily rate		
Travel costs	allowance	days x allowance (travel)		
	accommodation	days x accommodation cost		
	travel costs	public transportation + car rental + distance allowance		
Material costs	investment	estimate		
	consumables	estimate		
	data, purchasing, licenses	estimate		
	communication	estimate		

Source: Compiled by the report authors

Checklist 1 Ethical and cultural considerations of the implementation phase

✓ Legal requirements

- Obtain appropriate permits from authorities and landowners
- Follow regulations when working with genetic materials (Kyoto Protocol)

✓ Behaviour

- Respectfully acknowledge and include stakeholders
- Keep noise to a minimum
- Do not litter

✓ Data management

- Follow FAIR data sharing principles
- Adhere to privacy rules and regulations
- Mask sensitive data whilst providing as much transparency as possible

✓ Transparency

- Clearly mark monitoring plots with signage
- Provide sufficient information and a way to contact responsible authorities
- Make results available as soon as reasonably possible

✓ Risk mitigation

- Generate management programme acceptance by the local population
- Follow nature conservation regulations
- Avoid unintended effects of monitoring or management

Minimise your impact on nature

- Do not extract resources from the site
- Use the least invasive techniques
- Do not harm animals or habitats
- Avoid trampling vegetation or creating new paths
- Drive or park only in designated areas

Checklist 2 Field logistics

✓ Identify field team

- Calculate number of technicians
- Assign field crew leader
- Assign field data collection manager
- Obtain permissions and guides
- Check proper accommodation and transportation
- Schedule site visits
- ✓ Define methods of sample collection
 - Plot establishment
 - Tool configuration
 - Acquire appropriate equipment
 - Data management
 - Define backup system



- Post-evaluation
- Conduct training and pre-runs
- Assign specialised tasks
- Solicit feedback from field team
- Provide feedback to field team

Checklist 3 Safety and field training

✓ Training exercises

- Worker safety
- First aid training
- How to implement the protocols

Conduct the test run

Set up tools

1

1

- Collect data
- Analyse and compare data from different teams from the same plot
- Present results

Checklist 4 Safe field work and data acquisition

✓ Logistics / equipment list

- Check weather forecast
- Book accommodations
- Means of transport
- Food supply
- Energy supply
- First aid kit and sun cream
- Good shoes and outdoor clothing including rain protection
- Sample storage options determined
- Labelling systems / stickers prepared

Technical preparation

- All devices are charged
- Cooling devices are ready
- Cross-check the completeness of tools including user manual
- Spare batteries / rechargeable battery pack
- Spare pens and notebooks
- Archive the data in the simplest structure
- Rainproof protection for electronic devices and paper

√ Prepa

- Preparation for field work
- Daily backup for electronic devices
- Plot visitation plan
- Map or GPS device to find plots
- Check that all equipment is ready for transport
- Fully loaded devices and rechargeable battery pack

✓ Data collection

- Track your way
- Overview pictures
- Use optimal workflow
- Use systematic, uniform workflow
- Check data completeness before leaving the site
- Report issues to manager
- Track any changes to protocol

Checklist 5 Data management

✓ Collection of data and metadata

- Unique plot identifier for all types of data in every monitoring cycle, i.e. images, GIS, spreadsheets
- Define metadata
- Use simple, standardised data sheet structure
- Use standardised formats drop-down lists
- Determine quantitative vs. qualitative variables
- Label data / samples with date, time, name of collector and location of collection
- Note which method was used, which preserving agent was used (e.g. EtOH 96 per cent)
- Record site factors, i.e. habitat, weather, etc.
- Note special observations

✓ Data storage

- Data directory and naming based on institutional standard
- Enter data as soon as possible
- Archive the data in the simplest structure
- Data integration into existing structures and products
- Metadata protocol is fulfilled
- Mark the most recent version of data
- IT security standard implemented

✓ Data backup

- Daily data backup during processing period
- Scanning of survey data sheets
- Spatially separated data backup for long-term storage

✓ Data analysis

- Decide on which statistical approaches will be used
- Decide on who will analyse the data
- Store all interim results during analysis
- Document all steps of analysis

✓ Communication: Reporting the data

- Consider the format of reporting to different groups
- Keep in mind the frequency of reporting requirements
- Interpret the results to explain to stakeholders

✓ Archiving

- Delete sensitive data according to legal requirements
- Delete working documents at the end of project
- Check for duplicate files

Checklist 6 Communication

✓ Reporting to decision-makers

- Stakeholder reporting
- Scientific output
- Public outreach

√ Transparency

- Open data access
- Dashboard for a good overview
- Report findings regardless of the monitoring outcome
- Share the results with all parties

Author profiles

Dr Vanessa Adams is an Associate Professor in Conservation and Planning in the School of Geography, Planning, and Spatial Sciences at the University of Tasmania. Her research focuses on modelling dynamic social-ecological systems to inform conservation decisions that improve ecosystems and the communities they support. This means her research is broadly aligned with three themes: ecological modelling (to understand dynamic ecosystems), socioeconomic aspects of conservation (to understand the human dimensions of social-ecological systems), and conservation decision theory (to inform decision-making). She has worked in a variety of roles ranging from actuarial analyst for global consulting firm Mercer HR to research scientist at universities. She spent a year as a Fulbright scholar conducting research at University of Queensland in 2004 and completed her PhD at James Cook University in 2011.

Vanessa Berger, MSc, is a lecturer and senior researcher at Carinthia University of Applied Sciences (CUAS), Austria. Having worked in several protected areas as an ecological consultant, she is an experienced field ecologist. In addition to being a certified UAV pilot, she is a specialist in all phases of holistic data acquisition from preparation of data bases, collection of field data, post-processing of orthophotos, statistical and geospatial analysis, ecological modelling and report writing. At CUAS, she tests state-of-the-art monitoring technologies, comparing them with traditional methods and tools. She is involved in development of standardised monitoring tool platforms, training materials and video tutorials for diverse stakeholders.

Judith Botha holds a Master's degree in Science from the University of Pretoria. She has been working with the Scientific Services division at SANParks for just over 23 years and focuses on biophysical monitoring, analyses and data management.

Sunita Chaudhary is an Ecosystem Services Specialist at International Centre for Integrated Mountain Development (ICIMOD), a member of the National Council of Environmental Protection and Climate Change Management, Prime Minister's Office, Government of Nepal, and a lead author of IPBES nexus assessment. At ICIMOD, she is responsible for research, policy inputs, capacity building and advocacy of evidence-based policy, and contributes to the interface of science, policy and practice. Her applied and interdisciplinary work focuses on human-nature interactions of terrestrial and freshwater ecosystems in the Hindu Kush Himalaya region. Dr Chaudhary is an established forester with around 15 years of experience in natural resources management in Australia, Austria, and Hindu Kush Himalaya. She has a PhD from Macquarie University, Australia and an MSc in Management of Protected Areas, Austria. She also attended the graduate leadership programme at the University of Hawaii, USA. She was a visiting scholar at the Department of Geography, Cambridge University, United Kingdom. Dr Chaudhary has more than 40 publications in high impact factor peer-reviewed scientific journals, and regularly writes opinion article in the national, regional and international media outlets.

Daniel Dalton, PhD, is a Senior Researcher at the UNESCO Chair on Sustainable Management of Conservation Areas, Carinthia University of Applied Sciences, Austria. His current work involves conceptualisation of monitoring programmes and testing technological solutions for biodiversity monitoring in conservation areas and agroecosystems. He has had significant vocational experiences at ecological field stations in North America and Europe and was an applied entomological research scientist for more than a decade, publishing more than 50 journal articles, book chapters and Extension publications.

Stephan Halloy is a Senior Adviser on risk assessment and climate change at the Ministry for Primary Industries, New Zealand. Previously he was Professor of Ecology at several universities in Bolivia and Argentina, Science Coordinator for The Nature Conservancy, Conservation International and World Wildlife Fund in South America, and Scientist at Crop & Food Research, New Zealand. He has worked in sustainable development, conservation, dynamics, complex systems and climate change for fifty years. His contributions include the development of GLORIA (Global Observation Research Initiative in Alpine Environments) monitoring system in New Zealand, South America and Africa. He has published over 125 research papers.

Robbie Hart has a PhD in Biology from University of Missouri St. Louis and a BA from Swarthmore College. At Missouri Botanical Garden, he directs the William L. Brown Center, an endowed centre of excellence dedicated to the study of useful plants, understanding the relationships between humans, plants, and their environment, the conservation of plant species, and the preservation of traditional knowledge for the benefit of future generations. His own research is on highelevation plant ecology, climate change and ethnobotany, and particularly the areas in which these three topics overlap.

Michael Jungmeier is a professor nature conservation and sustainability at Carinthia University of Applied Sciences, Austria. As an ecologist and human geographer, he holds the UNESCO Chair on Sustainable Management of Conservation Areas. As professor, he plays a key role in different educational programs such as the international Master of Science programme in Management of Conservation Areas and the certification course in Nature Conservation Engineering.

Hanns Kirchmeir studied at the University of Vienna at the Department of Vegetation Ecology and Conservation Research and also worked there for 5 years on the UNESCO Man and the Biosphere project "Hemeroby of Austrian Forest Ecosystems". Since 2020, he has been Managing Director of E.C.O. Institute of Ecology in Klagenfurt, Austria and has worked there for more than 25 years as a partner and project manager. His key fields of expertise are in forest ecology, biodiversity monitoring and protected area management. He is engaged in long-term monitoring activities in protected areas as well as on the national level. Dr rer nat **Vid Švara** is a senior scientist at the UNESCO Chair on Sustainable Management of Conservation Areas at Carinthia University of Applied Sciences. He obtained a doctoral degree in natural sciences at the Goethe University in Frankfurt am Main, Germany and conducted his research at the Helmholtz Centre for Environmental Research in Leipzig, Germany. In his research, he focuses on biodiversity assessment from diverse environmental and organismal samples using molecular methods. Applying these methods, he is contributing to research and conservation activities in different European countries. He has published papers on species distribution, taxonomy, phylogeny, use of eDNA in biodiversity assessment and human impact on genetic diversity of freshwater fauna. He has also co-authored a book Fauna Helvetica Amphipoda on biodiversity of freshwater amphipods in Switzerland.

Katia Torres Ribeiro has a PhD in ecology from Rio de Janeiro Federal University where she investigated community ecology and insect-plant interactions. She has been part of the permanent staff of Brazil's system of protected areas since 2002. For seven years she worked in the Serra do Cipó National Park, leading research and fire management. For six of those years she coordinated the research area, implementing important support tools such as scholarship programs, financing and seminars on a national scale. As head of the research and monitoring coordination, also for six years, she developed the strategic research plan, the National Biodiversity Monitoring Program (Programa Monitora) and biodiversity information systems. She is currently director in the fields of participatory management, socio-biodiversity production and land regularisation. At the National School of Tropical Botany, she is a professor in the professional postgraduate course Biodiversity in Protected Areas, guiding research in the areas of integrated fire management, adaptive planning and management, participatory management, governance and superposition with traditional territories.
PROTECTED AREA AND OECM DEFINITIONS, MANAGEMENT CATEGORIES AND GOVERNANCE TYPES

IUCN defines a protected area as:

A clearly defined geographical space, recognised, dedicated and managed, through legal or other effective means, to achieve the long-term conservation of nature with associated ecosystem services and cultural values.

The definition is expanded by six management categories (one with a sub-division), summarized below.

Ia Strict nature reserve: Strictly protected for biodiversity and also possibly geological / geomorphological features, where human visitation, use and impacts are controlled and limited to ensure protection of the conservation values.

Ib Wilderness area: Usually large unmodified or slightly modified areas, retaining their natural character and influence, without permanent or significant human habitation, protected and managed to preserve their natural condition.

II National park: Large natural or near-natural areas protecting large-scale ecological processes with characteristic species and ecosystems, which also have environmentally and culturally compatible spiritual, scientific, educational, recreational and visitor opportunities.

III Natural monument or feature: Areas set aside to protect a specific natural monument, which can be a landform, sea mount, marine cavern, geological feature such as a cave, or a living feature such as an ancient grove.

IV Habitat/species management area: Areas to protect particular species or habitats, where management reflects this priority. Many will need regular, active interventions to meet the needs of particular species or habitats, but this is not a requirement of the category.

V Protected landscape or seascape: Where the interaction of people and nature over time has produced a distinct character with significant ecological, biological, cultural and scenic value: and where safeguarding the integrity of this interaction is vital to protecting and sustaining the area and its associated nature conservation and other values.

VI Protected areas with sustainable use of natural resources: Areas which conserve ecosystems, together with associated cultural values and traditional natural resource management systems. Generally large, mainly in a natural condition, with a proportion under sustainable natural resource management and where low-level non-industrial natural resource use compatible with nature conservation is seen as one of the main aims.

The category should be based around the primary management objective(s), which should apply to at least three-quarters of the protected area – the 75 per cent rule.

The management categories are applied with a typology of governance types – a description of who holds authority and responsibility for the protected area. IUCN defines four governance types:

Type A. Governance by government: Federal or national ministry/agency in charge; sub-national ministry or agency in charge (e.g. at regional, provincial, municipal level); government-delegated management (e.g. to NGO).

Type B. Shared governance: Transboundary governance (formal and informal arrangements between two or more countries); collaborative governance (through various ways in which diverse actors and institutions work together); joint governance (pluralist board or other multi-party governing body).

Type C. Private governance: Conserved areas established and run by individual landowners; non-profit organizations (e.g. NGOs, universities) and for-profit organizations (e.g. corporate landowners).

Type D. Governance by Indigenous peoples and local communities: Indigenous peoples' conserved areas and territories – established and run by Indigenous peoples; community conserved areas – established and run by local communities.

The Convention on Biological Diversity defines an "other effective area-based conservation measures" (OECM) as: A geographically defined area other than a Protected Area, which is governed and managed in ways that achieve positive and sustained long term outcomes for the in situ conservation of biodiversity, with associated ecosystem functions and services and, where applicable, cultural, spiritual, socioeconomic, and other locally relevant values.

This covers three main cases:

- 1. **Ancillary conservation** areas delivering in-situ conservation as a by-product of management, even though biodiversity conservation is not an objective (e.g. some war grave sites).
- Secondary conservation active conservation of an area where biodiversity outcomes are only a secondary management objective (e.g. some conservation corridors).
- 3. **Primary conservation** areas meeting the IUCN definition of a protected area, but where the governance authority (i.e. community, Indigenous peoples' group, religious group, private landowner or company) does not wish the area to be reported as a protected area.

For more information on the IUCN definition, categories and governance types, see Dudley (2008). *Guidelines for applying protected area management categories*, which can be downloaded at: https://doi.org/10.2305/IUCN.CH.2008.PAPS.2.en

For more on governance types, see Borrini-Feyerabend et al. (2013). *Governance of Protected Areas: From understanding to action*, which can be downloaded at https://portals.iucn.org/library/node/29138.

For more information on OECMs, see Jonas et al. (2023) *Site-level tool for identifying other effective area-based conservation measures (OECMs): first edition*, which can be downloaded at: https://portals.iucn.org/library/node/51296





INTERNATIONAL UNION FOR CONSERVATION OF NATURE

WORLD HEADQUARTERS Rue Mauverney 28 1196 Gland, Switzerland Tel: +41 22 999 0000 Fax: +41 22 999 0002 www.iucn.org www.iucn.org/resources/publications